

ADVERTIMENT. L'accés als continguts d'aquesta tesi queda condicionat a l'acceptació de les condicions d'ús establertes per la següent llicència Creative Commons:

ADVERTENCIA. El acceso a los contenidos de esta tesis queda condicionado a la aceptación de las condiciones de uso establecidas por la siguiente licencia Creative Commons:

WARNING. The access to the contents of this doctoral thesis it is limited to the acceptance of the use conditions set by the following Creative Commons license: https://creativecommons.org/licenses/?lang=en

BY NC ND

Sobolev Inequalities: Isoperimetry and Symmetrization

By

WALTER ANDRÉS ORTIZ VARGAS



Advisor : Dr. Joaquim Martín Pedret

A thesis submitted in partial fulfillment of the requirements for the degree of Doctor in Mathematics

Department de Matemátiques UNIVERSITAT AUTÓNOMA DE BARCELONA

Bellaterra, September 17, 2019

Joaquim Martín Pedret, Professor at the Department of Mathematics of the Autonomous University of Barcelona.

Certify:

That this dissertation presented to the Faculty of Sciences, department of Mathematics of Autonomous University of Barcelona by Walter Andrés Ortiz Vargas in fulfillment of the requirements for the degree of Doctor of Mathematics.

Bellaterra, September 17, 2019

Joaquim Martín Pedret

To my Parents

Loneliness is necessary to enjoy our own heart and to love, but To succeed in life, it is necessary to give something of our life to the greatest number of people. Beyle Henri

Acknowledgements

Primero que todo, agradecer al Dios del cielo, por permitirme alcanzar esta meta. Al professor Joaquim Martín per acceptar-me com el seu estudiant, per ser el meu exemple a seguir i guia en aquest procés, per les seves exigències, la seva infinita paciència, per dipositar la seva confiança en mi i ensenyar-me que el fer matemàtiques és sense pressa peró sense pausa; gràcies a la seva ajuda ha estat possible culminar aquest treball. Moltes gràcies "Profe Joaquim." Al professor Xavier Tolsa per donar-me l'oportunitat de fer el doctorat sota la financiació del seu projecte. I would like to express my sincere gratitude to Professor Tapio Rajala, for giving me the opportunity to visit the Departament of Mathematics and statistic, of Jyväskylä University in Finland . It was a great pleasure and even more an honor for me to work with you.

Al departamento de Matemáticas por acogerme como un miembro más de su equipo docente y a los profesores con quienes compartí docencia en distintas asignaturas. Agradezco a Francisco Carreño "don Paco" por los cafés y momentos " criollos" compartidos , a Beatriz Díaz "mi prima" momentos en el Caliu, a Maria José Calejo por los buenos momentos y tertulias a las 17:00 y los préstamos para el café (quedó la canción de los Morat). A Loli Garcia "Lolita"' mi veci, por todos momentos, permitirme entrar a la oficina después de tocar la puerta y los 499 favores durante estos 4 años. A Ignasi Utzet, por haberme encontrado el mejor sitio de trabajo en el departamento (despacho c1/-164). A María Padilla "Mafe" por acogerme en el despacho y sus enseñazas. Amanda Fernández "Faracha" mi amiga de batallas, financiadora de los "pocos" cafés que bebía y compi de despacho quien me acogió y enseñó todo sobre Barcelona gracias "Bellaterra".

A todos los amigos y conocidos en Barcelona, especialmente a Federica "Viic" (.l. la italiana más chévere), mi amiga y eterna compañera de piso, gracias por los momentos compartidos y las chapas. Finalmente agradezco y dedico este trabajo a mi familia: mis padres Walter y Dora, mis hermanos Marlon y Camila, y a mi abuelita Amanda, a pesar de la distancia siempre han sido testigos de los triunfos y dificultades en este camino.

Mathematics Genealogy

	Otto Menken
Jacob Bernoulli	Johann Wichmannshausen
Johann Bernoulli	Christian Hausenr
Leonhard Euler	Abraham Kaestner
Joseph Lagrange	Johann Pfaff
Jean-Baptiste Fourier	Simeon Poisson Karl von Langsdorf Carl Gauß
Gustav Dirichlet	Martin Omh Christian Gerling
	Rudolf Lipschitz Julius Puckler
Eduardo Torroja	C. Felix (Christian) Klein
	Julio Rey Pastor
	Ricardo San Juan
	Joan Augé
	Rafael Aguiló
	Joan Cerdá
	Joaquim Martín
	Walter A. Ortiz

Contents

Ab	Abstract ii		iii
1	Intr	oduction	1
2	Preliminaries		11
	2.1	Decreasing rearrangement	11
	2.2	Rearrangement invariant spaces	17
		2.2.1 Indices	19
		2.2.2 Examples	20
3 An embedding theorem for Besov spaces		embedding theorem for Besov spaces	23
	3.1	Introduction	23
	3.2	Doubling measures	27
		3.2.1 Examples	33
	3.3	Symmetrization inequalities	37
		3.3.1 Pointwise estimates for the rearrangement	39
	3.4	Sobolev–Besov Embedding	43
		3.4.1 Some new function spaces	44
	3.5	Uncertainty type inequalities	48
	3.6	Embedding into BMO and essential continuity $\ldots \ldots \ldots$	50
		3.6.1 Essential continuity	51
	3.7	Sobolev type embeddings	53
4	Sym	metrization inequalities for convex profile	63
	4.1	Introduction	63
	4.2	Symmetrization and Isoperimetry	67
	4.3	Sobolev–Poincaré and Nash type inequalities	72
		4.3.1 Sobolev–Poincaré inequalities	72
		4.3.2 Nash inequalities	76
	4.4	Examples and applications	78
		4.4.1 Cauchy type laws	78
		4.4.2 Extended sub-exponential law	82
		4.4.3 Weighted Riemannian manifolds with negative dimension	84

Bibliography

85

Abstract

The first part of the thesis is devoted to obtain a Sobolev type embedding result for Besov spaces defined on a doubling metric space. This will be done by obtaining pointwise estimates between the special difference $f_{\mu}^{**}(t) - f_{\mu}^{*}(t)$ (called oscillation of f_{μ}^{*}) and the X-modulus of smoothness defined by

$$E_X(f,r) \coloneqq \left\| \oint_{B(x,r)} |f(x) - f(y)| \, d\mu(y) \right\|_X.$$

(here f_{μ}^{*} is the decreasing rearrangement of f, $f_{\mu}^{**}(t) = \frac{1}{t} \int_{0}^{t} f_{\mu}^{*}(s) ds$, for all t > 0 and X a rearrangement invariant space on Ω .

In the second part of the thesis, to obtain symmetrization inequalities on probability metric spaces that admit a convex isoperimetric estimator which incorporate in their formulation the isoperimetric estimator and that can be applied to provide a unified treatment of sharp Sobolev-Poincaré and Nash type inequalities.

Chapter 1 Introduction

This monograph is devoted to the study of Sobolev type embedding results in the setting of:

- Besov spaces defined on doubling metric spaces (Chapter 3),
- Probability metric spaces with convex isoperimetric profile (Chapter 4).

The history of Sobolev embeddings started in the thirties of the last century with Sobolev's famous embedding theorem: [91]

$$W_p^1(\Omega) \subset L^r(\Omega) \tag{1.0.1}$$

where $\Omega \subset \mathbb{R}^n$ is a bounded domain with sufficiently smooth boundary, L^r , $1 \le r \le \infty$ stands for the Lebesgue space, and $W_p^1(\Omega)$, 1 , are the classical Sobolev spaces. The latter have been widely accepted as one of the crucial instruments in functional analysis – in particular in connection with PDES – and have played a significant role in numerous parts of mathematics for many years. Sobolev's famous result (1.0.1) holds for <math>p < n and r such that $\frac{1}{n} - \frac{1}{p} \ge -\frac{1}{r}$ (strictly speaking [91] covers the case $\frac{1}{n} - \frac{1}{p} > -\frac{1}{r}$ whereas the extension to $\frac{1}{n} - \frac{1}{p} = -\frac{1}{r}$ was obtained later). In the limiting case, when p = n, the inclusion (1.0.1) does not hold for $r = \infty$, whereas for all $1 \le r < \infty$

$$W_n^1(\Omega) \subset L^r(\Omega). \tag{1.0.2}$$

Roughly speaking, the theory of Sobolev inequalities originated in classical inequalities from which properties of real functions can be deduced from those of its derivatives. In fact, (1.0.2) can be understood as the impossibility of specifying integrability conditions of functions in $W_n^1(\Omega)$ by means of $L^r(\Omega)$ conditions. Inequalities (1.0.1) and (1.0.2) are not optimal. In order to get further refinements, it is necessary to deal with a wider class of spaces. In the sixties of the last century, Peetre [84], Trudinguer [97] and Pohozarev [85] independently found refinements of (1.0.1) expressed in terms of Orlicz spaces. In 1979, Hansson [43] and Brezis and Wainger [11] showed independently that $W_n^1(\Omega)$ is embedded in a Lorentz–Zygmund type space. Limiting Sobolev embeddings, in more general settings, have been investigated by several authors (see [70] and the references quoted therein).

If instead of working on bounded domains with a nice boundary, we work in the full space, Sobolev's embedding theorem in \mathbb{R}^n states that (see [93]¹ and the references quoted therein):

$$W_p^1(\mathbb{R}^n) \subset \begin{cases} L^{\frac{np}{n-p},p}(\mathbb{R}^n) & p < n \text{ (subcritical case)}, \\ L^{\infty,p}(\log L)^{-1}(\mathbb{R}^n) & p = n \text{ (critical case)}, \\ L^{\infty}(\mathbb{R}^n) & p > n \text{ (supercritical case)}. \end{cases}$$
(1.0.3)

The Lorentz spaces $L^{p,q}(\mathbb{R}^n)$ are defined as the collection of functions of finite function quasi-norm

$$\|f\|_{L^{p,q}} = \left(\int_0^\infty \left(s^{1/p} f^*(s)\right)^q \frac{ds}{s}\right)^{1/q},$$

when $0 < p, q < \infty$, and

$$||f||_{p,\infty} = \sup_{0 < t < 1} s^{1/p} f^*(s).$$

when $q = \infty$ (f^* denotes the decreasing rearrangement of f). The Lorentz-Zygmund spaces $L^{\infty,q}(\log L)^{-1}$, $1 \le q < \infty$, are defined as the set of functions for which the quasi-norm

$$||f||_{L^{\infty,q}(\log L)^{-1}} = \left(\int_0^\infty \left(\frac{f^{**}(t)}{1+\log^+\left(\frac{1}{t}\right)}\right)^q \frac{dt}{t}\right)^{1/q},$$

is finite (where $f_{\mu}^{**}(t) = \frac{1}{t} \int_0^t f_{\mu}^*(s) ds$). Generalizations of (1.0.3) have been considered by replacing $W_p^1(\mathbb{R}^n)$ by a Besov space.

Given $0 < s < 1, 1 \le p < \infty$ and $1 \le q \le \infty$, the Besov space $\dot{B}^s_{p,q}(\mathbb{R}^n)$ is the linear set of functions $f \in L^p_{loc}(\mathbb{R}^n)$ of finite quasi-norm

$$\|f\|_{\dot{B}^s_{p,q}(\mathbb{R}^n)} \coloneqq \left(\int_0^\infty (t^{-s}\omega_p(f,t))^q \frac{dt}{t}\right)^{1/q}$$

where

$$\omega_p(f,t) \coloneqq \sup_{|h| \le t} \|f(x+h) - f(x)\|_{L^p(\mathbb{R}^n)}$$

is the L^p -modulus of continuity. Here the parameters s and q give a finer gradation of smoothness. The scales of Besov spaces $\dot{B}^s_{p,q}$, on \mathbb{R}^n , or in domains of \mathbb{R}^n ,

¹In the introduction of that paper there is an excellent history of the evolution of this problem.

were introduced between 1959 and 1975. A comprehensive treatment of these function spaces and their history can be found in Triebel's monographs [94], [95].

The Sobolev embedding in this context² states that

$$\dot{B}_{p,q}^{s}(\mathbb{R}^{n}) \in \begin{cases} L^{\frac{np}{n-sp},q}(\mathbb{R}^{n}) & p < \frac{n}{s} \text{ (subcritical case)}, \\ L^{\infty,q}(\log L)^{-1}(\mathbb{R}^{n}) & p = \frac{n}{s} \text{ (critical case)}, \\ L^{\infty}(\mathbb{R}^{n}) & p > \frac{n}{s} \text{ (supercritical case)}. \end{cases}$$
(1.0.4)

One proof of the subcritical case is based on real interpolation. We recall briefly the construction of real interpolation spaces (see [6] for a complete treatment). Let (A_0, A_1) be a pair of quasi-Banach spaces that are compatible in the sense that both A_0 and A_1 are continuously embedded in some common Hausdorff topological vector space \mathcal{H} . The K-functional is defined, for t > 0 and $f \in A_0 + A_1$, by

$$K(t, f; A_0, A_1) = \inf_{f=f_0+f_1} \{ \|f_0\|_{A_0} + t \|f_1\|_{A_1} \}$$

For 0 < s < 1 and $0 < q \le \infty$, the real interpolation space $\overline{A}_{s,q} = (A_0, A_1)_{s,q}$ is the set of all $f \in A_0 + A_1$ such that

$$\|f\|_{(A_0,A_1)_{s,q}} \coloneqq \begin{cases} \left(\int_0^\infty \left(\frac{K(t,f;\vec{A})}{t^s} \right)^q \frac{dt}{t} \right)^{1/q}, & 0 \le q < \infty \\ \sup_{t>0} t^{-s} K(t,f;\vec{A}), & q = \infty, \end{cases}$$

is finite.

Since (cf. [6])

$$K(t,f;L^{p}(\mathbb{R}^{n}),\dot{W}^{1}_{p}(\mathbb{R}^{n})) = \inf_{f=f_{0}+f_{1}} \{ \|f_{0}\|_{L^{p}} + t \, \|f_{1}\|_{\dot{W}^{1}_{p}} \} \simeq \omega_{p}(f,t),$$

we get

$$\|f\|_{(L^{p}(\mathbb{R}^{n}),\dot{W}^{1}_{p}(\mathbb{R}^{n}))_{s,q}} = \|f\|_{\dot{B}^{s}_{p,q}(\mathbb{R}^{n})}.$$

Using the fact that $L^p(\mathbb{R}^n) \subset L^p(\mathbb{R}^n)$ and that $\dot{W}^1_p(\mathbb{R}^n) \subset L^{\frac{np}{n-p},p}(\mathbb{R}^n)$, we obtain by interpolation

$$\dot{B}^{s}_{p,q}(\mathbb{R}^{n}) = (L^{p}, \dot{W^{1}_{p}})_{s,q}(\mathbb{R}^{n}) \subset (L^{p}, L^{\frac{np}{n-p},p})_{s,q}(\mathbb{R}^{n}) = L^{\frac{np}{n-sp},q}(\mathbb{R}^{n}).$$

Our main objective in Chapter 3 will be to give an extension of (1.0.4) in the context of doubling metric spaces³. A theory of Besov spaces on metric measure

²(See for example DeVore, Riemenschneider and Sharpley [18], Netrusov [82], Goldman and Kerman [29], Caetano and Moura [12],[13], Martín [60], or Haroske and Schneider [44]).

³Metric spaces play a prominent role in many fields of mathematics. In particular, they constitute natural generalizations of manifolds, admitting all kinds of singularities and still providing rich geometric structure.

spaces was developed in [38], which is a generalization of the corresponding theory of function spaces on \mathbb{R}^n (see [94],[95],[96]), respectively, Ahlfors *n*-regular metric measure spaces (see [39],[41]).

There are several equivalent ways to define Besov spaces in the setting of a doubling metric space (see for example [27],[28],[38],[75],[74],[105],[45] and the references therein). In Chapter 3, we shall use the approach based on a generalization of the classical L^p -modulus of smoothness introduced in [27].

Let (Ω, d, μ) be a metric measure space equipped with a metric d and a Borel regular outer measure μ , for which the measure of every ball is positive and finite. Given t > 0, $0 and <math>f \in L^p_{loc}(\Omega)$, the L^p -modulus of smoothness is defined by

$$E_p(f,t) = \left(\int_{\Omega} \left(\int_{B(x,t)} |f(x) - f(y)|^p d\mu(y)\right) d\mu(x)\right)^{1/p}$$

where $f_B f(x) d\mu(x) \coloneqq \frac{1}{\mu(B)} \int_B f(x) d\mu(x)$ is the integral average of a locally integrable function f over B.

Definition 1.0.1. For $0 < s < \infty$, the homogeneous Besov space $\dot{\mathcal{B}}^s_{p,q}(\Omega)$ consists of those functions $f \in L^p_{loc}(\Omega)$ for which the seminorm

$$\|f\|_{\dot{\mathcal{B}}^s_{p,q}(\Omega)} \coloneqq \begin{cases} \left(\int_0^\infty \left(\frac{E_p(f,t)}{t^s}\right)^q \frac{dt}{t}\right)^{1/q}, & 0 < q < \infty, \\ \sup_{t>0} t^{-s} E_p(f,t), & q = \infty, \end{cases}$$

is finite.

This definition is rather concrete and gives the usual Besov space in the Euclidean setting since $E_p(f,t)$ is equivalent to the classical $L^p(\mathbb{R}^n)$ -modulus (see (3.1.2) in Section 3.1 below). Moreover, it has been shown by Müller and Yang [74] that it coincides with the definition based on test functions used earlier by Han [40], Han and Yang [42], and Yang [103], provided that Ω , besides being doubling, also satisfies a reverse doubling condition.

The abstract variant of (1.0.4) for metric spaces is only known in the following particular case (see [27] and [45]):

Theorem 1. Let Ω be a Q-regular metric space, i.e. there exists a $Q \ge 1$ and a constant $c_Q \ge 1$ such that

$$c_Q^{-1}r^Q \le \mu(B(x,r)) \le c_Q r^Q$$

for each $x \in X$, and for all $0 < r < diam \Omega$ (here diam Ω is the diameter of Ω). Suppose that 0 < s < 1 and $1 \le q \le \infty$. Then: 1. (See ([27, Thm. 5.1])) Suppose Ω satisfies a (1, p)-Poincaré inequality, i.e. if there exist constants $C_p \ge 0$ and $\lambda \ge 1$ such that

$$\int_{B} |f - f_B| d\mu \leq \left(\int_{\lambda B} g^p d\mu \right)^{1/p}$$

for any locally integrable functions f for all upper gradients⁴ g of f. Then

$$\dot{\mathcal{B}}^{s}_{p,q}(\Omega) \subset L^{p(Q),q}_{\mu}(\Omega) \tag{1.0.5}$$

for 1 , where <math>p(Q) = Qp/(Q - sp).

2. (See ([45, Thm. 4.4])) If Ω is geodesic, i.e. every pair of points can be jointed by a curve whose length is equal to the distance between the points, then (3.1.3) holds for $1 \le p < Q/s$.

The proof of this theorem is based on the real interpolation method, for example in ([27, Thm. 5.1]) under a (1, p)-Poincaré inequality assumption, the Besov space $\dot{\mathcal{B}}_{p,q}^s(\Omega)$ is realized as the real interpolation space $(L^p(\Omega), KS_{1,p}(\Omega))_{\alpha,q}$ between the corresponding $L^p(\Omega)$ and the Sobolev space of Korevaar and Schoen $KS_{1,p}(\Omega)$, consist of measurable functions f of finite norm⁵

$$\|f\|_{KS_{1,p}(\Omega)} \coloneqq \limsup_{t \to 0} \frac{E_p(f,t)}{t}.$$
 (1.0.6)

They proved that $E_p(f,t)$ is equivalent to the K-functional between $L_p(\Omega)$ and $KS_{1,p}(\Omega)$. Moreover if Ω is Q-regular, then

$$\|f\|_{L^{Q_p}_{\mu}(\Omega)} \leq \|f\|_{KS_{1,p}(\Omega)}, \qquad (1.0.7)$$

and, consequently, interpolation allows one to obtain embedding theorems.

The key point in the previous argument is the embedding (1.0.7), which is only known for *Q*-regular spaces.

The purpose of Chapter 3 will be to obtain a Sobolev type embedding result for Besov spaces defined on a doubling metric space. In our investigation we will avoid the use of interpolation techniques that require the presence of a Sobolev type space. The main idea will be to extend to the metric side the Euclidean oscillation inequality

$$f^{**}(t) - f^{*}(t) \le 2^{1/p} \frac{\omega_p(f, t^{1/n})}{t^{1/p}}, \quad t > 0 \ (1 \le p < \infty).$$

⁴A non-negative Borel function g is an upper gradient of a function $f: \Omega \to \mathbb{R}$ if $|f(y) - f(x)| \leq \int_{\gamma} gd, s$ for every x and $y \in \Omega$ and every rectifiable path γ in Ω with endpoints x and y (see [46],[27]).

⁵When Ω is a Riemannian manifold this definition yields the usual Sobolev space and the quantity in (3.1.4) is equivalent to the usual semi-quasinorm (see [56]).

(Here $f^{**}(t) = \frac{1}{t} \int_0^t f^*(s) ds$ (see [55], [57], [59], [63] and the references quoted therein)).

Chapter 3 is organized as follows: Section 3.2 contains basic definitions and technical results on doubling metric measure spaces. In Section 3.3 we obtain pointwise estimates of the oscillation $O_{\mu}(f,t) = f_{\mu}^{**}(t) - f_{\mu}^{*}(t)$ in terms of the X-modulus of smoothness defined by

$$E_X(f,r) \coloneqq \left\| f_{B(x,r)} \left| f(x) - f(y) \right| d\mu(y) \right\|_X$$

Here, X is a rearrangement invariant space⁶ on Ω . In Section 4.3 we define generalized Besov type spaces and use the oscillation inequalities obtained in the previous sections to derive embedding Sobolev theorems. In Section 3.5 we deal with generalized uncertainty Sobolev inequalities in the context of Besov spaces. In Section 3.6 a criterion for essential continuity and for the embedding into $BMO(\Omega)$ will be obtained. Finally in Section 3.7 we will study in detail the case $\dot{\mathcal{B}}_{p,q}^{s}(\Omega)$ for 0 < s < 1, 0 and <math>0 .

The results contained in Chapter 3 has been published in Journal of Mathematical Analysis and Applications (see [67]).

In the second part of this memoir (Chapter 4) we will study Sobolev inequalities in metric spaces with convex isoperimetric profile. In order to explain what our main objective will be, we now recall some definitions.

Let (Ω, d, μ) be a connected metric space equipped with a separable Borel probability measure μ . The perimeter or Minkowski content of a Borel set $A \subset \Omega$ is defined by

$$\mu^{+}(A) = \liminf_{h \to 0} \frac{\mu(A_h) - \mu(A)}{h},$$

where $A_h = \{x \in \Omega : d(x, A) < h\}$ is the open *h*-neighbourhood of *A*. The **isoperimetric profile** I_{μ} is defined as the pointwise maximal function $I_{\mu} : [0, 1] \rightarrow [0, \infty)$ such that

$$\mu^+(A) \ge I_\mu(\mu(A))$$

for all Borel sets A. An isoperimetric inequality measures the relation between the boundary measure and the measure of a set, by providing a lower bound on I_{μ} by some function $I:[0,1] \rightarrow [0,\infty)$ which is not identically zero.

The modulus of the gradient of a Lipschitz function f on Ω (briefly $f \in Lip(\Omega)$) is defined by⁷

$$|\nabla f(x)| = \limsup_{d(x,y) \to 0} \frac{|f(x) - f(y)|}{d(x,y)}$$

⁶I.e. such that if f and g have the same distribution function, then $||f||_X = ||g||_X$ (see Section 2.2 below).

⁷In fact one can define $|\nabla f|$ for functions f that are Lipschitz on every ball in (Ω, d) (cf. [7] for more details).

The equivalence between isoperimetric inequalities and Poincaré inequalities was obtained by Maz'ya. Maz'ya's method (see [16], [62] and [70]) shows that given $X = X(\Omega)$ a rearrangement invariant space⁸, the inequality

$$\left\| f - \int_{\Omega} f d\mu \right\|_{X} \le c \left\| |\nabla f| \right\|_{L^{1}}, \ f \in Lip(\Omega),$$
(1.0.8)

holds if, and only if, there exists a constant $c = c(\Omega) > 0$ such that for all Borel sets $A \subset \Omega$

$$\min(\phi_X(\mu(A)), \phi_X(1-\mu(A))) \le c\mu^+(A), \tag{1.0.9}$$

where $\phi_X(t)$ is the fundamental function⁹ of X :

$$\phi_X(t) = \|\chi_A\|_X$$
, with $\mu(A) = t$.

Motivated by this fact, we will say (Ω, d, μ) admits a concave isoperimetric estimator if there exists a function $I : [0, 1] \rightarrow [0, \infty)$ which is continuous, concave, increasing on (0, 1/2), symmetric about the point 1/2, such that I(0) = 0 and I(t) > 0 on (0, 1), and satisfies

$$I_{\mu}(t) \ge I(t), \ 0 \le t \le 1.$$

In recent work, Milman and Martín (see [61], [63]) proved that (Ω, d, μ) admits a concave isoperimetric estimator I if, and only if, the following symmetrization inequality

$$f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \le \frac{t}{I(t)} |\nabla f|_{\mu}^{**}(t), \ (f \in Lip(\Omega))$$
(1.0.10)

where $f_{\mu}^{**}(t) = \frac{1}{t} \int_{0}^{t} f_{\mu}^{*}(s) ds$, and f_{μ}^{*} is the non-increasing rearrangement of f with respect to the measure μ . If we apply a rearrangement invariant function norm X on Ω (see Sections 2.1 and 2.2 below) to (1.0.10), we obtain Sobolev–Poincaré type estimates of the form¹⁰

$$\left\| \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \right) \frac{I(t)}{t} \right\|_{\bar{X}} \le \left\| |\nabla f|_{\mu}^{**} \right\|_{\bar{X}}.$$
 (1.0.11)

To see how the isoperimetric profile helps to determine the correct spaces, consider the basic model cases (see [64], [65]).

Let $\Omega \subset \mathbb{R}^n$ be a Lipschitz domain of measure 1, $X = L^p(\Omega)$, $1 \le p \le n$, and p^* be the usual Sobolev exponent defined by $\frac{1}{p^*} = \frac{1}{p} - \frac{1}{n}$. Then

$$\left\| (f^{**}(t) - f^{*}(t)) \frac{I(t)}{t} \right\|_{L^{p}} \simeq \left\| (f^{**}(t) - f^{*}(t)) \right\|_{L^{p^{*,p}}}, \qquad (1.0.12)$$

⁸i.e. such that if f and g have the same distribution function then $||f||_X = ||g||_X$ (see Section 2.2 below).

⁹We can assume with no loss of generality that ϕ_X is concave (see Section 2.2.1 below).

¹⁰ \bar{X} denotes the representation space of X (see Section 2.2 below).

which follows from the fact that the isoperimetric profile is equivalent to $I(t) = c_n \min(t, 1-t)^{1-1/n}$, and Hardy's inequality (here $L^{p^*,p}$ is a Lorentz space (see Section 2.2 below)). In the case of \mathbb{R}^n with a Gaussian measure γ_n , and let $X = L^p$, $1 \le p < \infty$, then (compare with [23], [34]), since $I_{(\mathbb{R}^n,d,\gamma_n)}(t) \simeq t(\log 1/t)^{1/2}$ for t near zero, we have

$$\left\| \left(f_{\gamma_n}^{**}(t) - f_{\gamma_n}^*(t) \right) \frac{I(t)}{t} \right\|_{L^p} \simeq \left\| \left(f_{\gamma_n}^{**}(t) - f_{\gamma_n}^*(t) \right) \right\|_{L^p(Log)^{p/2}},$$
(1.0.13)

where $L^p(logL)^{p/2}$ is a Lorentz–Zygmund space (see Section 2.2).

In this fashion, in [61], [63], [64] and [65], Milman and Martín were able to provide a unified framework for studying the classical Sobolev inequalities and logarithmic Sobolev inequalities. Moreover, the embeddings (1.0.11) turn out to be the best possible in all the classical cases. However, the method used in the proof of these results cannot be applied to probability measures with heavy tails, since the isoperimetric estimators of such measures are convex, which means there exists a function $I:[0,1] \rightarrow [0,\infty)$ which is continuous, convex, increasing on (0,1/2), symmetric about the point 1/2, such that I(0) = 0 and I(t) > 0 on (0,1), and satisfying

$$I_{\mu}(t) \ge I(t), \ 0 \le t \le 1.$$



Figure 1.1: Isoperimetric profile

Therefore (unless $I(t) \simeq \min(t, 1-t)$), the Poincaré inequality

$$\left\|f - \int_{\Omega} f d\mu\right\|_{L^{1}} \le c \, \||\nabla f|\|_{L^{1}}, f \in Lip(\Omega),$$

never holds, which means that we cannot deduce from $|\nabla f| \in L^1$ that $f \in L^1$. Hence, a symmetrization inequality like (1.0.10) will not be possible, since f_{μ}^{**} is defined if, and only if, $f \in L^1$.

Chapter 4 is organized as follows. In Section 4.2 we obtain symmetrization inequalities which incorporate in their formulation the isoperimetric convex estimator. In Section 4.3 we use the symmetrization inequalities to derive Sobolev–Poincaré and Nash type inequalities. Finally in Section 4.4 we study in detail

several examples, such as, an α -Cauchy type law (the example 4.1.2), extended *p*-sub-exponential laws (the example 4.1.3), and *n*-dimensional weighted Riemannian manifolds that satisfy the CD(0, N) curvature condition with N < 0 (the example 4.1.2).

The results contained in that chapter 4 have been submitted for publication (see [68]).

Chapter 2 Preliminaries

In this chapter, we present the basic notation we shall use in the following chapters and briefly review some basic facts from the theory of rearrangement invariant spaces. We refer the reader to [6],[29],[53] or [86] for a complete treatment.

Throughout what follows we will work on a measure space (Ω, μ) with a separable, non-atomic, Borel measure μ . Let $\mathcal{M}(\Omega)$ be the set of all extended real-valued measurable functions on Ω . By $\mathcal{M}_0(\Omega)$ we denote the class of functions in $\mathcal{M}(\Omega)$ that are finite μ -a.e.

As usual, if $E \subset \Omega$ is μ -measurable, then, for $1 \leq p < \infty$, $L^p(E)$ is the space of μ -measurable functions f such that the norm $||f||_{L^p(A)} = (\int_A |f|^p d\mu)^{1/p}$ is finite. We define $L^{\infty}(E)$ similarly, but using $||f||_{L^{\infty}(A)} = ess \sup_A |f|$. $L^p_{loc}(\Omega)$ will denote functions that are p-integrable on balls.

The symbol $f \simeq g$ will indicate the existence of a universal constant c > 0 (independent of all parameters involved), thus $c^{-1}f \leq g \leq cf$, while $f \leq g$ means that $f \leq cg$.

2.1 Decreasing rearrangement

The distribution function μ_f of a function f in $\mathcal{M}_0(\Omega)$ is defined by

$$\mu_f(t) = \mu\{x \in \Omega : f(x) > t\} \quad (t \in \mathbb{R}).$$

In the literature it is common to denote the distribution function of |f| by μ_f , while here it is denoted by $\mu_{|f|}$ since we need to distinguish between the distribution function of f and that of |f|.

Two functions f and $g \in \mathcal{M}_0(\Omega)$ are said to be **equimeasurable** if $\mu_{|g|}(t) = \mu_{|f|}(t)$ for $t \ge 0$.

The signed decreasing rearrangement of a function $f \in \mathcal{M}_0(\Omega)$ $f_{\mu}^{\bigstar} : [0, \mu(\Omega)) \to \mathbb{R}$ is defined by

$$f_{\mu}^{\bigstar}(t) = \inf \left\{ s \in \mathbb{R} : \mu \{ x \in \Omega : \mu_f(x) > s \} \le t \right\}, \ t \in [0, \mu(\Omega)).$$

It follows readily from the definition that f^{\star}_{μ} is decreasing and that

$$(f+g)^{\bigstar}_{\mu}(t) \le f^{\bigstar}_{\mu}\left(\frac{t}{2}\right) + g^{\bigstar}_{\mu}\left(\frac{t}{2}\right), \quad (t>0).$$

Moreover,

$$f^{\bigstar}_{\mu}(0^+) = ess \sup f \quad \text{and} \quad f^{\bigstar}_{\mu}(\infty) = ess \inf f.$$
 (2.1.1)

The **decreasing rearrangement** f^*_{μ} of f is given by

$$f^*_{\mu}(t) = |f|^{\bigstar}_{\mu}(t)$$

In the next proposition we establish some basic properties of the decreasing rearrangement.

Proposition 2.1.1. Let f, g, f_i (i = 1, 2, ...,) belong to $\mathcal{M}_0(\Omega)$ and $\alpha \in \mathbb{R}$. Then (*i.*) $f_{\mu}^{\ddagger}(\mu_f(t)) \leq t$ for all $t \geq 0$ with $\mu_f(t) < \infty$;

- (*ii.*) $\mu_f(f_{\mu}^{\bigstar}(t)) \leq t \text{ for all } t \geq 0 \text{ with } f_{\mu}^{\bigstar}(t) < \infty;$
- (iii.) If $f \leq g$, then $f_{\mu}^{\bigstar} \leq g_{\mu}^{\bigstar}$;
- (*iv.*) $(\alpha f)^{\alpha}_{\mu} = \alpha f^{\alpha}_{\mu}$ and $(f + \alpha)^{\alpha}_{\mu} = f^{\alpha}_{\mu}(t) + \alpha;$
- (v.) If $|f_i| \uparrow |f|$, then $(f_i)^*_{\mu} \uparrow f^*_{\mu}$;
- (vi.) f^{\star}_{μ} is right continuous;
- (vii.) $f^*_{\mu}(s) = m_{\mu_{|f|}}(s), t \ge 0$ (where m denotes Lebesgue measure on $(0, \mu(\Omega))$;
- (viii.) f^{\bigstar}_{μ} and f^{\ast}_{μ} are equimeasurable with respect to Lebesgue measure on $(0, \mu(\Omega))$.

Example 2.1.1. Let f be a positive simple function, i.e.

$$f(x) = \sum_{j=1}^n b_j \chi_{F_j}(x),$$

where the coefficients b_j are positive and $F_j = \{x \in \Omega : f(x) = b_j\}.$

The distribution function is given by

$$\mu_f(\lambda) = \sum_{j=1}^n m_j \chi_{[b_{j+1}, b_j]}(\lambda),$$

where $m_j = \sum_{i=1}^{j} \mu(F_i)$, (j = 1, 2, ..., n) and $b_{n+1} = 0$ (see Figure 2.1). The decreasing rearrangement is given by

$$f^*_{\mu}(t) = \sum_{j=1}^n b_j \chi_{[0,m_j)}(t)$$



Figure 2.1: Graphs of f and μ_f



Figure 2.2: Graph of f_{μ}^{\star}

Example 2.1.2. This example shows how signed rearrangement works:



Figure 2.3: Graph of $f_{\mu}^{\,\bigstar}$

For any measurable set $E \subset \Omega$ and $f \in \mathcal{M}_0(\Omega)$,

$$\int_{E} |f(x)| d\mu \le \int_{0}^{\mu(E)} f_{\mu}^{*}(s) ds.$$
(2.1.2)

In fact,

$$\sup_{\mu(E)=t} \int_{E} |f(x)| d\mu = \int_{0}^{t} f_{\mu}^{*}(s) ds$$
(2.1.3)

and

$$\int_0^t f_{\mu}^{\bigstar}(s) ds = \sup\left\{\int_E f(s) d\mu : \mu(E) = t\right\}, \quad (t > 0).$$
(2.1.4)

The signed maximal function $f_{\mu}^{\,\pm\,\pm}$ is defined by

$$f_{\mu}^{\bigstar \bigstar} = \frac{1}{t} \int_0^t f_{\mu}^{\bigstar}(s) ds, \quad (t > 0).$$

Similarly, f_{μ}^{**} will denote the **maximal function** of f_{μ}^{*} defined by

$$f_{\mu}^{**}(t) = \frac{1}{t} \int_0^t f_{\mu}^*(s) ds, \quad (t > 0).$$

Some elementary properties of the maximal signed function are listed below.

Proposition 2.1.2. Let f, g and f_i (i = 1, 2, ...,) belong to $\mathcal{M}_0(\Omega)$ and $\alpha \in \mathbb{R}$. Then

- $(i.) \ f_{\mu}^{\,\bigstar} \leq f_{\mu}^{\,\bigstar\,\bigstar};$
- (*ii.*) If $f \leq g$, then $f_{\mu}^{\bigstar \bigstar} \leq g_{\mu}^{\bigstar \bigstar}$;
- (*iii.*) $(\alpha f)^{\bigstar \bigstar}_{\mu} = \alpha f^{\bigstar \bigstar}_{\mu};$
- $(iv.) \ (f+g)_{\mu}^{\bigstar \bigstar}(t) \leq f_{\mu}^{\bigstar \bigstar}(t) + g_{\mu}^{\bigstar \bigstar}(t), \ (t>0).$

Example 2.1.3. Let $\Omega = [0, \infty)$ and μ be Lebesgue measure on Ω . Define $f : [0, \infty) \rightarrow [0, \infty)$ by

$$f(x) = \begin{cases} 1 - (x - 1)^2 & if \ 0 \le x \le 2\\ 0 & if \ x > 2. \end{cases}$$

The distribution function can be easily computed:

$$\mu_f(\lambda) = \begin{cases} 2\sqrt{1-\lambda} & if \ 0 \le \lambda \le 1\\ 0 & if \ \lambda > 1, \end{cases}$$

and the decreasing rearrangement becomes

$$f_{\mu}^{*}(t) = \begin{cases} 1 - \frac{t^{2}}{4} & if \ 0 \le t \le 2\\ 0 & if \ t > 2, \end{cases}$$

Moreover,



The maximal function is given by

$$f_{\mu}^{**}(t) = \begin{cases} 1 - \frac{t^2}{12} & if \ 0 < t \le 2\\ \frac{4}{3t} & if \ t > 2 \end{cases}$$



Figure 2.5: Graph f_{μ}^{**}

Definition 2.1.1. Let f belong to $\mathcal{M}_0(\Omega)$. The oscillation of f^*_{μ} is defined by the special difference

$$O_{\mu}(f,t) = f_{\mu}^{**}(t) - f_{\mu}^{*}(t).$$

The functional $O_{\mu}(f,t)$ has certain technical disadvantages. It vanishes on constant functions and the operation $f \to O_{\mu}(f,t)$ is not subadditive.

Lemma 2.1.1. Let f belong to $\mathcal{M}_0(\Omega)$. Then

$$\frac{\partial}{\partial t} f_{\mu}^{**}(t) = -\frac{O_{\mu}(f,t)}{t}, \quad t > 0,$$
(2.1.5)

and the function $tO_{\mu}(f,t)$ is increasing in t.

Proof. By the definition of f_{μ}^{**} , and a simple computation, we get

$$\begin{aligned} \frac{\partial}{\partial t} f_{\mu}^{**}(t) &= \frac{\partial}{\partial t} \left(\frac{1}{t} \int_{0}^{t} f_{\mu}^{*}(s) \right) ds \\ &= -\frac{1}{t^{2}} \int_{0}^{t} f_{\mu}^{*}(s) ds + \frac{1}{t} f_{\mu}^{*}(t) \\ &= -\frac{1}{t} \left(\frac{1}{t} \int_{0}^{t} f_{\mu}^{*}(s) ds - f_{\mu}^{*}(t) \right) \\ &= -\frac{1}{t} \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \right). \end{aligned}$$

Using the fact that (see [14])

$$O_{\mu}(f,t) = \frac{1}{t} \int_{f_{\mu}^{*}(t)}^{\|f\|_{\infty}} \mu_{|f|}(s) ds, \qquad (2.1.6)$$

it follows that $tO_{\mu}(f,t)$ is increasing. Indeed, to see 2.1.6, let $[x]^+ = max(x,0)$. Then, for all y > 0, we have that

$$\int_0^\infty [f_\mu^*(x) - y]^+ dx = \int_0^\infty \mu_{[f_\mu^* - y]^+}(s) ds = \int_y^\infty \mu_{f_\mu^*}(s) ds = \int_y^{||f||_\infty} \mu_{|f|}(s) ds.$$
(2.1.7)

Inserting $y = f_{\mu}^{*}(t)$ in 2.1.7 and taking into account that f_{μ}^{*} is decreasing, we get

$$tO_{\mu}(f,t) = t(f_{\mu}^{**}(t) - f_{\mu}^{*}(t))$$

= $\int_{0}^{t} (f_{\mu}^{*}(x) - f_{\mu}^{*}(t))dx$
= $\int_{0}^{\infty} [f_{\mu}^{*}(x) - f_{\mu}^{*}(t)]^{+}dx$
= $\int_{f_{\mu}^{*}(x)}^{||f||_{\infty}} \mu_{|f|}(s)ds.$

Conditions like $f^*_{\mu}(\infty) = 0$ will appear often. The following proposition clarifies the significance of such conditions.

Proposition 2.1.3 (See [57]). If $\mu(\Omega) = \infty$, then $f^*_{\mu}(\infty) = 0$ if, and only if, $\mu_f(t)$ is finite for any t > 0

Proof. Suppose that $\mu_f(t_0) = \infty$ for some $t_0 > 0$. From the definition of rearrangement, we have that $f^*_{\mu}(t) \ge t_0$ for all t > 0.

Therefore the condition $f^*_{\mu}(\infty) = 0$ implies $\mu_f(t) < \infty$, for all t > 0.

Conversely, assume $f_{\mu}^{*}(t) \geq \varepsilon > 0$. This means that $\mu_{f}(\varepsilon) = \infty$. Thus the condition $\mu_{f}(t) < \infty, t > 0$ implies $f_{\mu}^{*}(\infty) = 0$.

Note that an application of L'Hôpital's rule to f_{μ}^{**} shows that the condition $f_{\mu}^{*}(\infty) = 0$ is equivalent to $f_{\mu}^{**}(\infty) = 0$.

Remark 2.1.1. By (2.1.5) and the Fundamental Theorem of Calculus, and using $f_{\mu}^{**}(\infty) = 0$, we have

$$f_{\mu}^{**}(t) = \int_{t}^{\infty} \frac{O_{\mu}(f,s)}{s} ds, \quad t > 0.$$

2.2 Rearrangement invariant spaces

Rearrangement invariant spaces play an important role in contemporary mathematics. They have many applications in various branches of analysis, including the theory of function spaces, interpolation theory, mathematical physics, and probability theory.

Definition 2.2.1. A function space $X(\Omega)$ is the linear space of all $f \in \mathcal{M}_0(\Omega)$ for which $||f||_{X(\Omega)} < \infty$, where $||\cdot||_{X(\Omega)}$ is a functional norm, i.e.

- (i.) $\|\cdot\|_{X(\Omega)}$ is a norm;
- (*ii.*) if $0 < g \le f$ a.e., then $||g||_{X(\Omega)} \le ||f||_{X(\Omega)}$;
- (*iii.*) if $0 < f_j \uparrow f$ a.e., then $||f_j||_{X(\Omega)} \uparrow ||f||_{X(\Omega)}$;
- (iv.) for any measurable set $E \subset \Omega$, $\|\chi_E\|_{X(\Omega)} < \infty$; and
- (v.) $\int_E |f(x)| dx \le ||f||_{X(\Omega)}$.

If, in the definition of a norm, the triangle inequality is weakened to the requirement that for some constant C_X where

$$||x + y||_X \le C_X(||x||_X + ||y||_X)$$

holds for all x and y, then we have a quasi-norm. A complete quasi-normed space is called a quasi-Banach space.

Definition 2.2.2. A Banach function space (resp. quasi-Banach function space) $X = X(\Omega)$ is called a rearrangement invariant (r.i.) space (resp. quasi-Banach rearrangement invariant (q.r.i.) space) if $g \in X$ implies that all μ -measurable functions f with the same rearrangement function with respect to the measure μ , i.e. such that $f_{\mu}^* = g_{\mu}^*$, also belong to X, and moreover $||f||_X = ||g||_X$.

For any r.i. space $X(\Omega)$ we have

$$L^{1}_{\mu}(\Omega) \cap L^{\infty}_{\mu}(\Omega) \subset X(\Omega) \subset L^{1}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega), \qquad (2.2.1)$$

with continuous embedding, where the space $L^{1}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$ consist of all functions $f \in \mathcal{M}_{0}(\Omega)$ that are representable as a sum f = g + h of functions $g \in L^{1}_{\mu}(\Omega)$ and $h \in L^{\infty}_{\mu}(\Omega)$. The norm in $L^{1}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$ is given by

$$||f||_{L^{1}_{\mu}(\Omega)+L^{\infty}_{\mu}(\Omega)} = \inf\{||g||_{L^{1}_{\mu}(\Omega)}+||h||_{L^{\infty}_{\mu}(\Omega)}\},\$$

where the infimum is taken over all representations f = g + h of the kind described above.

Consider the following norm on $L^1_{\mu}(\Omega) \cap L^{\infty}_{\mu}(\Omega)$:

$$||f||_{L^{1}_{\mu}(\Omega)\cap L^{\infty}_{\mu}(\Omega)} = \max\{||g||_{L^{1}_{\mu}(\Omega)}, ||h||_{L^{\infty}_{\mu}(\Omega)}\}.$$

If $\mu(\Omega) < \infty$, we obviously have

$$L^{\infty}_{\mu}(\Omega) \subset X(\Omega) \subset L^{1}_{\mu}(\Omega).$$

The associate space $X'(\Omega)$ of $X(\Omega)$ is the r.i. space of all $h \in \mathcal{M}_0(\Omega)$ for which the r.i. norm given by

$$\|h\|_{X'(\Omega)} = \sup_{\|g\| \le 1} \int_{\Omega} |g(x)h(x)| \, d\mu \tag{2.2.2}$$

is finite.

Therefore the following generalized Hölder inequality holds

$$\int_{\Omega} |g(x)h(x)| \, d\mu \le \|g\|_{X(\Omega)} \, \|h\|_{X'(\Omega)} \, .$$

A useful property states that if

$$\int_0^r f_\mu^*(s) ds \le \int_0^r g_\mu^*(s) ds,$$

for all r > 0, then for any r.i. space $X = X(\Omega)$ we have

 $\|f\|_X \le \|g\|_X$.

Let $X(\Omega)$ be an r.i. space. Then there exists a unique r.i. space (see [6, Theorem 4.10 and subsequent remarks]) $\bar{X} = \bar{X}(0, \mu(\Omega))$ on $((0, \mu(\Omega)), m)$, (m denotes the Lebesgue measure on the interval $(0, \mu(\Omega))$), such that

$$f \in X(\Omega) \Leftrightarrow f^*_{\mu} \in \bar{X}(0, \mu(\Omega)),$$

and furthermore,

$$\|f\|_{X(\Omega)} = \|f_{\mu}^{*}\|_{\bar{X}(0,\mu(\Omega))} = \|f_{\mu} \nleftrightarrow \|_{\bar{X}(0,\mu(\Omega))}.$$

 \overline{X} is called the **representation space** of $X(\Omega)$. The norm of $\overline{X}(0,\mu(\Omega))$ is given explicitly by

$$\|h\|_{\bar{X}(0,\mu(\Omega))} = \sup\left\{\int_0^{\mu(\Omega)} h^*(s)g^*_{\mu}(s)ds; \|g\|_{X'(\Omega)} \le 1\right\},$$

where the first rearrangement is taken with respect to the Lebesgue measure on $[0, \mu(\Omega))$.

2.2.1 Indices

We will present some definitions and properties related to indices (see [10] and [107]).

The dilation operator is defined by

$$D_{\frac{1}{s}}f(t) = \begin{cases} f_{\mu}^{*}\left(\frac{t}{s}\right) & 0 < t < s, \\ 0 & s < t < \mu(\Omega) \end{cases}$$

For each t > 0, we denote by $h_X(s)$ the norm of $D_{\frac{1}{s}}$, i.e.

$$h_X(s) = \sup_{f \in X} \frac{\left\| D_{\frac{1}{s}} f \right\|_X}{\|f\|_X}, \ s > 0$$

The upper and lower Boyd indices associated with an r.i. space X are defined by

$$\overline{\alpha}_X = \inf_{s>1} \frac{\ln h_X(s)}{\ln s} \quad \text{and} \quad \underline{\alpha}_X = \sup_{s<1} \frac{\ln h_X(s)}{\ln s}.$$
(2.2.3)

The **fundamental function** of X an r.i. space is defined by

$$\phi_X(s) = \|\chi_E\|_X,$$

where E is any measurable subset of Ω with $\mu(E) = t$.

Note that the particular choice of E is immaterial, due to the rearrangement invariance of X. We can assume without loss of generality that ϕ_X is concave (see [6]). Moreover, by Hölder's inequality,

$$\phi_{X'}(t)\phi_X(t) = t. \tag{2.2.4}$$

It is also useful sometimes to consider a slightly different set of indices obtained by means of replacing $h_X(s)$ in (2.2.3) by

$$M_X(s) = \sup_{t>0} \frac{\phi_X(ts)}{\phi_X(t)}, \ s > 0.$$

The corresponding indices are denoted by $\overline{\beta}_X$, $\underline{\beta}_X$, and will be referred to as the upper and lower fundamental indices of X. Actually, the relationship between $M_X(s)$ and $h_X(s)$ is that the computation of the former is exactly the computation of the latter but done only over functions of the form $f = \chi_{(0,a)}$. Therefore we have

$$0 \leq \underline{\alpha}_X \leq \beta_X \leq \beta_X \leq \overline{\alpha}_X \leq 1.$$

Lemma 2.2.1. (See [60] and [107]) Let Y be an r.i. space on (0,1). Let ϕ_Y be its fundamental function. Assume that $\phi_Y(0) = 0$. Then

- 1. If $\overline{\alpha}_Y < 1$, then for every $\overline{\alpha}_Y < \gamma < 1$, the function $\phi_Y(s)/s^{\gamma}$ is almost decreasing (i.e. $\exists c > 0 \ s.t. \ \phi_Y(s)/s^{\gamma} \le c\phi_Y(t)/t^{\gamma}$ whenever $t \le s$).
- 2. If $\underline{\alpha}_Y > 0$, then for every $0 < \gamma < \underline{\alpha}_Y$, the function $\phi_Y(s)/s^{\gamma}$ is almost increasing (i.e. $\exists c > 0 \ s.t. \ \phi_Y(s)/s^{\gamma} \le c\phi_Y(t)/t^{\gamma}$ whenever $t \ge s$).
- 3. If $\underline{\alpha}_{Y} > 0$, there exists a concave function $\hat{\phi}_{Y}$ and constant c > 0 such that

$$\hat{\phi}_Y(t) \simeq \phi_Y(t)$$
 and $c^{-1}\phi_Y(t)/t \le \frac{\partial}{\partial t}\hat{\phi}_Y(t) \le c\phi_Y(t)/t.$

We shall usually formulate conditions on r.i. spaces in terms of the Hardy operators defined by

$$(P_a f)(t) = t^{-a} \int_0^t x^a f(x) \frac{dx}{x}; \quad (Q_a f)(t) = t^{-a} \int_t^{\mu(\Omega)} x^a f(x) \frac{dx}{x}$$

for $t \in (0, \mu(\Omega))$ and $a \in (0, 1]$, $f \in \mathcal{M}_0((0, \mu(\Omega)))$.

It is known that X is an r.i. (resp. q.r.i) space, that P_a is bounded if, and only if, $\overline{\alpha}_X < a$, and that Q_a is bounded if, and only if, $a < \underline{\alpha}_X$ (see [6]).

2.2.2 Examples

In this section we present some examples of rearrangement invariant spaces.

• The L^p_{μ} -spaces consist of all $f \in \mathcal{M}_0(\Omega)$ for which

$$\|f\|_{L^p_{\mu}} \coloneqq \begin{cases} \left(\int_0^{\mu(\Omega)} \left(f^*_{\mu}(t) \right)^p dt \right)^{1/p}, & 0 0} f^*_{\mu}(t), & p = \infty, \end{cases}$$

is finite. Its fundamental function is given by

$$\phi_{L^p_{\mu}}(t) = \begin{cases} t^{1/p}, & 1 \le p \le \infty, \quad t > 0\\ 0, & p = \infty, & t = 0\\ 1, & p = \infty, & t > 0. \end{cases}$$

 $\begin{array}{l} L^p_\mu \text{ is q.r.i. if } 0$

• Assume that $0 < p, q \leq \infty$. The Lorentz spaces $L^{p,q}_{\mu}$ consist of all $f \in \mathcal{M}_0(\Omega)$ for which

$$\|f\|_{L^{p,q}_{\mu}} \coloneqq \left\{ \begin{array}{ll} \left(\int_{0}^{\mu(\Omega)} \left(t^{1/p} f^{*}_{\mu}(t) \right)^{q} dt \right)^{1/q}, & 0 < q < \infty \\ \sup_{t > 0} \left\{ t^{1/p} f^{*}_{\mu}(t) \right\}, & q = \infty, \end{array} \right.$$

is finite. Its fundamental function is given by

$$\phi_{L^{p,q}_{\mu}}(t) = t^{1/p}, \quad t > 0,$$

where $1 \leq p < \infty$. The Boyd indices are given by $\underline{\alpha}_{L^{p,q}_{\mu}} = \overline{\alpha}_{L^{p,q}_{\mu}} = \frac{1}{p}$.

- Lorentz Λ spaces are defined by the functional

$$\|f\|_{\Lambda^q(v)} = \left(\int_0^{\mu(\Omega)} f^*_{\mu}(s)^q v(s) ds\right)^{1/q},$$

where $0 < q < \infty$ and v is a weight decreasing $(v \ge 0$ measurable function) on $(0, \mu(\Omega))$.

By choosing $v(s) = s^{q/p-1}$ one obtains $\Lambda^q(v) = L^{p,q}_{\mu}$. If we take $v(s) = s^{q/p-1}(1 + \log \frac{1}{s})^{\alpha}$, then the $\Lambda^q(v) = L^{p,q}_{\mu}(\log L)^{\alpha}$ are the **Lorentz–Zygmund spaces**; or if we take $v(s) = s^{q/p1}(1 + \log \frac{1}{s})^{\alpha}(1 + \log \frac{1}{s})^{\beta}$, then the $\Lambda^q(v) = L^{p,q}_{\mu}(\log L)^{\alpha}(\log \log L)^{\beta}$ are the generalized Lorentz– Zygmund spaces. **Definition 2.2.3.** A function $A: [0, \infty) \rightarrow [0, \infty]$ is a Young function if

$$A(s) = \int_0^s a(t) \, dt, \qquad (2.2.5)$$

where $a : [0, \infty] \rightarrow [0, \infty]$ is an increasing, left continuous function which is neither identically zero nor identically infinite on $(0, \infty)$ and which satisfies a(0) = 0. A Young function is convex on the interval where it is finite.

• For a Young function A, the **Orlicz space** L^A_μ is the collection of all functions $f \in \mathcal{M}_0(\Omega)$ for which there exists a λ such that

$$\int_{\Omega} A\left(\frac{|f(x)|}{\lambda}\right) d\mu < \infty.$$

The Orlicz space L^A_{μ} is endowed with the Luxemburg norm

$$\|f\|_{L^A_{\mu}} = \inf\left\{\lambda > 0; \int_{\Omega} A\left(\frac{|f(x)|}{\lambda}\right) d\mu \le 1\right\}.$$

The fundamental function for L^A_{μ} is given by

$$\phi_{L^A_u}(t) = 1/A^{-1}(1/t), \qquad (2.2.6)$$

where A is a Young function on $(0, \infty)$. On the other hand (see [6]), we have

$$\int_{\Omega} A\left(|f(x)|\right) d\mu = \int_{0}^{\mu(\Omega)} A(f_{\mu}^{*})(t) dt.$$

Given an r.i. space X over $(0, \mu(\Omega))$, suppose X has been renormed so as to have a concave fundamental function ϕ_X . There are some useful **Lorentz** and **Marcinkiewicz** spaces, defined by the quasi-norms

$$\|f\|_{M(X)} = \sup_{t} f_{\mu}^{**}(t)\phi_{X}(t), \quad \|f\|_{\Lambda(X)} = \int_{0}^{\mu(\Omega)} f_{\mu}^{*}(t)d\phi_{X}(t).$$
(2.2.7)

It follows readily that

$$\phi_{M(X)}(t) = \phi_{\Lambda(X)}(t) = \phi_X(t).$$

Furthermore

$$\Lambda(X) \subset X \subset M(X), \tag{2.2.8}$$

and each of the embeddings has norm 1.

Chapter 3

A Sobolev type embedding theorem for Besov spaces defined on doubling metric spaces

3.1 Introduction

Analysis on metric measure spaces has been studied quite intensively in recent years; see, for example, Semmesś survey [87] for a more detailed discussion and references. A field of particular interest is the study of functional inequalities, like Sobolev and Poincaré inequalities on metric measure spaces; see, for example, [56],[36],[46], [32],[33],[54]. Since Hajłasz in [35] introduced Sobolev spaces on any metric measure space, a series of papers has been devoted to the construction and investigation of Sobolev spaces of various types on metric measure spaces; see, for example, [36],[46],[37],[24]. Recently, a theory of Besov spaces was developed in [38] which is a generalization of the corresponding theory of function spaces on \mathbb{R}^n (see [94],[95],[96]) respectively Ahlfors *n*-regular metric measure spaces (see [39],[41]).

There are several equivalent ways to define Besov spaces in the setting of a doubling metric space (see for example [27], [28], [38], [75], [74], [105], [45] and the references therein), in this chapter, we shall use the approach based on a generalization of the classical L^p -modulus of smoothness introduced in [27].

Recall that the L^p -modulus of smoothness of a function $f \in L^p_{loc}(\mathbb{R}^n)$ is defined by

$$\omega_p(f,t) \coloneqq \sup_{|h| \le t} \|f(x+h) - f(x)\|_{L^p(\mathbb{R}^n)},$$

where t > 0 and |h| is the Euclidean length of the vector h. As a general metric
space possesses no group structure, a modification to this definition is needed.

Let (Ω, d, μ) be a metric measure space equipped with a metric d and a Borel regular outer measure μ for which the measure of every ball is positive and finite. Given t > 0, $0 and <math>f \in L^p_{loc}(\Omega)$, the L^p -modulus of smoothness is defined by

$$E_p(f,t) = \left(\int_{\Omega} \left(\int_{B(x,t)} |f(x) - f(y)|^p \, d\mu(y) \right) d\mu(x) \right)^{1/p}, \tag{3.1.1}$$

. .

where $f_B f(x) d\mu(x) \coloneqq \frac{1}{\mu(B)} \int_B f(x) d\mu(x)$ is the integral average of a locally integrable function f over B.

 $E_p(f,t)$ is equivalent to the classical $L^p(\mathbb{R}^n)$ -modulus of smoothness of a function $f \in L^p_{loc}(\mathbb{R}^n)$. Indeed (see [27]),

$$E_{p}(f,t) = \left(\int_{\mathbb{R}^{n}} \left(\int_{B(x,t)} |f(x) - f(y)|^{p} \, dy \right) dx \right)^{1/p}$$

$$= \left(\int_{\mathbb{R}^{n}} \left(\int_{B(0,t)} |f(x+h) - f(x)|^{p} \, dh \right) dx \right)^{1/p}$$

$$\approx \sup_{|h| \le t} \|f(x+h) - f(x)\|_{L^{p}(\mathbb{R}^{n})} \coloneqq \omega_{p}(f,t), \quad (\text{see [55]}).$$
(3.1.2)

Definition 3.1.1. For $0 < s < \infty$, the homogeneous Besov space $\dot{\mathcal{B}}^s_{p,q}(\Omega)$ consists of those functions $f \in L^p_{loc}(\Omega)$ for which the seminorm

$$\|f\|_{\dot{\mathcal{B}}^s_{p,q}(\Omega)} \coloneqq \begin{cases} \left(\int_0^\infty \left(\frac{E_p(f,t)}{t^s}\right)^q \frac{dt}{t}\right)^{1/q}, & 0 < q < \infty, \\ \sup_{t>0} t^{-s} E_p(f,t), & q = \infty, \end{cases}$$

is finite.

This definition is rather concrete and by (3.1.2) gives the usual Besov space in the Euclidean setting. Moreover, it has been shown by Müller and Yang [74] that it coincides with the definition based on test functions and used earlier by Han [40], Han and Yang [42] and Yang [103], provided that Ω , besides being doubling, also satisfies a reverse doubling condition

In the Euclidean setting, the Sobolev embedding theorem states (see, for example, [74, Theorem. 1.15]) that there is a constant C > 0 such that

$$\|f\|_{L^{p^*,q}_{\mu}(\mathbb{R}^n)} \le C \|f\|_{\dot{B}^s_{p,q}(\mathbb{R}^n),}$$

where $p^* = np/(n - sp)$, and the Lorentz space $L^{p,q}(\mathbb{R}^n)$, consist of measurable functions f of finite norm

$$\|f\|_{L^{p,q}_{\mu}(\mathbb{R}^n)} = \left\|t^{\frac{1}{p}-\frac{1}{q}}f^*_{\mu}(t)\right\|_{L^q([0,\infty))},$$

 $(f_{\mu}^{*}$ denotes the decreasing rearrangement of f, see Section 2.1).

The abstract variant for metric spaces is only known in the following particular case (see [27] and [45]):

Theorem 2. Let Ω be a Q-regular metric space, i.e. there exists a $Q \ge 1$ and a constant $c_Q \ge 1$ such that

$$c_Q^{-1}r^Q \le \mu(B(x,r)) \le c_Q r^Q$$

for each $x \in X$, and for all $0 < r < diam \Omega$ (here diam Ω is the diameter of Ω). Suppose that 0 < s < 1 and $1 \le q \le \infty$. Then:

1. (See ([27, Thm. 5.1])) Suppose Ω satisfies a (1, p)-Poincare inequality, i.e. there exist constants $C_p \ge 0$ and $\lambda \ge 1$ such that

$$\int_{B} |f - f_B| d\mu \leq \left(\int_{\lambda B} g^p d\mu \right)^{1/p}$$

for any locally integrable function f for all upper gradients¹ g of f. Then, if 1 ,

$$\dot{\mathcal{B}}^{s}_{p,q}(\Omega) \subset L^{p(Q),q}_{\mu}(\Omega), \qquad (3.1.3)$$

where p(Q) = Qp/(Q - sp).

2. (See ([45, Thm. 4.4])) Suppose that Ω is geodesic, i.e. every pair of points can be joined by a curve whose length is equal to the distance between the points. Then (3.1.3) holds for $1 \le p < Q/s$.

The proof of this theorem is based on the real interpolation method; for example, in ([27, Thm. 5.1]) under a (1, p)-Poincaré inequality assumption, the Besov space $\dot{\mathcal{B}}^s_{p,q}(\Omega)$ is realized as the real interpolation space $(L^p(\Omega), KS_{1,p}(\Omega))_{\alpha,q}$ between the corresponding $L^p(\Omega)$ and the Sobolev space of Korevaar and Schoen $KS_{1,p}(\Omega)$, consisting of measurable functions f of finite norm²

$$\|f\|_{KS_{1,p}(\Omega)} \coloneqq \limsup_{t \to 0} \frac{E_p(f,t)}{t}.$$
(3.1.4)

In [27] it is proved that $E_p(f,t)$ is equivalent to the K-functional between $L^p(\Omega)$ and $KS_{1,p}(\Omega)$. Moreover if Ω is Q-regular, then

$$\|f\|_{L^{Qp}_{\mu}(\Omega)} \leq \|f\|_{KS_{1,p}(\Omega)}, \qquad (3.1.5)$$

and, consequently, interpolation allows one to obtain embedding theorems.

¹ For the definition of upper gradient, see, for example [46], [27].

²When Ω is a Riemannian manifold, this definition yields the usual Sobolev space and the quantity in (3.1.4) is equivalent to the usual semi-quasinorm (see [56]).

The key point in the previous argument is the embedding (3.1.5), which is only known for *Q*-regular spaces.

The purpose of this chapter is to obtain a Sobolev type embedding result for Besov spaces defined on a doubling metric space. In Theorem 2 we will see that if the embedding

$$\dot{\mathcal{B}}^{s}_{p,q}(\Omega) \subset L^{p^{\star},q}_{\mu}(\Omega)$$

holds for some $p^* > p$, then Ω does not have the collapsing volume property, i.e.

$$\inf_{x \in \Omega} \mu(B(x, 1)) > 0.$$
 (3.1.6)

In view of this result, we will need to limit the class of doubling spaces in which we going to work. Our framework will be (k, m)-spaces (see Section 3.2 below), i.e. there exist positive constants c_0 , C_0 , k, and m ($k \le m$) such that

$$c_0 \min(r^k, r^m) V_\mu(x, 1) \le V_\mu(x, r) \le C_0 \max(r^k, r^m) V_\mu(x, 1), \tag{3.1.7}$$

for all $x \in \Omega$ and $0 < r < \infty$, where $V_{\mu}(x, r) = \mu(B(x, r))$. To incorporate condition (3.1.6), we define

(i.) A (k,m)-space will be called **uniform** if there are constants c, C > 0 such that

$$c\min(r^k, r^m) \le V_\mu(x, r) \le C\max(r^k, r^m).$$
 (3.1.8)

(ii.) A (k, m)-space will be called **bounded from below** if there are constants d, D > 0 such that

$$d\min(r^k, r^m) \le V_{\mu}(x, r) \le D\max(r^k, r^m)V_{\mu}(x, 1).$$
(3.1.9)

In order to avoid using an embedding result like (3.1.5), which, recall, is only known for Q-regular metric spaces, we will obtain, for the above class of spaces, pointwise estimates between the oscillation of f^*_{μ} and the modulus of smoothness that will allow us to derive Sobolev type embedding results.

The results that we will obtain can be applied in a wide range of settings, for instance, to Ahlfors regular metric measure spaces (see [46]), Lie groups of polynomial volume growth (see [99],[100],[81],[76],[2]), Carnot–Carathéodory manifolds (see [81],[78],[79]), and to the boundaries of certain unbounded model domains of polynomial type in \mathbb{C}^N appearing in the work of Nagel and Stein (see [80],[81],[78],[79]) (see Section 3.2.1 below).

This chapter is organized as follows: Section 3.2 contain basic definitions and technical results on doubling metric measure spaces. In Section 3.3 we obtain pointwise estimates of the oscillation $O_{\mu}(f,t) = f_{\mu}^{**}(t) - f_{\mu}^{*}(t)$ in terms of the X-modulus of smoothness defined by

$$E_X(f,r) \coloneqq \left\| \oint_{B(x,r)} |f(x) - f(y)| \, d\mu(y) \right\|_X,$$

where X is a rearrangement invariant space on Ω . In Section 4.3 we define generalized Besov type spaces and use the oscillation inequalities obtained in the previous sections to derive embedding Sobolev theorems. In Section 3.5 we deal with generalized uncertainty Sobolev inequalities in the context of Besov spaces. In Section 3.6 a criterion for essential continuity and for the embedding into $BMO(\Omega)$ will be obtained. Finally, in Section 3.7 we will study in detail the case $\dot{\mathcal{B}}_{p,q}^{s}(\Omega)$ for 0 < s < 1, 0 and <math>0 .

The results contained in this chapter have been published in Journal of Mathematical Analysis and Applications (see [67]).

3.2 Doubling measures

In this section we briefly review basic facts about doubling metric spaces and some of their properties. The general framework will be a metric measure space (Ω, d, μ) with a metric d and a regular Borel measure μ for which the measure of every ball is positive and finite.

The ball with centre $x \in \Omega$ and radius r > 0 is defined by $B(x,r) = \{y \in \Omega : d(x,y) < r\}$. We shall denote by $V_{\mu}(x,r)$ the measure of the ball, i.e.

$$V_{\mu}(x,r) \coloneqq \mu(B(x,r)).$$

Definition 3.2.1. A metric measure space (Ω, d, μ) is called doubling if there exists a constant $C_D > 1$ such that for all $x \in \Omega$ and r > 0,

$$0 < V_{\mu}(x, 2r) \le C_D V_{\mu}(x, r) < \infty.$$
(3.2.1)

We will now present some examples.

Example 3.2.1.

- Let $\Omega = \mathbb{R}^n$, let d(x, y) = |x y| be the Euclidean metric, and let $\mu = \mathcal{L}^n$ be Lebesgue measure on Ω . Then $(\mathbb{R}^n, |.|, \mathcal{L}^n)$ is a doubling metric measure space with $C_D = 2^n$ (see [3], [46]).
- Let $\Omega = [-1, 0] \times [-1, -1] \cup [0, 1] \times \{0\}$, let d be the Euclidean metric, and let $\mu = \mathcal{L}^2|_{\Omega} + \mathcal{H}^1|_{[0,1] \times \{0\}}$, where \mathcal{H}^1 is the 1-dimensional Hausdorff measure. Then μ is doubling with $C_D = 4$ (see [3],[46]).
- Cantor sets (see [3]): Let H be a finite set having k points, $k \ge 2$, and

$$H^{\infty} = \{ x = (x_i)_{i \in \mathbb{N}} : x_i \in H \}.$$

Let $a \in (0,1)$. Then, $d_a : H^{\infty} \times H^{\infty} \to [0,\infty]$

$$d_a(x,y) = \begin{cases} 0, & \text{if } x = y \\ a^j, & \text{if } x_i = y_i \text{ for } i < j \text{ and } x_j \neq y_i \end{cases}$$

is a metric in H^{∞} . Let ν be a uniformly distributed probability measure on H. Define the measure μ on H^{∞} as the product measure of ν infinitely many times. In this case one can show that

$$\mu(B(x,a^j)) = k^{-j}$$

and that (H^{∞}, d_a, μ) is a doubling metric measure space with dimension s given by $a^s = k^{-1}$. If k = 2 and $a = \frac{1}{3}$, then H^{∞} is bi-Lipschitz equivalent to the standard $\frac{1}{3}$ -Cantor set.

• It is proved in [89], [90], and [58] that some curvature-dimension condition on metric measure spaces implies the doubling property of the considered measure.

Remark 3.2.1. Given $x \in \Omega$, the function $r \to \mu(B(x, r))$ is (usually) not continuous, thus given t > 0 there does not necessarily exist a ball B(x) centred at x such that $\mu(B(x)) = t$. However there is a ball B(x) centred at x such that $t/C_D \le \mu(B(x)) \le t$. Indeed, consider $r_0 = \sup \{r : V_\mu(x, r) < t/C_D\}$. Then

$$V_{\mu}(x,r) \leq t/C_D \leq V_{\mu}(x,2r) \leq C_D V_{\mu}(x,r) \leq t.$$

Following the proofs of [102, Theorem 1] and [92, Theorem 1.4], we obtain the following result:

Lemma 3.2.1. Let (Ω, d, μ) be a doubling measure space. Then, for all bounded subsets $A \subset \Omega$ with $\mu(A) > 0$, $x \in A$ and 0 < r < diam(A), we have

$$\frac{V_{\mu}(x,r)}{\mu(A)} \ge 2^{-m} \left(\frac{r}{diam(A)}\right)^m \tag{3.2.2}$$

where $m = \log_2 C_D$ and $diam(A) = \sup_{x,y \in A} d(x,y)^3$.

Proof. (See [102] and [31]). Let $B(x,r) \subset A$, suppose given 0 < r < diam(A), and put $m = \log_2\left(\frac{diam(A)}{r}\right)$. Then

$$\mu(A) \leq V_{\mu}(x,r) = V_{\mu}\left(x,\frac{2^{m}r}{2^{m}}\right) \leq C_{D}^{m}V_{\mu}\left(x,\frac{r}{2^{m}}\right)$$
$$\leq C_{D}^{m+1}V_{\mu}(x,r) = C_{D}C_{D}^{\log_{2}\left(\frac{diam(A)}{r}\right)}V_{\mu}(x,r)$$
$$\leq C_{D}\left(\frac{diam(A)}{r}\right)^{\log_{2}C_{D}}V_{\mu}(x,r),$$

³Inequality (3.2.2) is actually equivalent to the doubling property of the measure taking B(x, 2r) as the set A.

Finally, we have

$$\mu(A)C_D^{-1} \le \left(\frac{diam(A)}{r}\right)^{\log_2 C_D} V_\mu(x,r).$$

Therefore

$$\frac{V_{\mu}(x,r)}{\mu(A)} \ge 2^{-m} \left(\frac{r}{diam(A)}\right)^{m}.$$

In order to state the opposite inequality in Lemma 3.2.1 we shall need the following definition.

Definition 3.2.2. A metric space (Ω, d) is called uniformly perfect (with constant *a*) if it is not a singleton and if there exists a constant a > 1 such that

$$\Omega \smallsetminus B(x;r) \neq \emptyset \Rightarrow B(x;r) \smallsetminus B(x;r/a) \neq \emptyset$$

for all $x \in \Omega$ and r > 0.



Figure 3.1: Definition 3.2.2

Lemma 3.2.2. Let (Ω, d, μ) be doubling and uniformly perfect. Then there exist constants $D \ge 1$ and k > 0, depending only on the doubling constant C_D and the uniform perfectness constant a, such that

$$\frac{V_{\mu}(x,r)}{V_{\mu}(x,R)} \le D\left(\frac{r}{R}\right)^k \tag{3.2.3}$$

for all $x \in \Omega$ and $0 < r \le R < diam(\Omega)$.

Proof. (See [38]).

By definition 3.2.2 we have that for any constant a > 1, $B(x; ar) \setminus B(x; r) \neq \emptyset$. We shall show that there exist constants C > 1 and D > 1 such that for all $x \in X$ and 0 < ar < diam(X)/2

$$V_{\mu}(x, Cr) \ge DV_{\mu}(x, r).$$
 (3.2.4)

To this end, fix any $0 < \sigma \le 1$. Then, if 0 < r < diam(X)/2, we have $a(1 + \sigma)r < diam(X)$. Thus, by assumption,

$$B(x, a(1+\sigma)r) \setminus B(x, (1+\sigma)r) \neq \emptyset.$$
(3.2.5)

Let y be a point in the annulus. It is then easy to see that

$$B(y,\sigma r) \backslash B(x,r) = \emptyset, \ B(y,\sigma r) \subset B(x,(\sigma + a_0(1+\sigma))r)$$

and

$$B(x, (\sigma + a(1 + \sigma))r) \subset B(y, (\sigma + 2a(1 + \sigma))r), \qquad (3.2.6)$$

and so

$$V_{\mu}(x, (\sigma + a(1 + \sigma))r) \ge V_{\mu}(x, r) + V_{\mu}(y, \sigma r) \quad (by (3.2.5))$$

$$\ge V_{\mu}(x, r) + C_{2}^{-1} \left(\frac{\sigma}{\sigma + 2a(1 + \sigma)}\right)^{m} V_{\mu}(y, (\sigma + 2a(1 + \sigma))r)$$

$$\ge V_{\mu}(x, r) + C_{2}^{-1} \left(\frac{\sigma}{\sigma + 2a(1 + \sigma)}\right)^{m} V_{\mu}(x, (\sigma + a(1 + \sigma))r) \quad (by (3.2.6)).$$

This implies (3.2.3) with $C = \sigma + a(1 + \sigma)$ and

$$D = \left(1 - C_2^{-1} \left(\frac{\sigma}{\sigma + 2a(1+\sigma)}\right)^m\right)^{-1} > 1.$$

Let $0 < \rho \leq r$ and $1 \leq \lambda \rho \leq diam(X)/2$. Let $n := [\log_C \lambda] \geq \log_C \lambda - 1$. Then $C^n \rho \leq diam(X)/2$ and for any $0 \leq k \leq n - 1$,

$$V(x, C^{k+1}\rho) \ge DV(x, C^k\rho).$$

Iterating this inequality, we obtain

$$V(x,\lambda\rho) \ge D^n V(x,\rho) \ge D^{\log_C \lambda - 1} V(x,\rho)$$
$$= D^{-1} \lambda^{\log_C D} V(x,\rho).$$

Note that if a measure satisfies the above inequality with some constants $D \ge 1$ and k > 0, then by choosing $r < D^{1/k}R$ in (3.2.3) we have that the space is uniformly perfect with any constant bigger than $D^{1/k}$.

Example 3.2.2.

- Q-regular spaces are uniformly perfect (see [51]).
- Connected spaces are uniformly perfect (see [46]). Geodesic metric spaces and spaces that support a (1,p)-Poincare inequality are connected (see [52],[45]), and therefore are uniformly perfect.

Combining Lemmas (3.2.1) and (3.2.2), the following is true in doubling uniformly perfect measure metric spaces: There exist positive constants c_0 , C_0 , k, and m ($k \leq m$) depending only on the doubling constant of measure and the uniform perfectness constant of the space (Ω, d, μ) such that

$$c_0 \min(r^k, r^m) V_{\mu}(x, 1) \le V_{\mu}(x, r) \le C_0 \max(r^k, r^m) V_{\mu}(x, 1), \tag{3.2.7}$$

for all $x \in \Omega$ and $0 < r < \infty$.

Note that if $diam(\Omega) < \infty$, from (3.2.2) and (3.2.3) it follows that there exists constants c_1, C_1 such that

$$c_1 r^m \le V_\mu(x, r) \le C_1 r^k,$$
 (3.2.8)

for all $x \in \Omega$ and $0 < r < diam(\Omega)$.

Definition 3.2.3. Let $0 < k \le m$. Let (Ω, d, μ) be a metric measure space.

- (i.) (Ω, d, μ) will be called a (k, m)-space if (3.2.7) holds⁴.
- (ii.) A (k,m)-space will be called **uniform** if there are constants c, C > 0, such that

$$c\min(r^k, r^m) \le V_{\mu}(x, r) \le C\max(r^k, r^m).$$
 (3.2.9)

(iii.) A (k,m)-space will be called **bounded from below** if there are constants d, D > 0 such that

$$d\min(r^k, r^m) \le V_{\mu}(x, r) \le D\max(r^k, r^m)V_{\mu}(x, 1).$$
(3.2.10)

From (3.2.8), we have that doubling uniformly perfect measure metric spaces with $diam(\Omega) < \infty$ are uniform (k, m)-spaces.

Remark 3.2.2. It follows from (3.2.7) that a (k,m)-space is uniform (resp. bounded from below) if, and only if, $0 < \inf_{x \in \Omega} V_{\mu}(x,1) \leq \sup_{x \in \Omega} V_{\mu}(x,1) < \infty$, (resp. $0 < \inf_{x \in \Omega} V_{\mu}(x,1)$)

$$0 < \lim_{x \in \Omega} V_{\mu}(x, 1)).$$

⁴In fact (see ([105])) (Ω, d, μ) is a (k, m)-space if, and only if, it is doubling and uniformly perfect.

In the rest of this chapter, we shall use the following notation:

Notation 3. Let (Ω, d, μ) be a (k, m)-space.

(*i.*) For t > 0,

$$R(t) = \max\left(t^{m/k}, t^{k/m}\right), \ r(t) = \max\left(t^{1/k}, t^{1/m}\right).$$

(ii.) If (Ω, d, μ) is uniform, we write

 $\kappa_0 = 2C_D/c$

where c is the same constant as in (3.2.9).

(iii.) If (Ω, d, μ) is bounded from below, we write

 $\kappa_1 = 2C_D/d$

where d is the same constant as in (3.2.10).

Given (Ω, d, μ) a (k, m) space, we associate to the measure μ a new measure $\tilde{\mu}$ defined by

$$\tilde{\mu}(E) = \int_E \frac{d\mu(x)}{V(x,1)}$$

for any Borel set $E \subset \Omega$.

In the following lemma we obtain some properties of the measure $\tilde{\mu}$.

Lemma 3.2.3. Let (Ω, d, μ) be a (k, m)-space. Let $f \in \mathcal{M}_0(\Omega)$. Then:

(i.) For all r > 0,

$$\min(r^{k}, r^{m}) \oint_{B(x,r)} |f(y)| d\mu(y) \leq \int_{B(x,r)} |f(y)| d\tilde{\mu}(y)$$
(3.2.11)
$$\leq \max(r^{k}, r^{m}) \oint_{B(x,r)} |f(y)| d\mu(y).$$

Thus f is μ -locally integrable if, and only if, f is $\tilde{\mu}$ -locally integrable. Moreover, $(\Omega, d, \tilde{\mu})$ is a uniform (k, m)-space.

(ii.) If (Ω, d, μ) is uniform, then, for all measurable $E \subset \Omega$, we have that

$$\tilde{\mu}(E) \simeq \mu(E).$$

(*iii.*) If (Ω, d, μ) is bounded from below, then for all $f \in L^1_{\tilde{\mu}}(\Omega) + L^{\infty}_{\tilde{\mu}}(\Omega)$

$$f_{\tilde{\mu}}^{*}(t) \leq f_{\mu}^{*}(dt), \quad (t > 0),$$

where d is the same constant that appears in (3.2.10).

Proof. (i.) Using the doubling property and the fact that $B(x,r) \subset B(y,2r)$ whenever $y \in B(x,r)$, we get

$$\begin{split} \int_{B(x,r)} |f(y)| d\tilde{\mu}(y) &= \int_{B(x,r)} |f(y)| \frac{d\mu(y)}{V_{\mu}(y,1)} = \int_{B(x,r)} |f(y)| \frac{V_{\mu}(y,r)}{V_{\mu}(y,1)} \frac{d\mu(y)}{V_{\mu}(y,r)} \\ &\leq C_0 \max(r^k, r^m) \int_{B(x,r)} |f(y)| \frac{d\mu(y)}{V_{\mu}(y,r)} \text{ (by (3.2.7))} \\ &\leq C_D C_0 \max(r^k, r^m) \int_{B(x,r)} |f(y)| \frac{d\mu(y)}{V_{\mu}(y,2r)} \text{ (by (3.2.1))} \\ &\leq C_D C_0 \max(r^k, r^m) \frac{1}{V_{\mu}(x,r)} \int_{B(x,r)} |f(y)| d\mu(y). \end{split}$$

Similarly, if $y \in B(x, r)$, then $B(y, r) \subset B(x, 2r)$, thus

$$\begin{split} \int_{B(x,r)} |f(y)| d\tilde{\mu}(y) &= \int_{B(x,r)} |f(y)| \frac{d\mu(y)}{V_{\mu}(y,1)} = \int_{B(x,r)} |f(y)| \frac{V_{\mu}(y,r)}{V_{\mu}(y,1)} \frac{d\mu(y)}{V_{\mu}(y,r)} \\ &\geq c_0 \min(r^k, r^m) \int_{B(x,r)} |f(y)| \frac{d\mu(y)}{V_{\mu}(y,r)} \\ &\geq c_0 \min(r^k, r^m) \int_{B(x,r)} |f(y)| \frac{d\mu(y)}{V_{\mu}(x,2r)} \\ &\geq \frac{c_0}{C_D} \min(r^k, r^m) \frac{1}{V_{\mu}(x,r)} \int_{B(x,r)} |f(y)| d\mu(y). \end{split}$$

Taking f = 1 in (3.2.11), we obtain that $(\Omega, d, \tilde{\mu})$ is a uniform (k, m)-space. (ii.) This is obvious.

(iii.) From (3.2.10), we get

$$\tilde{\mu}_f(y) = \int_{\{x \in \Omega: |f(x)| > y\}} \frac{d\mu(y)}{V(y, 1)} \le \frac{1}{d} \int_{\{x \in \Omega: |f(x)| > y\}} d\mu(y) = \frac{\mu_f(y)}{d}.$$

Therefore,

$$\mu_f(y) \le dt \Rightarrow \tilde{\mu}_f(y) \le t;$$

 thus

$$f_{\tilde{\mu}}^*(t) = \inf\left\{y : \tilde{\mu}_f(y) \le t\right\} \le \inf\left\{y : \mu_f(y) \le dt\right\} = f_{\mu}^*(dt).$$

We end this section by giving some examples of spaces that satisfy Definition 3.2.3.

3.2.1 Examples

Closed subsets of \mathbb{R}^n (see [50])

We denote by m_n the *n*-dimensional Lebesgue measure on \mathbb{R}^n and by d_n the *n*-Euclidean distance.

- (i.) Consider $F \in \mathbb{R}^2 = \{(x_1, x_2)\}$ defined by $F = F_1 \cup F_2$, where $F_1 = \{(x_1 + 1)^2 + x_2^2 \leq 1\}$ and $F_2 = \{0 \leq x_1 \leq 2, x_2 = 0\}$. Let m_n denote the *n*-dimensional Lebesgue measure, for n = 1 distributed over the x_1 -axes, and put $d\lambda = x_1 dm_1$. Put $\mu = m_{2|F_1} + \lambda_{|F_2}$. Then (F, d_2, μ) is a (1, 2)-uniform space.
- (ii.) Let $F \subset \mathbb{R}^2$ be the set $F = \{0 \le x_1 \le 1, 0 \le x_2 \le x_1^{\gamma}\}$ where $\gamma > 1$, and $d\nu = x_1^{1-\gamma} dm_2$ and $\mu = \nu_{|F}$. Then (F, d_2, μ) is a (1, 2)-uniform space.
- (iii.) (See [50, Proposition 1]) For every closed subset $F \subset \mathbb{R}^n$ there is a measure μ with support F satisfying

$$\mu(B(x,r)) \le c_{\mu}r^{n}\mu(B(x,1)) \text{ and } c_{1} < \mu(B(x,1)) < c_{2}, x \in F.$$

Thus F is uniformly perfect, and there is a k > 0, depending only on c_{μ} and on the uniform perfectness constant of F, such that (F, d_n, μ) is a (k, n)-uniform space.

Muckenhoupt weights

A weight is a positive, locally integrable function on \mathbb{R}^n . For a given subset E of \mathbb{R}^n , let $w(E) \coloneqq \int_E w(x) dx$ and $|E| \coloneqq \int_E dx$. A weight w on \mathbb{R}^n is said to belong to the Muckenhoupt class A_p , $1 \le p < \infty$, (see [75]) if

$$[w]_{A_{p}} \coloneqq \begin{cases} \sup_{B} \left(\frac{1}{|B|} \int_{B} w(x) dx \right) \left(\frac{1}{|B|} \int_{B} \left(\frac{1}{w(x)} \right)^{\frac{1}{p-1}} dx \right)^{p-1} < \infty, & \text{if } 1 < p < \infty, \\ \sup_{B} \frac{\frac{1}{|B|} \int_{B} w(x) dx}{ess \inf_{x \in B} w(x)} < \infty, & \text{if } p = 1, \end{cases}$$

$$(3.2.12)$$

where the supremum is over all balls $B \subset \mathbb{R}^n$. For $p = \infty$, we define $A_{\infty} = \bigcup_{1 \leq p < \infty} A_p$. Given $w \in A_{\infty}$, we define

$$[w]_{A_{\infty}} \coloneqq \sup_{B} \frac{1}{w(B)} \int_{B} M(w\chi_{B})(x) dx$$

where M denotes the usual uncentred Hardy–Littlewood maximal operator. It is known that there is a positive dimensional constant c_n such that $[w]_{A_{\infty}} \leq c_n [w]_{A_n}$.

Given $w \in A_p$, it follows easily from (3.2.12) that if there exists an M > 0 such that $ess \inf_{|x| \ge M} w(x) = 0$, then $\inf_{x \in \mathbb{R}^n} V_{\mu}(x, 1) = 0$. Similarly, if $ess \sup_{|x| \ge M} w(x) = \infty$, then $\sup_{x \in \mathbb{R}^n} V_{\mu}(x, 1) = \infty$.

Proposition 3.2.1. Given $w \in A_p$ and $p \ge 1$, then (\mathbb{R}^n, d_n, w) is a $\left(\frac{n}{2^{n+1}[w]_{A_{\infty}}}, pn\right)$ -space.

Proof. Since $w \in A_p$, by [49, Theorem 2.3], we have that

$$\frac{1}{|B|} \int_B w^r(x) dx \le 2 \left(\frac{1}{|B|} \int_B w(x) dx \right)^r$$

where $r = 1 + \frac{1}{2^{n+1}[w]_{A_{\infty}} - 1}$. Therefore (see [26]), there exist constants c, C > 0 such that

$$c\left(\frac{|E|}{|B|}\right)^{p} \le \frac{w(E)}{w(B)} \le C\left(\frac{|E|}{|B|}\right)^{(r-1)/r}$$
(3.2.13)

for any measurable set E of the ball B. Considering in (3.2.13) $E = B(x,r) \subset B(x,1) = B$ if r < 1, or $E = B(x,1) \subset B(x,r) = B$ if r > 1, and elementary computation shows

$$\min\left(r^{\frac{n}{2^{n+1}[w]_{A_{\infty}}}}, r^{pn}\right)w(B(x,1)) \le w(B(x,r)) \le \max\left(r^{\frac{n}{2^{n+1}[w]_{A_{\infty}}}}, r^{pn}\right)w(B(x,1))$$

Example 3.2.3. Suppose $1 \le p < \infty$, $-n < \alpha \le \beta < n(p-1)$, and

$$w_{\alpha,\beta}(x) = \begin{cases} |x|^{\alpha} & \text{if } |x| \leq 1, \\ |x|^{\beta} & \text{if } |x| > 1. \end{cases}$$

Then $w_{\alpha,\beta}(x) \in A_p$, and

- (*i.*) If $-n < \beta < 0$, then $\inf_{x \in \mathbb{R}^n} V_{\mu}(x, 1) = 0$.
- (*ii.*) If $\beta = 0$, then $\sup_{x \in \mathbb{R}^n} V_{\mu}(x, 1) < \infty$.
- (*iii.*) If $0 < \beta < n(p-1)$, then $\sup_{x \in \mathbb{R}^n} V_{\mu}(x, 1) = \infty$.

Riemannian manifolds with nonnegative Ricci curvature (see [106])

For an n-manifold with nonnegative Ricci curvature, it is known that

$$c(n)Vol(B(x,1))r \le Vol(B(x,r)) \le \omega_n r^n$$

where c(n) is a constant and ω_n is the volume of the unit ball in \mathbb{R}^n . Thus, under the assumption that the manifolds do not have the collapsing volume property⁵, i.e. $\inf_x Vol(B(x,1)) > 0$, they are (1, n)-uniform spaces.

⁵In this context, this condition is related with the property of having finite topological type (see [88]).

Carnot–Carathéodory spaces (see [38])

Let Ω be a connected smooth manifold and suppose given k smooth real vector fields $\{X_1, ..., X_k\}$ on Ω satisfying Hörmander's condition of order m, that is, these vector fields together with their commutators of order at most m span the tangent space to Ω at each point. The control distances associated to the vector fields are defined as follows: for $x, y \in \Omega$ and $\delta > 0$, let $AC(x, y, \delta)$ denote the collection of absolutely continuous mappings $\varphi : [0,1] \to \Omega$ with $\varphi(0) = x$ and $\varphi(1) = y$ such that for almost every $t \in [0,1], \varphi'(t) = \sum_{j=1}^{k} a_j X_j(\varphi(t))$, with $|a_j| \leq \delta$. Then the control metric d(x, y) from x to y is the infimum of the set of $\delta > 0$ such that $AC(x, y, \delta) \neq \emptyset$. Hörmander's condition ensures that $d(x, y) < \infty$ for every $x, y \in \Omega$.

In this context, we have the following examples:

- (i.) (Compact case). If X is a compact n-dimensional Carnot–Carathéodory space with the distance associated to the vector fields and endowed with any fixed smooth measure μ with strictly positive density, then (X, d, μ) is a uniform (n, nm)-space.
- (ii.) (Noncompact case). Let $\Omega = \{(z, w) \in \mathbb{C}^2 : Im[w] > P(x)\}$, where P is a real, subharmonic, nonharmonic polynomial of degree m. Namely, Ω is an unbounded model domain of polynomial type in \mathbb{C}^2 . Then $X = \partial \Omega$ can be identified with $\mathbb{C} \times \mathbb{R} = \{(z, t) : z \in \mathbb{C}, t \in \mathbb{R}\}$ The basic (0,1) Levi vector field is then $\overline{Z} = \partial/\partial \overline{z} i(\partial P/\partial \overline{z})(\partial/\partial t)$, and we write $\overline{Z} = X_1 + iX_2$. The real vector fields $\{X_1, X_2\}$ and their commutators of orders $\leq m$ span the tangent space at each point. If we endow $\mathbb{C} \times \mathbb{R}$ with Lebesgue measure, then $X = \mathbb{C} \times \mathbb{R}$ is a (4, m + 2)-space.
- (iii.) (Noncompact case). Let G be a connected Lie group and fix a left invariant Haar measure μ on G. We assume that G has polynomial volume growth, that is, if U is a compact neighbourhood of the identity element e of G, then there is a constant C > 0 such that $\mu(U^n) \leq n^C$ for all $n \in \mathbb{N}$. Then there is a nonnegative integer n_{∞} such that $\mu(U^n) \simeq n^{n_{\infty}}$ as $n \to \infty$. Let $X_1, ..., X_n$ be left invariant vector fields on G that satisfy Hörmander's condition, that is, they together with their successive Lie brackets $[X_{i_1}, [X_{i_2}, [\ldots, X_{i_k}] \ldots]$ span the tangent space of G at every point of G. Let d be the associated control metric. Then this metric is left invariant and compatible with the topology on G and there is an $n_0 \in \mathbb{N}$, independent of x, such that $\mu(B(x,r)) \simeq r^{n_0}$ when $0 < r \leq 1$, and $\mu(B(x,r)) \simeq r^{n_0}$ when $r \geq 1$. From this, it follows that (G, d, μ) is a uniform $(\min\{n_0, n_{\infty}\}, \max\{n_0, n_{\infty}\})$ space.

3.3 Symmetrization inequalities for moduli

of continuity

Our first result in this section is to obtain embedding results for Besov spaces built on doubling measure spaces. This will be possible only if our space does not have the collapsing volume property.

Theorem 4. Let (Ω, d, μ) be a doubling metric space. Let X be a rearrangement invariant space with $1/p > \overline{\beta}_X$. Assume that the following embedding holds

$$\mathcal{B}^s_{p,q}(\Omega) \subset X.$$

Then Ω does not have the collapsing volume property, i.e.

$$\inf_{x \in \Omega} V_{\mu}(x, 1) > 0. \tag{3.3.1}$$

In particular, $\dot{\mathcal{B}}_{p,q}^{s}(\Omega) \subset L^{p^{*},q}_{\mu}(\Omega)$ for some $p^{*} > p$ implies (3.3.1).

Proof. We claim that the conditions on the indices imply that for $1/p > \varepsilon > \overline{\beta}_X$ and t sufficiently small

$$\frac{t^{1/p}}{\varphi_X(t)} \le t^{\frac{1}{p} - \overline{\beta}_X - \varepsilon}.$$
(3.3.2)

Indeed, suppose s, t > 0. Then

$$\frac{t^{1/p}}{\varphi_X(t)} = \frac{t^{1/p}}{\varphi_X(st)} \frac{\varphi_X(st)}{\varphi_X(t)} \le \frac{t^{1/p}}{\varphi_X(st)} M_X(s)$$

and hence for s = 1/t we get

$$\frac{t^{1/p}}{\varphi_X(t)} \le \frac{t^{1/p}}{\varphi_X(1)} M_X(\frac{1}{t}).$$

Let $1/p > \varepsilon > \overline{\beta}_X$. Then (see Lemma 2.2.1) for t sufficiently small,

$$\frac{t^{1/p}}{\varphi_X(t)} \leq \frac{t^{1/p}}{\varphi_X(1)} M_X(\frac{1}{t})$$
$$\leq \frac{t^{1/p}}{\varphi_X(1)} \left(\frac{1}{t}\right)^{\overline{\beta}_X + \varepsilon}$$
$$= \frac{t^{\frac{1}{p} - \overline{\beta}_X - \varepsilon}}{\varphi_X(1)},$$

as we wanted to see.

For a fixed $x_0 \in \Omega$, we define the Lipschitz function

$$u_{x_0}(y) \coloneqq \begin{cases} (2 - d(x_0, y)) & \text{if } y \in B(x_0, 2) \smallsetminus B(x_0, 1) \\ 1 & \text{if } y \in B(x_0, 1) \\ 0 & \text{if } y \in \Omega \smallsetminus B(x_0, 2). \end{cases}$$

It is easily seen that

$$g_{x_0}(y) = \chi_{B(x_0,2)}(y)$$

is a generalized gradient, i.e.

$$|u_{x_0}(x) - u_{x_0}(y)| \le d(x, y) |g_{x_0}(x) + g_{x_0}(y)|.$$
(3.3.3)

By Fubini's theorem,

$$E_{p}(u_{x_{0}},t)^{p} \leq 2^{p} \int_{\Omega} |u_{x_{0}}(x)|^{p} d\mu(x) + 2^{p} \int_{\Omega} \int_{B(x,t)} |u_{x_{0}}(y)|^{p} d\mu(y) d\mu(x) \quad (3.3.4)$$

$$\leq 2^{p} ||u_{x_{0}}||_{p}^{p} + 2^{p} \int_{\Omega} |u_{x_{0}}(y)|^{p} \left(\int_{B(y,t)} \frac{1}{\mu(B(x,t))} d\mu(x) \right) d\mu(y)$$

$$\leq ||u_{x_{0}}||_{p}^{p},$$

the last estimate following from the doubling property of μ and since $B(y,t) \subset B(x,2t)$ whenever $x \in B(y,t)$.

By (3.3.3) and using a similar argument as in (3.3.4), we get

$$E_{p}(u_{x_{0}},t)^{p} = \int_{\Omega} \left(\int_{B(x,t)} |u_{x_{0}}(x) - u_{x_{0}}(y)|^{p} d\mu(y) \right) d\mu(x)$$

$$\leq \int_{\Omega} \left(\int_{B(x,t)} d(x,y)^{p} |g_{x_{0}}(x) + g_{x_{0}}(y)|^{p} d\mu(y) \right) d\mu(x)$$

$$\leq t^{p} \left(\int_{\Omega} |g_{x_{0}}(x)|^{p} d\mu(x) + \int_{\Omega} \int_{B(x,t)} |g_{x_{0}}(y)|^{p} d\mu(y) d\mu(x) \right)$$

$$\leq t^{p} \|g_{x_{0}}\|_{p}^{p}.$$
(3.3.5)

Thus, combining (3.3.4) and (3.3.5) with the doubling property, we get

$$E_p(u_{x_0}, t) \leq \min(\|u_{x_0}\|_p, t \|g_{x_0}\|_p)$$

$$\leq \min(V_\mu(x_0, 2)^{1/p}, tV_\mu(x_0, 2)^{1/p})$$

$$\leq \min(1, t)V_\mu(x_0, 1)^{1/p}.$$

Therefore

$$||u_{x_0}||_{\dot{\mathcal{B}}^s_{p,q}(\Omega)} \le V_{\mu}(x_0,1)^{1/p}.$$

Since

$$||u_{x_0}||_X \ge \varphi_X(V_\mu(x_0, 1))$$

by hypothesis, we have that

$$1 \le \frac{V_{\mu}(x_0, 1)^{1/p}}{\varphi_X(V_{\mu}(x_0, 1))}.$$
(3.3.6)

If $\inf_{x\in\Omega} V_{\mu}(x,1) = 0$, we can select a sequence $V_{\mu}(x_n,1) \to 0$. Thus, for n big enough, (3.3.6) and (3.3.2) imply

$$1 \le V_{\mu}(x_n, 1)^{\frac{1}{p} - \beta_X - \varepsilon},$$

which is impossible since $\frac{1}{p}-\overline{\beta}_X-\varepsilon>0.$

Recall that our aim is to obtain embedding results for Besov spaces built on doubling measure spaces. Therefore, in view of Theorem 2, it is reasonable to assume that Ω is uniformly perfect (since *Q*-regular spaces are uniformly perfect). Moreover, if we make an additional hypothesis (e.g. Ω supports a (1, p)-Poincaré inequality or Ω is geodesic), then Ω is connected and therefore uniformly perfect. Taking into account these considerations and the previous theorem, our framework in what follows will be a (k, m)-space uniformly or bounded from below.

In order to simplify the notation, throughout what follows we will assume that $\mu(\Omega) = \infty$, since all the results that we obtain can be immediately adapted to the case of finite measure.

3.3.1 Pointwise estimates for the rearrangement

For $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$ and X an r.i. space on Ω , we define:

(i) The gradient at scale r

$$(\nabla_r^{\mu} f)(x) = \int_{B(x,r)} |f(x) - f(y)| d\mu(y), \quad (r > 0).$$

(ii.) The X-modulus of continuity $E_X: (0, \infty) \times X \to [0, \infty)$,

$$E_X(f,r) \coloneqq \|(\nabla^{\mu}_r f)\|_X.$$

Remark 3.3.1. If $X = L^p_{\mu}$, by Hölder's inequality,

$$E_{L^p_{\mu}}(f,r) = \left(\int_{\Omega} \left(\int_{B(x,r)} |f(x) - f(y)| d\mu(y) \right)^p d\mu(x) \right)^{1/p}$$

$$\leq \left(\int_{\Omega} \left(\int_{B(x,t)} |f(x) - f(y)|^p d\mu(y) \right) d\mu(x) \right)^{1/p}$$

$$= E_p(f,r)$$

where $E_p(f,r)$ is the L^p_{μ} -modulus of smoothness defined in (3.1.1).

The aim of this section is to obtain pointwise estimates for the oscillation $O_{\mu}(f,t)$ in terms of the functional $E_X(f,t)$ (see [55],[61] for some related results). The next lemma will be useful in what follows.

Lemma 3.3.1. Let $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$. Let $x \in \Omega$ and t > 0 be such that there is a ball $B_t(x)$ centred at x with $\mu(B_t(x)) = t$. Then

$$f_{\mu}^{**}(t/2) - f_{\mu}^{**}(t) \le \left(\delta_t^{\mu} f\right)_{\mu}^{**}(t/2),$$

where

$$(\delta_t^{\mu} f)(x) = \frac{1}{t} \int_{B_t(x)} |f(x) - f(y)| d\mu(y).$$

Proof. Since

 $|f(x)|\chi_{B_t(x)}(y) \le |f(x) - f(y)|\chi_{B_t(x)}(y) + |f(y)|\chi_{B_t(x)}(y),$

integrating with respect to $d\mu(y)$ yields that

$$\begin{split} |f(x)|t &\leq \int_{B_t(x)} |f(x) - f(y)| d\mu(y) + \int_{B_t(x)} |f(y)| d\mu(y) \\ &\leq \int_{B_t(x)} |f(x) - f(y)| d\mu(y) + \int_0^t f_\mu^*(s) ds \quad (by \ (2.1.2)). \end{split}$$

Now integrating with respect to $d\mu(x)$ over a subset $E \subset \Omega$ with $\mu(E) = t/2$, we get

$$\begin{split} \int_{E} |f(x)| d\mu(x) &\leq \int_{E} \frac{1}{t} \int_{B_{t}(x)} |f(x) - f(y)| d\mu(y) d\mu(x) + \int_{E} \frac{1}{t} \left(\int_{0}^{t} f_{\mu}^{*}(s) ds \right) d\mu(x) \\ &= \int_{E} \left(\delta_{t}^{\mu} f \right)(x) d\mu(x) + \frac{1}{2} \int_{0}^{t} f_{\mu}^{*}(s) ds \end{split}$$

By (2.1.3), taking the supremum over all such sets E, we obtain

$$\int_0^{t/2} f_{\mu}^*(s) ds \le \int_0^{t/2} \left(\delta_t^{\mu} f\right)_{\mu}^*(s) ds + \frac{1}{2} \int_0^t f_{\mu}^*(s) ds,$$

or equivalently

$$f_{\mu}^{**}(t/2) - f_{\mu}^{**}(t) \le \left(\delta_t^{\mu} f\right)_{\mu}^{**}(t/2).$$

Theorem 5. Let $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$. Let X be an r.i. space on Ω .

(i.) If (Ω, d, μ) is uniform, then for all t > 0,

$$O_{\mu}(f,t) \leq \frac{1}{\kappa_0 t} \frac{R(\kappa_0 t)}{\phi_X(\kappa_0 t)} E_X(f,r(\kappa_0 t)).$$
(3.3.7)

(ii.) If (Ω, d, μ) is bounded form below, then for all t > 0,

$$O_{\tilde{\mu}}(f,t) \leq \frac{1}{\kappa_1 1 t} \frac{R(\kappa_1 t)}{\phi_X(\kappa_1 t)} E_X(f,r(\kappa_1 t)).$$

Proof. (i.) Given $x \in \Omega$ and t > 0, by Remark 9 there is a ball B(x) centred at x such that $t/C_D \leq \mu(B(x)) \leq t$. We denote by z the measure of this ball, i.e. $\mu(B_z(x)) = z$, with $t/C_D \leq z \leq t$. From (3.2.9) it follows that

$$\begin{cases} \mu(B_z(x)) \le t \le V_\mu(x, (t/c)^{1/m}) \le C (t/c)^{k/m} & \text{if } t < c, \\ \mu(B_z(x)) \le t \le V_\mu(x, (t/c)^{1/k}) \le C (t/c)^{m/k} & \text{if } t \ge c, \end{cases}$$

i.e.

$$\mu(B_z(x)) \le t \le V_\mu(x, r(t/c)) \le CR(t/c).$$
(3.3.8)

Obviously, $B_z(x) \subset B(x, r(t/c))$, and thus

$$\begin{aligned} \left(\delta_{z}^{\mu}f\right)(x) &= \frac{1}{z} \int_{B_{z}(x)} |f(x) - f(y)| \, d\mu(y) \\ &\leq \frac{C_{D}}{t} \int_{B(x, r(t/c))} |f(x) - f(y)| \, d\mu(y) \\ &\leq CC_{D} \frac{R(t/c)}{t} \int_{B(x, r(t/c))} |f(x) - f(y)| \, d\mu(y) \quad (by \ (3.3.8)) \\ &= CC_{D} \frac{R(t/c)}{t} \left(\bigtriangledown_{r(t/c)}^{\mu}f \right)(x). \end{aligned}$$

Taking rearrangements, we get

$$(\delta_{z}^{\mu}f)_{\mu}^{*}(s) \leq CC_{D}\frac{R(t/c)}{t} \left(\nabla_{r(t/c)}^{\mu}f\right)_{\mu}^{*}(s), \ s > 0,$$

which implies

$$(\delta_z^{\mu} f)_{\mu}^{**}(s) \le CC_D \frac{R(t/c)}{t} \left(\nabla_{r(t/c)}^{\mu} f \right)_{\mu}^{**}(s), \ s > 0.$$

On the other hand,

$$\left(\nabla_{r(t/c)}^{\mu} f \right)_{\mu}^{**}(s) \leq \frac{1}{\phi_X(s)} \sup_{s} \left(\phi_X(s) \left(\nabla_{r(t/c)}^{\mu} f \right)_{\mu}^{**}(s) \right)$$

$$= \frac{1}{\phi_X(s)} \left\| \left(\nabla_{r(t/c)}^{\mu} f \right) \right\|_{M(X)} \quad (by \ (2.2.7))$$

$$\leq \frac{1}{\phi_X(s)} \left\| \left(\nabla_{r(t/c)}^{\mu} f \right) \right\|_X \quad (by \ (2.2.8))$$

$$= \frac{1}{\phi_X(s)} E_X(f, r(t/c)).$$

Combining this inequality and Lemma 3.3.1, we obtain

$$f_{\mu}^{**}(z/2) - f_{\mu}^{**}(z) \le (\delta_{z}^{\mu}f)_{\mu}^{**}(z/2) \le CC_{D} \frac{R(t/c)}{t\phi_{X}(z/2)} E_{X}(f, r(t/c)).$$

By Remark 2.1.1, we get

$$f_{\mu}^{**}(z/2) - f_{\mu}^{**}(z) = \int_{z/2}^{z} \left(f_{\mu}^{**}(s) - f_{\mu}^{*}(s) \right) \frac{ds}{s} \ge \frac{f_{\mu}^{**}(z/2) - f_{\mu}^{*}(z/2)}{2}.$$

In summary,

$$O_{\mu}(f, z/2) \le 2CC_D \frac{R(t/c)}{t\phi_X(z/2)} E_X(f, r(t/c)).$$
(3.3.9)

Finally, using that $t/C_D \leq z \leq t$, we get

$$\begin{aligned} \frac{t}{2C_D}O_{\mu}\left(f,\frac{t}{2C_D}\right) &\leq \frac{z}{2}O_{\mu}\left(f,\frac{z}{2}\right) \text{ (by Remark 2.1.1)} \\ &\leq 2CC_D\frac{z/2}{\phi_X(z/2)}\frac{R\left(t/c\right)}{t}E_X(f,r(t/c)) \text{ (by (3.3.9))} \\ &\leq 4CC_D\frac{R\left(t/c\right)}{\phi_X(t/2)}E_X(f,r(t/c)) \text{ (since } \frac{s}{\phi_X(s)} \text{ increases)} \\ &\leq 4CC_D\left(\sup_{t>0}\frac{\phi_X(2t/c)}{\phi_X(t/2)}\right)\frac{R\left(t/c\right)E_X(f,r(t/c))}{\phi_X(t/c)} \\ &= 4CC_DM_X(2/c)\frac{R\left(t/c\right)E_X(f,r(t/c))}{\phi_X(t/c)},\end{aligned}$$

which implies (3.3.7). (ii.) By Lemma 3.2.3, we get $L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega) \subset L^1_{\tilde{\mu}}(\Omega) + L^{\infty}_{\tilde{\mu}}(\Omega)$ since $\tilde{\mu}$ is doubling. By Remark 9, given $x \in \Omega$ and t > 0, there is a ball $B_z(x)$ centred at xsuch that $t/C_D \leq \tilde{\mu}(B_z(x)) = z \leq t$. Then

$$\left(\delta_{z}^{\tilde{\mu}} f \right)(x) \leq \frac{C_{D}}{t} \int_{B(x,r(t/d))} |f(x) - f(y)| \frac{d\mu(y)}{V(y,1)}$$

$$\leq C_{D} D \frac{R(t/d)}{t} \int_{B(x,r(t/d))} |f(x) - f(y)| d\mu(y) \quad (by \ (3.2.11))$$

$$= C_{D} D \frac{R(t/d)}{t} \left(\nabla_{r(t/d)}^{\mu} f \right)(x).$$

Taking rearrangements with respect to $\tilde{\mu}$, we have that for all s > 0

$$\left(\delta_z^{\tilde{\mu}}f\right)_{\tilde{\mu}}^*(s) \leq C_D D \frac{R(t/d)}{t} \left(\nabla_{r(t/d)}^{\mu}f\right)_{\tilde{\mu}}^*(s)$$
$$\leq C_D D \frac{R(t/d)}{t} \left(\nabla_{r(t/d)}^{\mu}f\right)_{\mu}^*(sd)$$
 (by Lemma 3.2.3).

Hence,

$$\begin{split} \left(\delta_{z}^{\tilde{\mu}}f\right)_{\tilde{\mu}}^{**}(s) &= \frac{1}{s} \int_{0}^{s} \left(\delta_{z}^{\tilde{\mu}}f\right)_{\tilde{\mu}}^{*}(y)dy \\ &\leq C_{D}D\frac{R(t/d)}{t}\frac{1}{s} \int_{0}^{s} \left(\nabla_{r\left(\frac{t}{d}\right)}^{\mu}f\right)_{\mu}^{*}(yd)dy \\ &= C_{D}D\frac{R(t/d)}{t}\frac{1}{sd} \int_{0}^{sd} \left(\nabla_{r\left(\frac{t}{d}\right)}^{\mu}f\right)_{\mu}^{*}(y)dy \\ &= C_{D}D\frac{R(t/d)}{t} \left(\nabla_{r(t/d)}^{\mu}f\right)_{\mu}^{**}(sd). \end{split}$$

and

$$\left(\nabla^{\mu}_{r(t/d)}f\right)^{**}_{\mu}(sd) \leq \frac{1}{\phi_X(sd)}E_X(f,r(t/d)).$$

Thus,

$$f_{\tilde{\mu}}^{**}(z/2) - f_{\tilde{\mu}}^{**}(z) \le DC_D \frac{R(t/d)}{t\phi_X(zd/2)} E_X(f, r(t/d)).$$

Now we finish the proof as in part (i).

3.4 A Sobolev type embedding result for Besov spaces

Definition 3.4.1. Let (Ω, d, μ) be a (k, m)-space. Let X be a r.i. space on Ω and let Y be an r.i. space on $[0, \infty)$ over $[0, \infty)$ with respect to Lebesgue measure. Let 0 < s < 1. The **Besov space** $\mathring{B}^s_{(k,m),X,Y}(\Omega)$ is the set of those functions in $L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$ for which the semi-norm

$$\|f\|_{\mathring{B}^{s}_{(k,m),X,Y}(\Omega)} \coloneqq \left\|\frac{r(t)^{-s}E_{X}(f,r(t))}{\phi_{Y}(t)}\right\|_{Y}$$

 $is\ finite.$

Remark 3.4.1. We write $\mathring{B}^{s}_{(k,m),p,q}(\Omega)$ if $X = L^{p}_{\mu}(\Omega)$ and $Y = L^{q}([0,\infty))$ $(1 \le p < \infty, \ 1 \le q \le \infty)$. In this case $\phi_{Y}(t) = t^{1/q}$, thus if $1 \le q < \infty$,

$$\begin{split} \|f\|_{\mathring{B}^{s}_{(k,m),p,q}(\Omega)} &\leq \left\|\frac{E_{p}(f,r(t))}{(r(t))^{s}t^{1/q}}\right\|_{L^{q}} \quad (by \; Remark \; 3.3.1) \\ &= \left(\int_{0}^{\infty} \left(\frac{E_{p}(f,\max\left(t^{1/k},t^{1/m}\right))}{\max\left(t^{1/k},t^{1/m}\right)^{s}}\right)^{q}\frac{dt}{t}\right)^{1/q} \\ &= \left(\int_{0}^{1} \left(\frac{E_{p}(f,t^{1/m})}{t^{s/m}}\right)^{q}\frac{dt}{t} + \int_{1}^{\infty} \left(\frac{E_{p}(f,t^{1/k})}{t^{s/k}}\right)^{q}\frac{dt}{t}\right)^{1/q} \\ &\simeq \left(\int_{0}^{1} \left(\frac{E_{p}(f,t)}{t^{s}}\right)^{q}\frac{dt}{t} + \int_{1}^{\infty} \left(\frac{E_{p}(f,t)}{t^{s}}\right)^{q}\frac{dt}{t}\right)^{1/q} \\ &= \left(\int_{0}^{\infty} \left(\frac{E_{q}(f,t)}{t^{s}}\right)^{q}\frac{dt}{t}\right)^{1/q}. \end{split}$$

Therefore,

$$\dot{\mathcal{B}}^{s}_{p,q}(\Omega) \subset \dot{B}^{s}_{(k,m),p,q}(\Omega).$$

Similarly,

$$\dot{\mathcal{B}}^{s}_{p,\infty}(\Omega) \subset \mathring{B}^{s}_{(k,m),p,\infty}(\Omega).$$

3.4.1 Some new function spaces

Following [59], we shall now construct the range spaces for our generalized Besov–Sobolev embedding theorem.

Definition 3.4.2. Let (Ω, d, μ) be a (k, m)-space. Given $s \in \mathbb{R}$, we define

$$v_s(t) \coloneqq \frac{t}{R(t)r(t)^s} = \min\left(t^{1-\frac{m+s}{k}}, t^{1-\frac{k+s}{m}}\right)$$

and

$$S_{\mu}^{X,Y}(v_s) = \left\{ f : \|f\|_{S_{\mu}^{X,Y}(v_s)} = \left\| v_s(t) \frac{\phi_X(t)}{\phi_Y(t)} O_{\mu}(f,t) \right\|_Y < \infty \right\},\$$

where ϕ_X is the fundamental function of X, an r.i. space on Ω , and Y is an r.i. space on $[0, \infty)$ with respect to Lebesgue measure.

Note that these spaces are not necessarily linear and, in particular, $\|.\|_{S^{X,Y}_{\bar{\mu}}(v_s)}$ is not necessarily a norm.

Given an r.i. space X, we shall say that Y satisfies the Q(s, (k, m), X)condition if there exists a constant C > 0 such that

$$\left\| v_s(t) \frac{\phi_X(t)}{\phi_Y(t)} Qf(t) \right\|_Y \le C \left\| v_s(t) \frac{\phi_X(t)}{\phi_Y(t)} f(t) \right\|_Y$$

The following lemmas will be useful in what follows. A consequence of our first lemma is that if Y satisfies the Q(s, (k, m), X)-condition, then $S^{X,Y}_{\mu}(v_s)$ is a Banach space.

Lemma 3.4.1. Let X, Y be two r.i. spaces. If Y satisfies the Q(s, (k, m), X)-condition, then for all $f_{\mu}^{*}(\infty) = 0$,

$$\|f\|_{S^{X,Y}_{\mu}(v_s)} \simeq \left\|v_s(t)\frac{\phi_X(t)}{\phi_Y(t)}f^{**}_{\mu}(t)\right\|_Y, \qquad (3.4.1)$$

with constants of equivalence independent of f.

Proof. Obviously,

$$\left\| v_s(t) \frac{\phi_X(t)}{\phi_Y(t)} O_\mu(f, t) \right\|_Y \le \left\| v_s(t) \frac{\phi_X(t)}{\phi_Y(t)} f_\mu^{**}(t) \right\|_Y$$

Conversely, from $\frac{d}{dt}f_{\mu}^{**}(t) = -\frac{f_{\mu}^{**}(t) - f_{\mu}^{*}(t)}{t}$ and the Fundamental Theorem of Calculus, we have

$$f_{\mu}^{**}(t) = \int_{t}^{\infty} \left(f_{\mu}^{**}(s) - f_{\mu}^{*}(s) \right) \frac{ds}{s} = Q \left(f_{\mu}^{**} - f_{\mu}^{*} \right) (t),$$

and the result follows by the Q(s, (k, m), X)-condition.

The next result gives a useful criterion for checking the validity of a Q(s, (k, m), X)-condition.

Lemma 3.4.2. Let X, Y be two r.i. spaces. Suppose that

$$\int_{1}^{\infty} t^{\frac{m+s}{k}-1} h_Y(1/t) M_X(1/t) M_Y(t) \frac{dt}{t} < \infty.$$
(3.4.2)

Then Y satisfies the Q(s, (k, m), X)-condition.

Proof. Let us write $v_s \coloneqq v$. We have

$$\begin{split} v(t)\frac{\phi_X(t)}{\phi_Y(t)}Qf(t) &= \int_t^\infty v(t)\frac{\phi_X(t)}{\phi_Y(t)}f(x)\frac{dx}{x} = \int_1^\infty v(t)\frac{\phi_X(t)}{\phi_Y(t)}f(tx)\frac{dx}{x} \\ &\leq \int_1^\infty v(tx)f(tx)\frac{\phi_X(xt)}{\phi_Y(xt)}\sup_{t>0}\frac{v(t)}{v(tx)}\sup_{t>0}\frac{\phi_X(t)}{\phi_X(xt)}\sup_{t>0}\frac{\phi_Y(xt)}{\phi_Y(t)}\frac{dx}{x} \\ &= \int_1^\infty v(tx)f(tx)\frac{\phi_X(xt)}{\phi_Y(xt)}\sup_{t>0}\frac{v(t)}{v(tx)}M_X(1/x)M_Y(x)\frac{dx}{x}. \end{split}$$

Applying Minkowski's inequality, we obtain

$$\begin{aligned} \left\| v(t) \frac{\phi_X(t)}{\phi_Y(t)} Qf(t) \right\|_Y &\leq \int_1^\infty \left\| v(tx) f(tx) \frac{\phi_X(xt)}{\phi_Y(xt)} \right\|_Y \sup_{t>0} \frac{v(t)}{v(tx)} M_X(1/x) M_Y(x) \frac{dx}{x} \\ &\leq \int_1^\infty \left(\sup_{t>0} \frac{v(t)}{v(tx)} \right) h_Y(1/x) M_X(1/x) M_Y(x) \frac{dx}{x} \left\| v(t) f(t) \frac{\phi_X(t)}{\phi_Y(t)} \right\|_Y \end{aligned}$$

Finally, an elementary computation shows that if x > 1, then

$$\sup_{t>0}\frac{v(t)}{v(tx)}=x^{\frac{m+s}{k}-1}.$$

Remark 3.4.2. In terms of indices, it is easy to see that (3.4.2) is equivalent to the inequality

$$\frac{m+s}{k} - 1 < \underline{\alpha}_Y - \overline{\beta}_Y + \underline{\beta}_X.$$

Moreover, if $\phi_X(t)/\phi_Y(t)$ is equivalent to an increasing function, then starting from $\varphi_{s,(k,m)}(t)\frac{\phi_X(t)}{\phi_Y(t)}Qf(t) \leq c \int_t^{\infty} \varphi_{s,(k,m)}(t)\frac{\phi_X(x)}{\phi_Y(x)}f(x)\frac{dx}{x}$ and following the same steps as in the proof of the previous lemma, we see that

$$\int_{1}^{\infty} t^{\frac{m+s}{k}-1} h_Y\left(\frac{1}{t}\right) \frac{dt}{t} < \infty$$

implies that Y satisfies the Q(s, (k, m), X)-condition.

In the particular case that $Y = L^q$ we can obtain a more specific result.

Lemma 3.4.3. Let X be an r.i. space. Let $s \in (0,1)$ and suppose $q \ge 1$. Let $v(t) = \varphi_s(t)t^{-1/q}\phi_X(t)$. Then the following statements are equivalent. i) $\frac{m+s}{k} - 1 < \underline{\beta}_X$. ii) L^q satisfies the Q(s, (k, m), X)-condition. iii) If $f^*(\infty) = 0$, $\|v(t) \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t)\right)\|_{L^q_{\mu}} \simeq \|v(t)f_{\mu}^{**}(t)\|_{L^q_{\mu}}$.

Proof. (i) \rightarrow (ii) Since $\underline{\alpha}_Y = \overline{\beta}_Y = \frac{1}{q}$, Remark 3.4.2 applies. ii) \rightarrow iii) follows from Lemma 3.4.1. To conclude the proof, we show iii) \rightarrow i): By Fubini we readily see that (we need $f^*_{\mu}(\infty) = 0$, otherwise Qf^*_{μ} does not exist)

$$Qf_{\mu}^{*} = Q \circ Pf_{\mu}^{*} = Qf_{\mu}^{*} + Pf_{\mu}^{*} = P \circ Qf_{\mu}^{*} = (Qf_{\mu}^{*})^{**}$$

Thus

$$\left(Qf_{\mu}^{*}\right)^{**}(t) - Qf_{\mu}^{*}(t) = Pf_{\mu}^{*}(t) = f_{\mu}^{**}(t).$$
(3.4.3)

Consider now the r.i. space H defined by the norm

$$\|h\|_{H} = \|v(t)f_{\mu}^{**}(t)\|_{L_{\mu}^{q}}$$

Then, by condition iii),

$$\phi_H(r) = \left\| \chi_{[0,r]} \right\|_H \simeq \left\| v(t) \left(\chi_{[0,r]}^{**}(t) - \chi_{[0,r]}^{*}(t) \right) \right\|_{L^q_\mu}$$
$$= r \left(\int_r^\infty v(t)^q \frac{dt}{t^q} \right)^{1/q}.$$

On the other hand, since ϕ_X is increasing and $\varphi_s(t)/t$ is decreasing,

$$\int_{r}^{\infty} v(t)^{q} \frac{dt}{t^{q}} = \int_{r}^{\infty} \left(\varphi_{s}(t)t^{-1/q}\phi_{X}(t)\right)^{q} \frac{dt}{t^{q}}$$
$$\geq \int_{r}^{2r} \phi_{X}(t)^{q} \left(\frac{\varphi_{s}(t)}{t}\right)^{q} \frac{dt}{t}$$
$$\geq \phi_{X}(r)^{q} \left(\frac{\varphi_{s}(2r)}{2r}\right)^{q}.$$

Similarly, since $\phi_X(t)/t$ is decreasing,

$$\int_{r}^{\infty} \left(\varphi_{s}(t)t^{-1/q}\phi_{X}(t)\right)^{q} \frac{dt}{t^{q}} \leq \left(\frac{\phi_{X}(r)}{r}\right)^{q} \int_{r}^{\infty} \varphi_{s}(t)^{q} \frac{dt}{t}$$
$$= \left(\frac{\phi_{X}(r)}{r}\right)^{q} \int_{r}^{\infty} \left(\min\left(t^{1-\frac{m+s}{k}}, t^{1-\frac{k+s}{m}}\right)\right)^{q} \frac{dt}{t}$$
$$\leq \left(\frac{\phi_{X}(r)}{r}\right)^{q} \int_{r}^{\infty} \left(t^{1-\frac{m+s}{k}}\right)^{q} \frac{dt}{t}$$
$$\simeq \left(\frac{\phi_{X}(r)}{r}\right)^{q} \left(r^{1-\frac{m+s}{k}}\right)^{q} \quad (\text{since } k \leq m).$$

Thus, if r > 1,

$$\phi_H(r) \simeq \phi_X(r) r^{1 - \frac{m+s}{k}}$$

and

$$\underline{\beta}_{H} = \left(1 - \frac{m+s}{k}\right) + \underline{\beta}_{X}.$$

Finally, since

$$\begin{split} \|f\|_{H} &= \left\| v(t)(\left(Qf_{\mu}^{*}\right)^{**}(t) - Qf_{\mu}^{*}(t)\right) \right\|_{L_{\mu}^{q}} \quad (\text{by (3.4.3)}) \\ &\simeq \left\| v(t)\left(Qf_{\mu}^{*}\right)^{**}(t) \right\|_{L_{\mu}^{q}} \quad (\text{by condition } iii)) \\ &= \left\|Qf_{\mu}^{*}\right\|_{H}, \end{split}$$

it follows that $Q:H \to H$ is bounded, which implies that $\underline{\beta}_H > 0,$ thus

$$\underline{\beta}_X > \frac{m+s}{k} - 1.$$

c			_
			- 1
			- 1
			- 1
-	-	-	-

From Theorem 5 we get immediately the following generalization of the Sobolev embedding theorem for Besov spaces.

Theorem 6. Let (Ω, d, μ) be a (k, m)-space, X, Y r.i. spaces, and 0 < s < 1. Then

(i.) If (Ω, d, μ) is uniform,

$$\mathring{B}^{s}_{(k,m),X,Y}(\Omega) \subset S^{X,Y}_{\mu}(v_s).$$

Moreover if Y satisfies the Q(s, (k, m), X)-condition, then for all $f^*_{\mu}(\infty) = 0$,

$$\left\| v_s(t) \frac{\phi_X(t)}{\phi_Y(t)} f_{\mu}^{**}(t) \right\|_Y \leq \|f\|_{\dot{B}^s_{(k,m),X,Y}(\Omega)}.$$

(ii.) If (Ω, d, μ) is bounded from below,

$$\mathring{B}^{s}_{(k,m),X,Y}(\Omega) \subset S^{X,Y}_{\tilde{\mu}}(v_s).$$

Moreover if Y satisfies the Q(s, (k, m), X)-condition, then for all $f^*_{\tilde{\mu}}(\infty) = 0$,

$$\left\| v_s(t) \frac{\phi_X(t)}{\phi_Y(t)} f_{\tilde{\mu}}^{**}(t) \right\|_Y \leq \|f\|_{\dot{B}^s_{(k,m),X,Y}(\Omega)}.$$

Proof. (i.) Let $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$. Then from (3.3.7) we know that there is a constant $\kappa > 0$ such that

$$O_{\mu}(f,t) \leq \frac{1}{\kappa_0 t} \frac{R(\kappa_0 t)}{\phi_X(\kappa_0 t)} E_X(f,r(\kappa_0 t)), \quad t > 0.$$

Thus,

$$\begin{split} \|f\|_{S^{X,Y}_{\mu}(v_{s})} &\leq \left\|\frac{t}{R(t)r(t)^{s}} \frac{\phi_{X}(t)}{\phi_{Y}(t)} \frac{1}{\kappa_{0}t} \frac{R(\kappa_{0}t)}{\phi_{X}(\kappa_{0}t)} E_{X}(f,r(\kappa_{0}t))\right\|_{Y} \\ &= \left\|\frac{R(\kappa_{0}t)r(\kappa_{0}t)}{R(t)r(t)^{s}} \frac{\phi_{X}(t)\phi_{Y}(\kappa_{0}t)}{\phi_{X}(\kappa_{0}t)\phi_{Y}(t)} \frac{1}{\kappa_{0}} \frac{r(\kappa_{0}t)^{-s}}{\phi_{Y}(\kappa_{0}t)} E_{X}(f,r(\kappa_{0}t))\right\|_{Y} \\ &\leq \sup_{t>0} \left(\frac{R(\kappa_{0}t)r(\kappa_{0}t)}{R(t)r(t)^{s}} \frac{\phi_{X}(t)\phi_{Y}(\kappa_{0}t)}{\phi_{X}(\kappa_{0}t)\phi_{Y}(t)} \frac{1}{\kappa_{0}}\right) \left\|\frac{r(\kappa_{0}t)^{-s}}{\phi_{Y}(\kappa_{0}t)} E_{X}(f,r(\kappa_{0}t))\right\|_{Y} \\ &\leq h_{Y}(1/\kappa_{0}) \left\|\frac{r(t)^{-s}}{\phi_{Y}(t)} E_{X}(f,r(t))\right\|_{Y} \\ &\leq \|f\|_{\mathring{B}^{s}_{(k,m),X,Y}(\Omega)} \,. \end{split}$$

Part (ii) is analogous.

3.5 Uncertainty type inequalities

The purpose of this section is to extend the generalized uncertainty Sobolev inequalities obtained in [66] to the context of Besov spaces.

Definition 3.5.1. Let (Ω, d, μ) be a (k, m)-space, X, Y r.i. spaces, and 0 < s < 1. We will say that a μ -measurable function $w : \Omega \to (0, \infty)$ is an (s, (k, m), X, Y)admissible weight if

$$[w] \coloneqq \sup_{t>0} \left(\left(\left(\frac{1}{w}\right)_{\mu}^{*}(t) \right)^{s} \frac{1}{v_{s}(t) \frac{\phi_{X}(t)}{\phi_{Y}(t)}} \right) < \infty.$$

Theorem 7. Let (Ω, d, μ) be a uniform (k, m)-space, let X, Y be r.i. spaces, suppose 0 < s < 1, and let w be an (s, (k, m), X, Y)-admissible weight. Assume that Y satisfies the Q(s, (k, m), X)-condition. Let $\alpha > 0$. Then for all $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$ such that $f^*_{\mu}(\infty) = 0$, we have that

$$\|f\|_{Y} \leq [w]^{\frac{\alpha}{\alpha+1}} \|f\|_{\dot{B}^{s}_{(k,m),X,Y}(\Omega)}^{\frac{\alpha}{\alpha+1}} \|w^{\alpha s}f\|_{Y}^{\frac{1}{\alpha+1}}.$$
(3.5.1)

Proof. Since $f_{\mu}^{*}(\infty) = 0$, by the Fundamental Theorem of Calculus and (3.3.7), we get

$$f_{\mu}^{**}(t) = \int_{t}^{\infty} \left(f_{\mu}^{**}(s) - f_{\mu}^{*}(s) \right) \frac{ds}{s}$$

$$\leq \int_{t}^{\infty} \frac{1}{\kappa_{0}s} \frac{R(\kappa_{0}s)}{\phi_{X}(\kappa_{0}s)} E_{X}(f, r(\kappa_{0}s)) \frac{ds}{s}.$$

$$(3.5.2)$$

Then

$$\begin{split} \|f\|_{Y} &= \left\| f\left(\frac{w}{w}\right)^{s} \right\|_{Y} \leq \left\| f\left(\frac{w}{w}\right)^{s} \chi_{\left\{w \leq r^{1/s}\right\}} \right\|_{Y} + \left\| f\left(\frac{w}{w}\right)^{s} \chi_{\left\{w > r^{1/s}\right\}} \right\|_{Y} \\ &\leq r \left\| f\left(\frac{1}{w}\right)^{s} \right\|_{Y} + \left\| f\left(\frac{w}{w}\right)^{s\alpha} \chi_{\left\{w > r^{1/s}\right\}} \right\|_{Y} \\ &\leq r \left\| f\left(\frac{1}{w}\right)^{s} \right\|_{Y} + r^{-\alpha} \left\| w^{\alpha s} f \right\|_{Y}. \end{split}$$

Now we estimate the first term:

$$\begin{split} \left\| f\left(\frac{1}{w}\right)^{s} \right\|_{Y} &\leq \left\| f_{\mu}^{*}(t) \left(\left(\frac{1}{w}\right)_{\mu}^{*}(t) \right)^{s} \frac{v_{s}(t) \frac{\phi_{X}(t)}{\phi_{Y}(t)}}{v_{s}(t) \frac{\phi_{X}(t)}{\phi_{Y}(t)}} \right\|_{Y} \\ &\leq \left[w \right] \left\| f_{\mu}^{*}(t) v_{s}(t) \frac{\phi_{X}(t)}{\phi_{Y}(t)} \right\|_{Y} \\ &\leq \left[w \right] \left\| f_{\mu}^{**}(t/2) v_{s}(t) \frac{\phi_{X}(t)}{\phi_{Y}(t)} \right\|_{Y} \\ &\leq \left[w \right] \left\| v_{s}(t)(t) \frac{\phi_{X}(t)}{\phi_{Y}(t)} \int_{t}^{\infty} \frac{1}{\kappa_{0s}} \frac{R(\kappa_{0s})}{\phi_{X}(\kappa_{0s})} E_{X}(f, r(\kappa_{0}s)) \frac{ds}{s} \right\|_{Y} \quad (by (3.5.2)) \\ &\leq \left[w \right] \left\| \frac{r(t)^{-s} E_{X}(f, r(t))}{\phi_{Y}(t)} \right\|_{Y} \quad (by the Q(s, (k, m), X) - condition) \\ &\leq \left[w \right] \| f \|_{\mathring{B}^{s}_{(k,m), X, Y}(\Omega)} . \end{split}$$

In summary, we have proved that there is an absolute constant A > 0 such that

$$\|f\|_{Y} \le A[w]r \|f\|_{\dot{B}^{s}_{(k,m),X,Y}(\Omega)} + r^{-\alpha} \|w^{\alpha s}f\|_{Y}.$$
(3.5.3)

Selecting the value $r = \left(\frac{\|w^{s\alpha}f\|_{Y}}{2A[w]\|f\|_{\dot{B}^{s}_{(k,m),X,Y}(\Omega)}}\right)^{\frac{1}{1+\alpha}}$ to compute (3.5.3) balances the two terms and we obtain the multiplicative inequality (3.5.1).

Remark 3.5.1. The connection with the notion of isoperimetric weight introduced in [66] is the following: consider the case $(\mathbb{R}^n, |\cdot|, \mathcal{L}^n)$. Obviously, it is a uniform (k, m)-space with $k = m = \frac{1}{n}$. Let $X = Y = L^q$. Then

$$[w] \coloneqq \sup_{t>0} \left(\left(\left(\frac{1}{w}\right)_{\mu}^{*}(t) \right)^{s} t^{\frac{s}{n}} \right) = \sup_{t>0} \left(\left(\left(\frac{1}{w}\right)_{\mu}^{*}(t) \right) t^{\frac{1}{n}} \right)^{s}.$$

Thus w is admissible if, and only if, $\frac{1}{w} \in L^{n,\infty}$ (i.e. w is an isoperimetric weight). Let $\alpha > 0$, $1 \le q < \infty$, and 0 < s < 1 with s < n/q. By Remark 3.4.2, L^q satisfies the $Q(s, (\frac{1}{n}, \frac{1}{n}), L^q)$ -condition. Then by Theorem 7, if $\frac{1}{w} \in L^{n,\infty}$,

$$\|f\|_{l^{q}} \leq (2\kappa [w])^{\frac{\alpha}{\alpha+1}} \|f\|_{\dot{B}^{s}_{p,q}(\mathbb{R}^{n})}^{\frac{\alpha}{\alpha+1}} \|w^{\alpha s}f\|_{L^{q}}^{\frac{1}{\alpha+1}},$$

where $\mathring{B}^{s}_{p,q}(\mathbb{R}^{n})$ is the classical Euclidean Besov space.

3.6 Embedding into *BMO* and essential continuity

Definition 3.6.1. Let $f : \Omega \to \mathbb{R}$ be a locally integrable function on Ω . Then f is said to have bounded mean oscillation (written $f \in BMO$) if the seminorm is given by

$$\|f\|_{BMO_{\mu}(\Omega)} = \sup_{B} \left\{ \int_{B} |f - f_{B}| d\mu \right\} < \infty.$$

Here B denotes any ball of Ω .

Theorem 8. Let (Ω, d, μ) be a uniform (k, m)-space. Then

$$\|f\|_{BMO_{\mu}(\Omega)} \leq \sup_{t>0} \frac{R(t)}{t\phi_X(t)} E_X(f, r(t))$$

Proof. Let $B \coloneqq B(x)$ be a ball centred at x. Since (Ω, d, μ) is a uniform (k, m)-space, we have that

$$\mu(B) \le \mu(B(x, r(\mu(B)/c)) \le CR(\mu(B)/c).$$

Then

$$\begin{split} I &\coloneqq \int_{B} \left| f(y) - \int_{B(x)} f(s) d\mu(s) \right| d\mu(y) \\ &\leq \int_{B} \left(\int_{B(x)} |f(y) - f(s)| d\mu(s) \right) d\mu(y) \\ &\leq \int_{B} \int_{B(x, r(\mu(B)/c))} |f(y) - f(s)| d\mu(s) d\mu(y) \\ &\leq C \frac{R(\mu(B)/c)}{\mu(B)} \int_{B} \left(\int_{B(x, r(\mu(B)/c))} |f(s) - f(y)| d\mu(s) \right) d\mu(y) \\ &\leq C \frac{R(\mu(B)/c)}{\mu(B)} E_{X}(f, r(\mu(B)/c)) \left\| \chi_{B(x)} \right\|_{X'} \quad (by \ (2.2.4)) \\ &= C \frac{R(\mu(B)/c)}{\phi_{X}(\mu(B))} E_{X}(f, r(\mu(B)/c)). \end{split}$$

Using this estimate and Remark 3.2.1, we get

$$\begin{split} \|f\|_{BMO_{\mu}(\Omega)} &= \sup_{B} \int_{B} \left| f(y) - \int_{B} f(s) d\mu(s) \right| d\mu(y) \\ &\leq \sup_{\mu(B)} C \frac{R(\mu(B)/c)}{\mu(B)\phi_{X}(\mu(B))} E_{X}(f, r(\mu(B)/c)) \\ &\leq \sup_{t>0} \sup_{t/C_{D} \leq \mu(B) \leq t} C \frac{R(\mu(B)/c)}{\mu(B)\phi_{X}(t/C_{D})} E_{X}(f, r(\mu(B)/c)) \\ &\leq \sup_{t>0} \frac{R(t/c)}{t\phi_{X}(t)} E_{X}(f, r(t/c)) \leq \sup_{t>0} \frac{R(t)}{t\phi_{X}(t)} E_{X}(f, r(t)). \end{split}$$

3.6.1 Essential continuity

Theorem 9. Let (Ω, d, μ) be a uniform (k, m)-space. Let X be an r.i. space on Ω . Now suppose $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$ satisfies

$$\int_0^\infty \frac{R(t)}{t\phi_X(t)} E_X(f,r(t)) \frac{dt}{t} < \infty.$$

Then f is μ -locally essentially continuous.

Proof. If $f \ge 0$, Theorem 5 implies

$$f_{\mu}^{\bigstar \bigstar}(t) - f_{\mu}^{\bigstar}(t) = f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \leq \frac{1}{\kappa_0 t} \frac{R(\kappa_0 t)}{\phi_X(\kappa_0 t)} E_X(f, r(\kappa_0 t)).$$

If f is bounded from below and $c = \inf(f)$, then $f - c \ge 0$, and therefore

$$O_{\mu}(f-c)(t) \leq \frac{1}{\kappa_{0}t} \frac{R(\kappa_{0}t)}{\phi_{X}(\kappa_{0}t)} E_{X}(f-c,r(\kappa_{0}t))$$
$$\leq \frac{1}{\kappa_{0}t} \frac{R(\kappa_{0}t)}{\phi_{X}(\kappa_{0}t)} E_{X}(f,r(\kappa_{0}t)).$$

By Proposition 2.1.1 (ix),

$$f_{\mu}^{\bigstar \bigstar}(t) - f_{\mu}^{\bigstar}(t) = (f - c)_{\mu}^{**}(t) - (f - c)_{\mu}^{*}(t),$$

and thus

$$f_{\mu}^{\bigstar \bigstar}(t) - f_{\mu}^{\bigstar}(t) \leq \frac{1}{\kappa_0 t} \frac{R(\kappa_0 t)}{\phi_X(\kappa_0 t)} E_X(f, r(\kappa_0 t)).$$

Let $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$, and let *B* be a ball. Given $n \in \mathbb{N}$, we consider $f_n = \max(f\chi_B, -n)$. Since f_n is bounded from below, we get

$$(f_n)^{\bigstar \bigstar}_{\mu}(t) - (f_n)^{\bigstar}_{\mu}(t) \leq \frac{1}{\kappa_0 t} \frac{R(\kappa_0 t)}{\phi_X(\kappa_0 t)} E_X(f_n, r(\kappa_0 t))$$
$$\leq \frac{1}{\kappa_0 t} \frac{R(\kappa_0 t)}{\phi_X(\kappa_0 t)} E_X(f, r(\kappa_0 t)).$$

Let $0 < a < \mu(B)$. By the Fundamental Theorem of Calculus,

$$\int_{a}^{\mu(B)} \left((f_n)_{\mu}^{\bigstar \bigstar}(t) - (f_n)_{\mu}^{\bigstar}(t) \right) \frac{dt}{t} = \frac{1}{a} \int_{0}^{a} (f_n)_{\mu}^{\bigstar}(t) dt - \frac{1}{\mu(B)} \int_{0}^{\mu(B)} (f_n)_{\mu}^{\bigstar}(t) dt$$

Since $f_n(z) \to f\chi_B(z)$ μ -a.e. and $|f_n| \le |f\chi_B|$, we have

$$\frac{1}{a} \int_{0}^{a} (f_{n})_{\mu}^{\ddagger}(t) dt - \frac{1}{\mu(B)} \int_{0}^{\mu(B)} (f_{n})_{\mu}^{\ddagger}(t) dt \coloneqq (F)$$

$$(F) \xrightarrow[n \to \infty]{a} \frac{1}{a} \int_{0}^{a} (f\chi_{B})_{\mu}^{\ddagger}(t) dt - \frac{1}{\mu(B)} \int_{0}^{\mu(B)} (f\chi_{B})_{\mu}^{\ddagger}(t) dt$$

Letting $a \to 0$, we get

$$(f\chi_B)^{\bigstar \bigstar}_{\mu}(0) - (f\chi_B)^{\bigstar \bigstar}_{\mu}(\mu(B)) \leq \int_0^{\mu(B)} \frac{1}{\kappa_0 t} \frac{R(\kappa_0 t)}{\phi_X(\kappa_0 t)} E_X(f, r(\kappa_0 t)) \frac{dt}{t}$$
$$\leq \int_0^{k\mu(B)} \frac{R(t)}{t\phi_X(t)} E_X(f, r(t)) \frac{dt}{t}.$$

By (2.1.1),

$$ess \sup f\chi_B - \frac{1}{\mu(B)} \int_0^{\mu(B)} (f\chi_B)^{\bigstar}_{\mu}(t) dt$$
$$\leq \int_0^{\kappa_0 \mu(B)} \frac{R(t)}{t\phi_X(t)} E_X(f, r(t)) \frac{dt}{t}.$$

Similarly, considering $-f\chi_B$ instead of $f\chi_B$, we obtain

$$\frac{1}{\mu(B)} \int_0^{\mu(B)} (-f\chi_B)_{\mu}^{\bigstar}(s) ds - ess \inf(f\chi_B)$$
$$\leq \int_0^{\kappa_0 \mu(B)} \frac{R(t)}{t\phi_X(t)} E_X(f, r(t)) \frac{dt}{t}.$$

Since $f\chi_B$ and $-f\chi_B$ are both supported on B, we have that

$$\int_{0}^{\mu(B)} (f\chi_{G})_{\mu}^{\bigstar}(s) ds = \int_{B} f d\mu \text{ and } \int_{0}^{\mu(B)} (-f\chi_{G})_{\mu}^{\bigstar}(s) ds = -\int_{B} f d\mu.$$

Adding these results, we have that for $\mu\text{-almost every } x,y\in B$

$$\begin{aligned} |f(x) - f(y)| &\leq ess \sup(f\chi_B) - ess \inf(f\chi_B) \\ &\leq 2 \int_0^{\kappa_0 \mu(B)} \frac{R(t)}{t\phi_X(t)} E_X(f, r(t)) \frac{dt}{t}. \end{aligned}$$

and μ -locally essentially continuity follows.

3.7 Sobolev type embeddings for homogeneous Besov spaces

In this section we going to consider in detail Sobolev type embeddings for the homogeneous Besov spaces $\dot{\mathcal{B}}_{p,q}^{s}(\Omega)$ where 0 .

First of all, note that an elementary computation shows (see Remark 3.4.1) that

$$\|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)} \simeq \begin{cases} \left(\int_{0}^{\infty} \left(\frac{E_{p}(f,r(t))}{r(t)^{s}}\right)^{q} \frac{dt}{t}\right)^{1/q}, & 0 < q < \infty, \\ \sup_{t>0} r(t)^{-s} E_{p}(f,r(t)), & q = \infty. \end{cases}$$

In case $1 \leq p < \infty$ and $1 \leq q \leq \infty$, our results will be a consequence of the theory developed in the previous sections. However, for 0 < p, q < 1, $L^p_{\mu}(\Omega)$ and $L^q([0,\infty))$ are not Banach spaces, thus the previous theory cannot be applied.

Lemma 3.7.1. Let $0 . Let <math>f \in L^p_\mu(\Omega) + L^\infty_\mu(\Omega)$. Then:

1. If (Ω, d, μ) is uniform, then for all t > 0 we have that

$$O_{\mu}(|f|^{p},t) \leq \frac{R(\kappa_{0}t)}{(\kappa_{0}t)^{2}} E_{p}(f,\kappa_{0}t)^{p}$$

2. If (Ω, d, μ) is bounded from below, then for all t > 0 we have that

$$O_{\tilde{\mu}}(|f|^{p},t) \leq \frac{R(\kappa_{1}t)}{(\kappa_{1}t)^{2}}E_{p}(f,r(\kappa_{1}t))^{p}.$$

Proof. Let B = B(x) be a ball centred at x. Since 0 , we have that

$$|f(x)|^{p}\chi_{B(x)}(y) \leq |f(x) - f(y)|^{p}\chi_{B(x)}(y) + |f(y)|^{p}\chi_{B(x)}(y).$$

Integrating with respect to $d\mu(y)$, we have that

$$\begin{split} |f(x)|^{p}\mu(B) &\leq \int_{B(x)} |f(x) - f(y)|^{p} d\mu(y) + \int_{B(x)} |f(y)|^{p} d\mu(y) \\ &\leq \int_{B(x)} |f(x) - f(y)|^{p} d\mu(y) + \int_{0}^{\mu(B)} (|f|^{p})^{*}_{\mu}(s) ds \quad (by \ (2.1.3)). \end{split}$$

Now integrating with respect to $d\mu(x)$ over a subset $E \subset \Omega$ with $\mu(E) = \mu(B)/2$, we get

$$\begin{split} \int_{E} |f(x)|^{p} d\mu(x) &\leq \int_{E} \int_{B(x)} |f(x) - f(y)|^{p} d\mu(y) d\mu(x) + \int_{E} \frac{1}{\mu(B)} \left(\int_{0}^{\mu(B)} f_{\mu}^{*}(s) ds \right) d\mu(x) \\ &\leq \int_{\Omega} \int_{B(x)} |f(x) - f(y)|^{p} d\mu(y) d\mu(x) + \frac{1}{2} \int_{0}^{\mu(B)} f_{\mu}^{*}(s) ds \end{split}$$

By (2.2.2), taking the supremum over all such sets E, we obtain

$$\int_{0}^{\mu(B)/2} \left(|f|^{p} \right)_{\mu}^{*}(s) ds \leq \int_{\Omega} \oint_{B(x)} |f(x) - f(y)|^{p} d\mu(y) d\mu(x) + \frac{1}{2} \int_{0}^{\mu(B)} \left(|f|^{p} \right)_{\mu}^{*}(s) ds.$$

Equivalently,

$$(|f|^{p})_{\mu}^{**}(\mu(B)/2) - (|f|^{p})_{\mu}^{**}(\mu(B)) \leq \frac{1}{\mu(B)} \int_{\Omega} \int_{B(x)} |f(x) - f(y)|^{p} d\mu(y) d\mu(x)$$

Now (i) and (ii) follow in the same way as Theorem 5.

Definition 3.7.1. Suppose $0 and <math>0 < q \le \infty$ and let v be a weight on $(0,\infty)$. The space $S^{p,q}_{\mu}(v)$ is the collection of all μ -measurable functions such that $\|f\|_{S^{p,q}_{\mu}(v)} < \infty$, where

$$\|f\|_{S^{p,q}_{\mu}(v)} = \left(\int_0^\infty O_{\mu}(|f|^p, t)^{\frac{q}{p}}v(t)dt\right)^{1/q}$$

Remark 3.7.1. For p = 1 the spaces $S^{1,q}(v)$ were introduced in [14]. Note that if $1 \le p < \infty$ and $1 \le q \le \infty$, then

$$S^{L^p,L^q}_{\mu}(v_s) = S^{1,q}_{\mu}(v_s)$$

Corollary 10. Let (Ω, d, μ) be a (k, m)-space. Let 0 < s < 1 and $0 , <math>0 < q \le \infty$. Let

$$v(t) = \min\left(t^{1+\frac{1}{\max(1,p)} - \frac{m+s\min(1,p)}{k}}, t^{1+\frac{1}{\max(1,p)} - \frac{k+s\min(1,p)}{m}}\right)^{\frac{q}{\min(1,p)}} \frac{1}{t}.$$

Then:

(i.) If (Ω, d, μ) is uniform, then

$$\dot{\mathcal{B}}^{s}_{p,q}(\Omega) \subset S^{\min(1,p),q}_{\mu}(v).$$

(ii.) If (Ω, d, μ) is bounded from below, then

$$\dot{\mathcal{B}}^{s}_{p,q}(\Omega) \subset S^{\min(1,p),q}_{\tilde{\mu}}(v).$$

Proof. Part (i.) In the case $1 \le p < \infty$ the proof given in Theorem 6 works. In case 0 , then from Lemma 3.7.1 it follows that

$$\left(\frac{t^2}{t^{p/q}R(t)r(t)^{sp}}O_{\mu}(|f|^p,t)\right)^{1/p} \le \frac{r(\kappa_0 t)^{-s}E_p(f,r(\kappa_0 t))}{(\kappa_0 t)^{1/q}}$$

and the result is obtained by taking $L^q([0,\infty))$ -(quasi) norms on both sides.

Part (ii) can be proved in the same way.

The following lemma will be useful in what follows.

Lemma 3.7.2. (see [5, Lemma 5.4]). Let $1 \le q < \infty$, and suppose that (w, v) is a pair of weights satisfying the following condition: there exists C > 0 such that for all 0 < t < 1,

$$\left(\int_{0}^{t} w(s)ds\right)^{1/q} \left(\int_{t}^{1} \frac{v(s)^{\frac{-1}{q-1}}}{s^{\frac{q}{q-1}}}ds\right)^{(q-1)/q} \le C.$$

Then

$$\left(\int_{0}^{1} f_{\mu}^{**}(s)^{q} w(s) ds\right)^{1/q} \leq \left(\int_{0}^{1} \left(f_{\mu}^{**}(s) - f_{\mu}^{*}(s)\right)^{q} v(s) ds\right)^{1/q} + \left(\int_{0}^{1} w(s) ds\right)^{1/q} \int_{0}^{1} f_{\mu}^{*}(t) ds.$$

Lemma 3.7.3. Let 0 < q < 1 and b > 0, then

$$\left(\int_0^1 t^b f_{\mu}^{**}(t)^q \frac{dt}{t}\right)^{1/q} \leq \left(\int_0^1 t^b \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t)\right)^q \frac{dt}{t}\right)^{1/q} + f_{\mu}^{**}(1).$$

Proof. We integrate by parts and obtain

$$\int_{0}^{1} t^{b} f_{\mu}^{**}(t)^{q} \frac{dt}{t} = \frac{1}{b} \left[t^{b} f_{\mu}^{**}(t)^{q} \right]_{0}^{1} + \frac{q}{b} \int_{0}^{1} t^{b} f_{\mu}^{**}(t)^{q-1} \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \right) \frac{dt}{t}$$
$$\leq \frac{1}{b} \left[t^{b} f^{**}(t)^{q} \right]_{0}^{1} + \frac{q}{b} \int_{0}^{1} t^{b} \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \right)^{q} \frac{dt}{t} \text{ (since } q < 1 \text{)}$$

Now

$$\left[t^b f_{\mu}^{**}(t)^q\right]_0^b = f_{\mu}^{**}(1)^q - \lim_{t \to 0} t^b f_{\mu}^{**}(t)^q.$$

To finish the proof we need to show that the previous limit is finite. This is obvious if $f_{\mu}^{**}(0) < \infty$. If $f_{\mu}^{**}(0) = \infty$, taking into account that $t(f_{\mu}^{**}(t) - f_{\mu}^{*}(t))$ is increasing, we get

$$t\left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t)\right)\left(\int_{t}^{1} s^{b-1-q}\right)^{1/q} \leq \left(\int_{t}^{1} s^{q} \left(f_{\mu}^{**}(s) - f_{\mu}^{*}(s)\right)^{q} s^{b-1-q}\right)^{1/q}$$
$$= \left(\int_{t}^{1} s^{b} \left(f_{\mu}^{**}(s) - f_{\mu}^{*}(s)\right)^{q} \frac{ds}{s}\right)^{1/q}.$$

If t < 1/2, then

$$\left(\int_{t}^{1} s^{b-1-q}\right)^{1/q} \ge \left(\int_{t}^{2t} s^{b-1-q}\right)^{1/q} \simeq t^{b/q-1}, \text{ if } b \neq q,$$

and

$$\left(\int_{t}^{1} s^{b-1-q}\right)^{1/q} \ge \left(\int_{t}^{2t} \frac{1}{s}\right)^{1/q} \simeq 1, \text{ if } b = q.$$

Thus

$$t^{b/q} \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \right) \leq \left(\int_{t}^{1} s^{b} \left(f_{\mu}^{**}(s) - f_{\mu}^{*}(s) \right)^{q} \frac{ds}{s} \right)^{1/q}$$

Finally, by L'Hopital's rule,

$$\lim_{t \to 0} t^{b/q} f_{\mu}^{**}(t) = \lim_{t \to 0} \frac{f_{\mu}^{**}(t)}{t^{-b/q}} = \lim_{t \to 0} \frac{-(f_{\mu}^{**}(t) - f_{\mu}^{*}(t))/t}{-\frac{b}{q}t^{-b/q-1}}$$
$$= \lim_{t \to 0} \frac{t^{b/q} (f_{\mu}^{**}(t) - f_{\mu}^{*}(t))}{b/q} \leq \left(\int_{0}^{1} s^{b} (f_{\mu}^{**}(s) - f_{\mu}^{*}(s))^{q} \frac{ds}{s}\right)^{1/q}.$$

Lemma 3.7.4. Given $a < b < \infty$, we define

$$v(t) = \frac{\min(t^a, t^b)}{t}.$$

Let $0 < q \le \infty$ and $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$, with $f^*_{\mu}(\infty) = 0$. Then

(i.) If $0 < a < b < \infty$, then

$$\|f\|_{S^{1,q}_{\mu}(v)} \simeq \left(\int_0^\infty f^{**}_{\mu}(t)^q v(t) dt\right)^{1/q}.$$

(ii.) If $a \leq 0$, then

$$\left(\int_0^1 f_{\mu}^{**}(t)^q v(t) dt\right)^{1/q} \leq \|f\|_{S_{\mu}^{1,q}(v)} + f_{\mu}^{**}(1).$$

(iii.) If b = 0 and q > 1, then

$$\left(\int_0^\infty \left(\frac{f_{\mu}^{**}(t)^q}{1+\ln\frac{1}{t}}\right)^q \frac{dt}{t}\right)^{1/q} \le \|f\|_{S_{\mu}^{1,q}(v)} + f_{\mu}^{**}(1).$$

(iv.) If b = 0 and $q \le 1$ or b < 0 and $0 < q \le \infty$, then

$$||f||_{\infty} \leq ||f||_{S^{1,q}_{\mu}(v)} + f^{**}_{\mu}(1).$$

Proof. (i.) By [14, Corollary 4.3.] we only need to check that

$$\int_0^r v(t)dt \le r^q \int_r^\infty \frac{v(t)}{t^q} dt, \quad r > 0.$$

Pick $0 < \varepsilon < a$. Then

$$\begin{split} \int_0^r v(t)dt &= \int_0^r \min(t^{a-\varepsilon}, t^{b-\varepsilon}) \frac{dt}{t^{1-\varepsilon}} \leq \min(r^{a-\varepsilon}, r^{b-\varepsilon}) \int_0^r \frac{dt}{t^{1-\varepsilon}} \\ &\leq \min(r^a, r^b) \leq \min(r^a, r^b) \frac{r^q}{r^q} \\ &\leq \min(r^a, r^b) r^q \int_r^{2r} \frac{dt}{s^{q+1}} \leq r^q \int_r^{2r} v(t) \frac{dt}{s^q} \\ &\leq r^q \int_r^\infty \frac{v(t)}{t^q} dt. \end{split}$$

(ii.) By Lemma 3.7.3,

$$\left(\int_0^1 t^b f_{\mu}^{**}(t)^q \frac{dt}{t}\right)^{1/q} \leq \left(\int_0^1 t^b O_{\mu}(f,t)^q \frac{dt}{t}\right)^{1/q} + f_{\mu}^{**}(1).$$

(iii.) By Lemma 3.7.2 with $w(t) = \left(\frac{1}{1+\ln(\frac{1}{t})}\right)^q \frac{1}{s}$ and $v(t) = \frac{1}{t}$, we get

$$\left(\int_{0}^{1} \left(\frac{f_{\mu}^{**}(t)}{1+\ln\left(\frac{1}{t}\right)}\right)^{q} \frac{dt}{t}\right)^{1/q} \leq \left(\int_{0}^{1} O_{\mu}(f,t)^{q} \frac{dt}{t}\right)^{1/q} + f_{\mu}^{**}(1)$$
$$\leq \|f\|_{S_{\mu}^{1,q}(v)} + f_{\mu}^{**}(1)$$

and

$$\left(\int_{1}^{\infty} \left(\frac{f_{\mu}^{**}(t)}{1+\ln\left(\frac{1}{t}\right)}\right)^{q} \frac{dt}{t}\right)^{1/q} \le f_{\mu}^{**}(1) \left(\int_{1}^{\infty} \left(\frac{1}{1+\ln\left(\frac{1}{t}\right)}\right)^{q} \frac{dt}{t}\right)^{1/q} \le f_{\mu}^{**}(1).$$

(iv.) If b = 0 and q = 1, then

$$||f||_{\infty} = f_{\mu}^{**}(0) = \int_{0}^{1} O_{\mu}(f,t) \frac{dt}{t} + f_{\mu}^{**}(1) \le ||f||_{S_{\mu}^{1,q}(v)} + f_{\mu}^{**}(1).$$

If 0 < q < 1, let 0 < r < 1. Then

$$f_{\mu}^{**}(r) - f_{\mu}^{**}(1) = \int_{r}^{1} O_{\mu}(f,t) \frac{dt}{t} = f_{\mu}^{**}(r)^{1-q} \int_{r}^{1} \frac{O_{\mu}(f,t)}{f_{\mu}^{**}(r)^{1-q}} \frac{dt}{t}$$
(3.7.1)

$$\leq \int_{r}^{1} \frac{O_{\mu}(f,t)}{f_{\mu}^{**}(t)^{1-q}} \frac{dt}{t} \leq \int_{r}^{1} \frac{O_{\mu}(f,t)}{O_{\mu}(f,t)^{1-q}} \frac{dt}{t}$$

$$= f_{\mu}^{**}(r)^{1-q} \int_{r}^{1} O_{\mu}(f,t)^{q} \frac{dt}{t}$$

$$\leq (1-q) f_{\mu}^{**}(r) + q \left(\int_{0}^{1} O_{\mu}(f,t)^{q} \frac{dt}{t}\right)^{1/q}$$
(3.7.2)

thus

$$qf_{\mu}^{**}(r) \le q\left(\int_{0}^{1} O_{\mu}(f,t)^{q} \frac{dt}{t}\right)^{1/q} + f_{\mu}^{**}(1)$$

which implies

$$||f||_{\infty} = f_{\mu}^{**}(0) \le \left(\int_{0}^{1} O_{\mu}(f,t)^{q} \frac{dt}{t}\right)^{1/q} + \frac{f_{\mu}^{**}(1)}{q}.$$

If b < 0 and $1 \le q \le \infty$, then

$$\begin{aligned} f_{\mu}^{**}(0) - f_{\mu}^{**}(1) &= \int_{0}^{1} O_{\mu}(f, t) \frac{dt}{t} \\ &\leq \left(\int_{0}^{1} \left(t^{b} O_{\mu}(f, t) \right)^{q} \frac{dt}{t} \right)^{1/q} \left(\int_{0}^{1} \left(t^{-b} \right)^{\frac{q}{q-1}} \frac{dt}{t} \right)^{\frac{q-1}{q}} \\ &\leq \left(\int_{0}^{1} \left(t^{b} O_{\mu}(f, t) \right)^{q} \frac{dt}{t} \right)^{1/q}. \end{aligned}$$

If b < 0 and 0 < q < 1, then

$$f_{\mu}^{**}(0) - f_{\mu}^{**}(1) = \int_{0}^{1} O_{\mu}(f, t) \frac{dt}{t} \le \int_{0}^{1} t^{b} O_{\mu}(f, t) \frac{dt}{t}$$

and we finish the proof in the same way as in (3.7.1).

Now we are ready to establish our Sobolev embedding theorem for homogeneous Besov spaces $\dot{\mathcal{B}}^s_{p,q}(\Omega)$. Motivated by the classical theory, we will distinguish three cases: The subcritical case, when there is an embedding into a Lorentz type space; the critical case, when $\dot{\mathcal{B}}^{s}_{p,q}(\Omega)$ is embedded into a logarithmic Lorentz space; and the supercritical case, when the Besov space is embedded into L^{∞} .

Theorem 11. Let (Ω, d, μ) be a uniform (k, m)-space. Let 0 < s < 1, $0 , <math>0 < q \le \infty$ and $f \in L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$, with $\left(|f|^{\min(1,p)}\right)^{*}_{\mu}(\infty) = 0$. Then:

1. Subcritical case:

a) If
$$s \min(1, p) < k(1 + \frac{1}{\max(1, p)}) - m$$
, then
 $\|f\|_{L^{\alpha,q}_{\mu}(\Omega) + L^{\beta,q}_{\mu}(\Omega)} \leq \|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)}$,
where
 $\frac{\min(1, p)}{\alpha} = 1 + \frac{1}{\max(1, p)} - \frac{k + s \min(1, p)}{m}$,

and

$$\frac{\min(1,p)}{\beta} = 1 + \frac{1}{\max(1,p)} - \frac{m + s\min(1,p)}{k}.$$

b) If
$$k(1 + \frac{1}{\max(1,p)}) - m < s \min(1,p) < m(1 + \frac{1}{\max(1,p)}) - k$$
, then

$$\left(\int_{0}^{1} \left(t^{\frac{1}{\alpha}}f_{\mu}^{**}(t)\right)^{q} \frac{dt}{t}\right)^{1/q} \leq \|f\|_{\dot{B}^{s}_{p,q}(\Omega)} + \|f\|_{L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)}.$$
(3.7.3)

Here,

$$\frac{\min(1,p)}{\alpha} = 1 + \frac{1}{\max(1,p)} - \frac{m + s\min(1,p)}{k}.$$

2. Critical case:

If $s \min(1, p) = m(1 + \frac{1}{\max(1, p)}) - k$, then

a) If q > 1, we get

$$\left(\int_{0}^{\infty} \left(\frac{f_{\mu}^{**}(t)}{1+\ln\left(\frac{1}{t}\right)}\right)^{q} \frac{dt}{t}\right)^{1/q} \leq \|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)} + \|f\|_{L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)}.$$

b) If $0 < q \le 1$, we get

$$\|f\|_{\infty} \leq \|f\|_{\dot{B}^{s}_{p,q}(\Omega)} + \|f\|_{L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)}.$$

3. Supercritical case:

$$\begin{split} If s \min(1,p) > m(1 + \frac{1}{\max(1,p)}) - k, \ then \\ \|f\|_{\infty} \le \|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)} + \|f\|_{L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)}. \end{split}$$



Figure 3.2: Theorem 11.

Proof. The proof follows from Lemma 3.7.4. We will employ (3.7.3) if 0 . Then

$$v(t) = \min\left(t^{2-\frac{m+sp}{k}}, t^{2-\frac{k+sp}{m}}\right)^{\frac{q}{p}} \frac{1}{t}$$

with $2 - \frac{m+sp}{k} \leq 0$. Now by Lemma 3.7.4, applied to $|f|^p$ and q/p, we have that

$$\left(\int_{0}^{1} \left(\frac{1}{t} \int_{0}^{t} f_{\mu}^{*}(s)^{p} ds\right)^{q/p} t^{\left(2-\frac{k+sp}{m}\right)\frac{q}{p}} \frac{dt}{t}\right)^{p/q} \leq \left\|\left|f\right|^{p}\right\|_{S^{1,q/p}_{\mu}(v)} + \left(\left|f\right|^{p}\right)^{**}_{\mu}(1).$$
Obviously

$$\left(\int_0^1 \left(f_{\mu}^*(s)t^{\left(2-\frac{k+sp}{m}\right)\frac{1}{p}}\right)^q dt\right)^{p/q} \le \left(\int_0^1 \left(\frac{1}{t}\int_0^t f_{\mu}^*(s)^p ds\right)^{q/p} t^{\left(2-\frac{k+sp}{m}\right)\frac{q}{p}} \frac{dt}{t}\right)^{p/q}.$$

Since

$$\begin{aligned} \||f|^{p}\|_{S^{1,q/p}_{\mu}(v)} &= \left(\int_{0}^{\infty} O_{\mu}(|f|^{p},t)^{\frac{q}{p}}v(t)dt\right)^{p/q} = \|f\|_{S^{p,q}_{\mu}(v)}^{p} \\ &\leq \|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)}^{p} \text{ (by Corollary 10),} \end{aligned}$$

we have that

$$\left(\int_{0}^{1} \left(f_{\mu}^{*}(s)t^{\left(2-\frac{k+sp}{m}\right)\frac{1}{p}}\right)^{q} dt\right)^{p/q} \leq \|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)}^{p} + \left(|f|^{p}\right)^{**}_{\mu}(1)$$

and thus

$$\left(\int_0^1 \left(f_{\mu}^*(s) t^{\left(2 - \frac{k + s_p}{m}\right)\frac{1}{p}} \right)^q dt \right)^{1/q} \le \|f\|_{\dot{\mathcal{B}}^s_{p,q}(\Omega)} + \left(\left(|f|^p\right)^{**}_{\mu}(1) \right)^{1/p}$$
$$= \|f\|_{\dot{\mathcal{B}}^s_{p,q}(\Omega)} + \|f\|_{L^p_{\mu}(\Omega) + L^\infty_{\mu}(\Omega)} .$$

All the other cases can be proved in the same way.

With the same proof as that of Theorem 11, we obtain the following theorem.

Theorem 12. Let (Ω, d, μ) be a (k, m)-space bounded from below. Suppose $f \in L^1_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$. Then Theorem 11 holds, considering f^{**}_{μ} and f^*_{μ} instead of f^{**}_{μ} and f^*_{μ} .

Proof. The proof is done using the arguments of Lemma 3.2.3, Theorem 5, and Corollary 6. $\hfill \Box$

By Theorems 8, 9 and 7, we obtain

Corollary 13. Let (Ω, d, μ) be a uniform (k, m)-space. Then

$$\|f\|_{BMO_{\mu}(\Omega)} \leq \sup_{0 < t < 1} t^{k-m(1+1/p)} E_p(f,t) + \sup_{t > 1} t^{m-k(1+1/p)} E_p(f,t).$$

(ii.) If

(i.)

$$\int_0^1 t^{k-m(1+1/p)} E_p(f,t) \frac{dt}{t} + \int_1^\infty t^{m-k(1+1/p)} E_p(f,t) \frac{dt}{t} < \infty,$$

then f is μ -locally essentially continuous.

(iii.) Let $s < k(1 + \frac{1}{p}) - m$, and let w > 0 be such that

$$[w] := \sup_{t>0} \left(\left(\left(\frac{1}{w}\right)_{\mu}^{*}(t) \right)^{s} \frac{1}{\min\left(t^{1-\frac{m+s}{k}}, t^{1-\frac{k+s}{m}}\right) t^{\frac{1}{p}-\frac{1}{q}}} \right) < \infty.$$

Then, for all $\alpha > 0$ and $q \ge 1$, we have that

$$\|f\|_{L^q} \le \left(2\kappa \left[w\right]\right)^{\frac{\alpha}{\alpha+1}} \|f\|_{\dot{\mathcal{B}}^{s,q}_{p,q}(\Omega)}^{\frac{\alpha}{\alpha+1}} \|w^{\alpha s}f\|_{L^q}^{\frac{1}{\alpha+1}}.$$

Now, we collect the results for the case when (Ω, d, μ) is Q-regular.

Theorem 14. Let (Ω, d, μ) be Q-regular. Let $0 , <math>0 < q \le \infty$, 0 < s < 1, and $f \in L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)$ with $\left(|f|^{\min(1,p)}\right)^{*}_{\mu}(\infty) = 0$. Then:

(i.) Subcritical case $s < \frac{Q}{p}$:

$$\|f\|_{L^{p(Q),q}_{\mu}(\Omega)} \leq \|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)}$$

where p(Q) = Qp/(Q - sp).

Moreover, let w > 0 be such that

$$[w] \coloneqq \sup_{t>0} \left(\left(\left(\frac{1}{w}\right)_{\mu}^{*}(t) \right)^{s} \frac{1}{t^{\frac{1}{p} - \frac{1}{q} - \frac{s}{Q}}} \right) < \infty.$$

If $1 \le p, q < \infty$, then for all $\alpha > 0$, we have that

$$\|f\|_{L^q} \leq [w]^{\frac{\alpha}{\alpha+1}} \|f\|_{\dot{\mathcal{B}}^s_{p,q}(\Omega)}^{\frac{\alpha}{\alpha+1}} \|w^{\alpha s} f\|_{L^q}^{\frac{1}{\alpha+1}}.$$

(ii.) Critical case $s = \frac{Q}{p}$:

a) If
$$q > 1$$
, then

$$\left(\int_{0}^{\infty} \left(\frac{f_{\mu}^{**}(t)}{1+\ln\left(\frac{1}{t}\right)}\right)^{q} \frac{dt}{t}\right)^{1/q} \leq \|f\|_{\dot{\mathcal{B}}^{Q/p}_{p,q}(\Omega)} + \|f\|_{L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)}$$

b) If $0 < q \le 1$, then

$$\|f\|_{\infty} \leq \|f\|_{\dot{\mathcal{B}}^{s}_{p,q}(\Omega)} + \|f\|_{L^{\min(1,p)}_{\mu}(\Omega) + L^{\infty}_{\mu}(\Omega)}.$$

c) If $p \ge 1$, we get: i.

$$\|f\|_{BMO_{\mu}(\Omega)} \leq \|f\|_{\dot{\mathcal{B}}^{Q/p}_{p,\infty}(\Omega)}.$$

ii. If $f \in ||f||_{\dot{B}^{Q/p}_{p,1}(\Omega)}$, then f is μ -locally essentially continuous.

(iii.) Supercritical case $s > \frac{Q}{p}$



Figure 3.3: Theorem 14

Chapter 4

Symmetrization inequalities for probability metric spaces with convex isoperimetric profile

4.1 Introduction

Let (Ω, d, μ) be a connected metric space equipped with a separable Borel probability measure μ . The perimeter or Minkowski content of a Borel set $A \subset \Omega$ is defined by

$$\mu^{+}(A) = \liminf_{h \to 0} \frac{\mu(A_{h}) - \mu(A)}{h},$$

where $A_h = \{x \in \Omega : d(x, A) < h\}$ is an open *h*-neighbourhood of *A*. The **isoperimetric profile** I_{μ} is defined as the pointwise maximal function $I_{\mu} : [0, 1] \rightarrow [0, \infty)$ such that

$$\mu^+(A) \ge I_\mu(\mu(A)),$$

holds for all Borel sets A. An isoperimetric inequality measures the relation between the boundary measure and the measure of a set, by providing a lower bound on I_{μ} by some function $I:[0,1] \rightarrow [0,\infty)$ which is not identically zero.

The modulus of the gradient of a Lipschitz function f on Ω (briefly $f \in Lip(\Omega)$) is defined by¹

$$|\nabla f(x)| = \limsup_{d(x,y) \to 0} \frac{|f(x) - f(y)|}{d(x,y)}.$$

The equivalence between isoperimetric inequalities and Poincaré inequalities was obtained by Maz'ya, whose method (see [70], [62] and [16]) shows that given

¹In fact one can define $|\nabla f|$ for functions f that are Lipschitz on every ball in (Ω, d) (cf. [7] for more details).

CHAPTER 4. SYMMETRIZATION INEQUALITIES FOR CONVEX PROFILE

 $X = X(\Omega)$ a rearrangement invariant space², the inequality

$$\left\| f - \int_{\Omega} f d\mu \right\|_{X} \le c \left\| |\nabla f| \right\|_{L^{1}}, \ f \in Lip(\Omega),$$

$$(4.1.1)$$

holds if, and only if, there exists a constant $c = c(\Omega) > 0$ such that for all Borel sets $A \subset \Omega$,

$$\min(\phi_X(\mu(A)), \phi_X(1 - \mu(A))) \le c\mu^+(A), \tag{4.1.2}$$

where $\phi_X(t)$ is the fundamental function³ of X :

$$\phi_X(t) = \|\chi_A\|_X$$
, with $\mu(A) = t$.

Motivated by this fact, we will say (Ω, d, μ) admits a concave isoperimetric estimator if there exists a function $I : [0, 1] \rightarrow [0, \infty)$ that is continuous, concave, increasing on (0, 1/2), symmetric about the point 1/2, satisfies I(0) = 0, and I(t) > 0 on (0, 1), such that

$$I_{\mu}(t) \ge I(t), \ 0 \le t \le 1.$$

In the recent work of Milman and Martín (see [61], [63]) it was proved that (Ω, d, μ) admits a concave isoperimetric estimator I if, and only if, the following symmetrization inequality holds,

$$f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \le \frac{t}{I(t)} |\nabla f|_{\mu}^{**}(t), \ (f \in Lip(\Omega))$$
(4.1.3)

where $f_{\mu}^{**}(t) = \frac{1}{t} \int_{0}^{t} f_{\mu}^{*}(s) ds$, and f_{μ}^{*} is the non-increasing rearrangement of f with respect to the measure μ . If we apply a rearrangement invariant function norm X on Ω (see Section 2.2) to (4.1.3) we obtain Sobolev–Poincaré type estimates of the form⁴

$$\left\| \left(f_{\mu}^{**}(t) - f_{\mu}^{*}(t) \right) \frac{I(t)}{t} \right\|_{\bar{X}} \leq \left\| |\nabla f|_{\mu}^{**} \right\|_{\bar{X}}.$$
(4.1.4)

Example 4.1.1. (See [64], [65].) Let $\Omega \subset \mathbb{R}^n$ be a Lipschitz domain of measure 1, $X = L^p(\Omega)$, $1 \le p \le n$, and p^* be the usual Sobolev exponent defined by $\frac{1}{p^*} = \frac{1}{p} - \frac{1}{n}$. Then

$$\left\| (f^{**}(t) - f^{*}(t)) \frac{I(t)}{t} \right\|_{L^{p}} \simeq \left\| (f^{**}(t) - f^{*}(t)) \right\|_{L^{p^{*,p}}}, \tag{4.1.5}$$

which follows from the fact that the isoperimetric profile is equivalent to $I(t) = c_n \min(t, 1-t)^{1-1/n}$ and from Hardy's inequality (here $L^{p^*,p}$ is a Lorentz space (see Section 2.2.2)). In the case where we consider \mathbb{R}^n with Gaussian measure

²i.e. such that if f and g have the same distribution function then $||f||_X = ||g||_X$ (see Section 2.2).

³We can assume with no loss of generality that ϕ_X is concave.

⁴The spaces \bar{X} were defined in Section 2.2.

 γ_n , and let $X = L^p$, $1 \le p < \infty$, then (compare with [34], [23]) since $I_{(\mathbb{R}^n, d, \gamma_n)}(t) \simeq t(\log 1/t)^{1/2}$ for t near zero, we have

$$\left\| \left(f_{\gamma_n}^{**}(t) - f_{\gamma_n}^{*}(t) \right) \frac{I(t)}{t} \right\|_{L^p} \simeq \left\| \left(f_{\gamma_n}^{**}(t) - f_{\gamma_n}^{*}(t) \right) \right\|_{L^p(Log)^{p/2}}, \tag{4.1.6}$$

where $L^p(\log L)^{p/2}$ is a Lorentz-Zygmund space (see Section 2.2.2).

In this fashion, in [61], [63], [64] and [65], Martín and Milman were able to provide a unified framework to study the classical Sobolev inequalities and logarithmic Sobolev inequalities. Moreover, the embeddings (4.1.4) turn out to be the best possible in all the classical cases. However the method used in the proof of the previous results cannot be applied with probability measures with heavy tails, since isoperimetric estimators of such measures are non concave.

Let us illustrate this phenomenon with some examples (see [15, Propositions 4.3 and 4.4] for examples 4.1.2 and 4.1.3 and [72] for example 4.1.4).

Example 4.1.2. (α -Cauchy type law). Let $\alpha > 0$. Consider the probability measure space (\mathbb{R}^n, d, μ) where d is the Euclidean distance and μ is defined by $d\mu(x) = V^{-(n+\alpha)}dx$ with $V : \mathbb{R}^n \to (0, \infty)$ convex. Then there exists a C > 0 such that for any measurable set $A \subset \mathbb{R}^n$

$$\mu^+(A) \ge C \min (\mu(A), 1 - \mu(A))^{1+1/\alpha}$$

Example 4.1.3. (Extended p-sub-exponential law). Let $p \in (0, 1)$. Consider the probability measure on \mathbb{R}^n defined by $d\mu(x) = (1/Z_p) e^{-V^p(x)} dx$ for some positive convex function $V : \mathbb{R}^n \to (0, \infty)$. Then there exists a C > 0 such that for any measurable set $A \subset \mathbb{R}^n$

$$\mu^{+}(A) \geq C \min(\mu(A), 1 - \mu(A)) \left(\log \frac{1}{\min(\mu(A), 1 - \mu(A))} \right)^{1 - 1/p}.$$

Example 4.1.4. Let (M^n, g, μ) be an n-dimensional weighted Riemannian manifold $(n \ge 2)$ that satisfies the CD(0, N) curvature condition with N < 0. Then for every Borel set $A \subset (M^n, g)$,

$$\mu^+(A) \ge C \min(\mu(A), 1 - \mu(A))^{-1/N}$$

Motivated by these examples, we will say (Ω, d, μ) admits a convex isoperimetric estimator if there exists a function $I : [0,1] \rightarrow [0,\infty)$ that is continuous, convex, increasing on (0, 1/2), symmetric about the point 1/2, such that I(0) = 0and I(t) > 0 on (0, 1), and satisfies

$$I_{\mu}(t) \ge I(t), \ 0 \le t \le 1.$$

CHAPTER 4. SYMMETRIZATION INEQUALITIES FOR CONVEX PROFILE

The purpose of this chapter is to obtain symmetrization inequalities on probability metric spaces that admit a convex isoperimetric estimator which incorporate in their formulation the isoperimetric estimator and that can be applied to provide a unified treatment of sharp Sobolev–Poincaré and Nash type inequalities. Note that if I is a convex isoperimetric estimator, then

$$I(t) \preceq \min\left(t, 1 - t\right).$$

Therefore (unless $I(t) \simeq \min(t, 1-t)$), the Poincaré inequality

$$\left\| f - \int_{\Omega} f d\mu \right\|_{L^{1}} \leq c \left\| |\nabla f| \right\|_{L^{1}}, f \in Lip(\Omega),$$

never holds, which means that we cannot use $|\nabla f| \in L^1$ to deduce that $f \in L^1$. Hence a symmetrization inequality like (4.1.3) will not be possible since f_{μ}^{**} is defined if, and only if, $f \in L^1$.



Figure 4.1: Isoperimetric profile

This chapter is organized as follows. In Section 4.2 we obtain symmetrization inequalities which incorporate in their formulation the isoperimetric convex estimator. In Section 4.3 we use the symmetrization inequalities to derive Sobolev–Poincaré and Nash type inequalities. Finally, in Section 4.4, we study in detail Examples 4.1.2, 4.1.3 and 4.1.4.

The results contained in this chapter have been submitted for publication (see [68]).

Definition 4.1.1. Let $f \in \mathcal{M}_0(\Omega)$. We say that m(f) is a median value of f if

$$\mu\{f \ge m(f)\} \ge \frac{1}{2} \text{ and } \mu\{f \le m(f)\} \ge \frac{1}{2}.$$

Lemma 4.1.1. $f^{\bigstar}_{\mu}(1/2)$ is a median value of f.

Proof. Definition 4.1.1 is equivalent to

 $\mu \{f > m(f)\} \le 1/2; \text{ and } \mu \{f < m(f)\} \le 1/2.$

Now,

$$\mu\left\{f < f_{\mu}^{*}(1/2)\right\} = \mu\left\{-f > -f_{\mu}^{*}(1/2)\right\},\$$

but from

$$(-f)^*_{\mu}(t) = -f^*_{\mu}(1-t)$$

it follows that

$$(-f)^*_{\mu}(1/2) = -f^*_{\mu}(1/2).$$

Consequently

$$\mu \left\{ f < f_{\mu}^{*} (1/2) \right\} = \mu \left\{ -f > -f_{\mu}^{*} (1/2) \right\}$$

= $\mu \left\{ -f > (-f)_{\mu}^{*} (1/2) \right\}$
 $\leq 1/2.$

Therefore $f_{\mu}^{\dot{\alpha}}\left(\frac{1}{2}\right)$ is a median value as was to be shown.

Remark 4.1.1. If f has zero median and f_{μ}^{\ddagger} is continuous, then $f_{\mu}^{\ddagger}\left(\frac{1}{2}\right) = 0$.

4.2 Symmetrization and Isoperimetry

We will assume in what follows that (Ω, d, μ) is a connected measure metric space equipped with with a separable, non-atomic, Borel probability measure μ which admits a convex isoperimetric estimator.

In order to balance generality with power and simplicity, we will assume throughout the paper that our spaces satisfy the following condition.

Condition 4.2.1. We assume that Ω is such that for every $f \in Lip(\Omega)$ and every $c \in \mathbb{R}$ we have that $|\nabla f(x)| = 0$, a.e. on the set $\{x : f(x) = c\}$.

Theorem 15. Let $I : [0,1] \rightarrow [0,\infty)$ be a convex isoperimetric estimator. The following statements are equivalent:

(i.) Isoperimetric inequality: for all Borel sets $A \subset \Omega$,

$$\mu^{+}(A) \ge I(\mu(A)). \tag{4.2.1}$$

CHAPTER 4. SYMMETRIZATION INEQUALITIES FOR CONVEX PROFILE

(ii.) Ledoux's inequality (cf [57]): for all $f \in Lip(\Omega)$,

$$\int_{-\infty}^{\infty} I(\mu_f(s)) \le \int_{\Omega} |\nabla f(x)| \, d\mu. \tag{4.2.2}$$

(iii.) For all functions $f \in Lip(\Omega)$, f^{\bigstar}_{μ} is locally absolutely continuous, and

$$\int_{0}^{t} ((-f_{\mu}^{\bigstar})'I(s))^{*} ds \leq \int_{0}^{t} |\nabla f|_{\mu}^{*}(s) ds.$$
(4.2.3)

(The second rearrangement on the left hand side is with respect to Lebesgue measure).

(iv.) Bobkov's inequality (cf. [7]): For all bounded $f \in Lip(\Omega)$ with m(f) = 0, and for all s > 0,

$$\int_{\Omega} |f(x)| d\mu \leq \beta_1(s) \int_{\Omega} |\nabla f(x)| d\mu + s Osc_{\mu}(f), \qquad (4.2.4)$$

where
$$Osc_{\mu}(f) = ess \sup f - ess \inf f$$
, and $\beta_1(s) = \sup_{s < t \le 1/2} \frac{t - s}{I(t)}$.

Proof. $(i.) \rightarrow (ii.)$ By the co-area inequality applied to f (cf. [8, Lemma 3.1]) and the isoperimetric inequality (4.2.1), it follows that

$$\int_{\Omega} |\nabla f(x)| d\mu \ge \int_{-\infty}^{\infty} \mu^+(\{f > s\}; \Omega) ds$$
$$\ge \int_{0}^{\infty} I(\mu_f(s)) ds .$$

 $(ii.) \rightarrow (iii.)$ Let $-\infty < t_1 < t_2 < \infty$. The smooth truncations of f are defined by

$$f_{t_1}^{t_2}(x) = \begin{cases} t_2 - t_1 & \text{if } f(x) \ge t_2, \\ f(x) - t_1 & \text{if } t_1 < f(x) < t_2, \\ 0 & \text{if } f(x) \le t_1. \end{cases}$$

Figure 4.2: $f_{t_1}^{t_2}(x)$

Obviously, $f_{t_1}^{t_2} \in Lip(\Omega)$. Thus (4.2.1) implies

$$\int_{-\infty}^{\infty} I(\mu_{f_{t_1}^{t_2}}(s)) ds \leq \int_{\Omega} \left| \nabla f_{t_1}^{t_2}(x) \right| d\mu.$$

By condition 4.2.1,

$$|\nabla f_{t_1}^{t_2}| = |\nabla f|_{\chi_{\{t_1 < f < t_2\}}},$$

and, moreover,

$$\int_{-\infty}^{\infty} I(\mu_{f_{t_1}^{t_2}}(s)) ds = \int_{t_1}^{t_2} I(\mu_{f_{t_1}^{t_2}}(s)) ds.$$

Observing that $t_1 < z < t_2$,

$$\mu\{f \ge t_2\} \le \mu_{f_{t_1}^{t_2}}(z) \le \mu\{f > t_1\}.$$

Consequently, by the properties of I, we have

$$\int_{t_1}^{t_2} I(\mu_{f_{t_1}^{t_2}}(z)) dz \ge (t_2 - t_1) \min\{I(\mu\{f \ge t_2\}), I(\mu\{f > t_1\})\}.$$
(4.2.5)

We now show that f_{μ}^{\bigstar} is locally absolutely continuous. Indeed, for s > 0 and h > 0, pick $t_1 = f_{\mu}^{\bigstar}(s+h)$, $t_2 = f_{\mu}^{\bigstar}(s)$. Then

$$s \le \mu\{f(x) \ge f_{\mu}^{\bigstar}(s)\} \le \mu_{f_{t_1}^{t_2}}(s) \le \mu\{f(x) > f_{\mu}^{\bigstar}(s+h)\} \le s+h.$$
(4.2.6)

Combining (4.2.5) and (4.2.6) yields

$$(f_{\mu}^{\bigstar}(s) - f_{\mu}^{\bigstar}(s+h)) \min\{I(s+h), I(s)\} \le \int_{\{f_{\mu}^{\bigstar}(s) < f < f_{\mu}^{\bigstar}(s+h)\}} |\nabla f(x)| \, d\mu \quad (4.2.7)$$

which implies that f_{μ}^{\bigstar} is locally absolutely continuous on [a, b] (0 < a < b < 1). Indeed, for any finite family of non-overlapping intervals $\{(a_k, b_k)\}_{k=1}^r$, with $(a_k, b_k) \subset [a, b]$ and $\sum_{k=1}^r (b_k - a_k) \leq \delta$, we have

$$\mu\{\bigcup_{k=1}^{r}\{f_{\mu}^{\bigstar}(b_{k}) < f < f_{\mu}^{\bigstar}(a_{k})\}\} = \sum_{k=1}^{r}\mu\{f_{\mu}^{\bigstar}(b_{k}) < f < f_{\mu}^{\bigstar}(a_{k})\} \le \sum_{k=1}^{r}(b_{k} - a_{k}) \le \delta.$$

Therefore, combining this fact with (4.2.7), we have

$$\begin{split} \sum_{k=1}^{r} (f_{\mu}^{\bigstar}(a_{k}) - f_{\mu}^{\bigstar}(b_{k})) \min\{I(a), I(b)\} &\leq \sum_{k=1}^{r} (f_{\mu}^{\bigstar}(a_{k}) - f_{\mu}^{\bigstar}(b_{k})) \min\{I(a_{k}), I(b_{k})\} \\ &\leq \sum_{k=1}^{r} \int_{\{f_{\mu}^{\bigstar}(b_{k}) < f < f_{\mu}^{\bigstar}(a_{k})\}} |\nabla f(x)| \, d\mu \\ &= \int_{\bigcup_{k=1}^{r} \{f_{\mu}^{\bigstar}(b_{k}) < f < f_{\mu}^{\bigstar}(a_{k})\}} |\nabla f(x)| \, d\mu \\ &\leq \int_{0}^{\sum_{k=1}^{r} (b_{k} - a_{k})} |\nabla f|_{\mu}^{\ast}(t) \, dt \\ &\leq \int_{0}^{\delta} |\nabla f|_{\mu}^{\ast}(t) \, dt, \end{split}$$

CHAPTER 4. SYMMETRIZATION INEQUALITIES FOR CONVEX PROFILE

and the local absolute continuity follows. Now, (4.2.7) implies

$$\frac{\left(f_{\mu}^{\bigstar}(s) - f_{\mu}^{\bigstar}(s+h)\right)}{h}\min(I(s+h), I(s)) \leq \int_{\{f_{\mu}^{\bigstar}(s+h) < f < f_{\mu}^{\bigstar}(s)\}} |\nabla f(x)| d\mu$$
$$\leq \frac{1}{h} \int_{\{f_{\mu}^{\bigstar}(s+h) < f \leq f_{\mu}^{\bigstar}(s)\}} |\nabla f(x)| d\mu.$$

Letting $h \to 0$,

$$(-f_{\mu}^{\bigstar})'(s)I(s) \leq \frac{\partial}{\partial s} \int_{\{f > f_{\mu}^{\bigstar}(s)\}} |\nabla f(x)| \, d\mu$$

Let us consider a finite family of intervals (a_i, b_i) , i = 1, ..., m, with $0 < a_1 < b_1 \le a_2 < b_2 \le \cdots \le a_m < b_m < 1$. Then

$$\begin{split} \int_{\cup_{1\leq i\leq m}(a_i,b_i)} \left(-f_{\mu}^{\bigstar}\right)'(s)I(s)ds &\leq \int_{\cup_{1\leq i\leq m}(a_i,b_i)} \left(\frac{\partial}{\partial s}\int_{\left\{|f|>f_{\mu}^{\bigstar}(s)\right\}} |\nabla f(x)|\,d\mu(x)\right)ds \\ &= \sum_{i=1}^{m} \int_{\left\{f_{\mu}^{\bigstar}(b_i)<|f|\leq f_{\mu}^{\bigstar}(a_i)\right\}} |\nabla f(x)|\,d\mu(x) \\ &= \sum_{i=1}^{m} \int_{\left\{f_{\mu}^{\bigstar}(b_i)<|f|< f_{\mu}^{\bigstar}(a_i)\right\}} |\nabla f(x)|\,d\mu(x) \text{ (by condition (4.2.1))} \\ &= \int_{\cup_{1\leq i\leq m}\left\{f_{\mu}^{\bigstar}(b_i)<|f|< f_{\mu}^{\bigstar}(a_i)\right\}} |\nabla f(x)|\,d\mu(x) \\ &\leq \int_{0}^{\sum_{i=1}^{m}(b_i-a_i)} |\nabla f|_{\mu}^{\ast}(s)ds. \end{split}$$

Now by a routine limiting process we can show that for any measurable set $E \subset (0, 1)$ with Lebesgue measure equal to t we have

$$\int_{E} (-f_{\mu}^{\bigstar})'(s)I(s)ds \leq \int_{0}^{|E|} |\nabla f|_{\mu}^{\ast}(s)ds.$$

Therefore

$$\int_{0}^{t} ((-f_{\mu}^{\star})'(\cdot)I(\cdot))^{*}(s)ds \leq \int_{0}^{t} (|\nabla f|_{\mu}^{*}(\cdot))^{*}(s)ds, \qquad (4.2.8)$$

where the second rearrangement is with respect to Lebesgue measure. Now, since $|\nabla f|^*_{\mu}(s)$ is decreasing, we have

$$\left(\left|\nabla f\right|_{\mu}^{*}(\cdot)\right)^{*}(s) = \left|\nabla f\right|_{\mu}^{*}(s),$$

and thus (4.2.8) yields

$$\int_0^t \left(\left(-f_{\mu}^{\bigstar} \right)'(\cdot) I(\cdot) \right)^*(s) ds \le \int_0^t \left| \nabla f \right|_{\mu}^*(s) ds.$$

 $(iii.) \rightarrow (iv.)$ Assume first that $f \in Lip(\Omega)$ is positive, and bounded, with m(f) = 0. By (iii) we have that $f_{\mu}^{\bigstar} = f_{\mu}^{\ast}$ (since $f \ge 0$) is locally absolutely continuous and $f_{\mu}^{\ast}(1/2) = 0$ (since m(f) = 0). Let $0 < s < z \le 1/2$. Then

$$\begin{split} \int_{\Omega} |f(x)| \, d\mu &= \int_{0}^{1/2} f_{\mu}^{*}(z) dz = \int_{0}^{1/2} \int_{z}^{1/2} (-f_{\mu}^{*})'(x) dx dz = \\ &= \int_{0}^{1/2} z(-f_{\mu}^{*})'(z) dz - s \int_{0}^{1/2} (-f_{\mu}^{*})'(z) dz + s \int_{0}^{1/2} (-f_{\mu}^{*})'(z) dz \\ &= \int_{0}^{1/2} \frac{z - s}{I(z)} (-f_{\mu}^{*})'(z) I(z) dz + s \int_{0}^{1/2} (-f_{\mu}^{*})'(z) dz \\ &\leq \sup_{s < z \le 1/2} \frac{z - s}{I(z)} \int_{0}^{1/2} (-f_{\mu}^{*})'(z) I(z) dz + s \int_{0}^{1/2} (-f_{\mu}^{*})'(z) dz \\ &\leq \beta_{1}(s) \int_{0}^{1/2} (-f_{\mu}^{*})'(z) I(z) dz + s \int_{0}^{1/2} (-f_{\mu}^{*})'(z) dz. \end{split}$$

Since

$$s \int_0^{1/2} (-f_{\mu}^*)'(z) = s(f_{\mu}^*(0^+) - f_{\mu}^*(1/2)) \le s Osc_{\mu}(f),$$

we get

$$\begin{split} \int_{\Omega} |f(x)| \, d\mu &\leq \beta_1(s) \int_0^t (-f_{\mu}^*)'(z) I(z) \, dz + s \, Osc_{\mu}(f) \\ &\leq \beta_1(s) \int_0^{1/2} \left((-f_{\mu}^*)'(\cdot) I(\cdot) \right)^*(t) \, dt + s \, Osc_{\mu}(f) \\ &\leq \beta_1(s) \int_0^{1/2} |\nabla f|_{\mu}^*(t) \, dt + s \, Osc_{\mu}(f) \quad (by \ (4.2.3)) \\ &= \beta_1(s) \int_{\Omega} |\nabla f(x)| \, d\mu + s \, Osc_{\mu}(f) \end{split}$$

In the general case, we follow [7, Lemma 8.3]. Apply the previous argument to $f^+ = \max(f, 0)$ and $f^- = \max(-f, 0)$, which are positive, Lipschitz and have median zero, and we obtain

$$\begin{split} & \int_{\{f>0\}} |f(x)| \, d\mu \leq \beta_1(s) \, \int_{\{f>0\}} |\nabla f(x)| \, d\mu + sOsc_\mu(f^+), \\ & \int_{\{f<0\}} |f(x)| \, d\mu \leq \beta_1(s) \, \int_{\{f><0\}} |\nabla f(x)| \, d\mu + sOsc_\mu(f^-). \end{split}$$

Adding the two inequalities and since $Osc_{\mu}(f^{-}) + Osc_{\mu}(f^{+}) \leq Osc_{\mu}(f)$, we get (4.2.4).

 $(iv) \rightarrow (i.)$ This part was proved in [7, Lemma 8.3], we include its proof for the sake of completeness. Given a Borel set $A \subset \Omega$ we may approximate the indicator function χ_A by functions with finite Lipschitz seminorm (see [8]) to derive $\mu(A) \leq \beta_1(s)\mu^+(A) + s$. Therefore, if $\mu(A) = t$,

$$t - s \le \beta_1(s)\mu^+(A)$$

thus the optimal choice should be

$$I(t) = \sup_{0 < s < t} \frac{t - s}{\beta_1(s)}.$$

4.3 Sobolev–Poincaré and Nash type inequalities

The isoperimetric inequality implies weaker Sobolev–Poincaré and Nash type inequalities. In what follows, we will analyse both.

4.3.1 Sobolev–Poincaré inequalities

The isoperimetric Hardy operator Q_I is the operator defined on Lebesgue measurable functions on (0, 1) by

$$Q_I f(t) = \int_t^{1/2} f(s) \frac{ds}{I(s)}, \qquad 0 < t < 1/2,$$

where I is a convex isoperimetric estimator. In this section we consider the possibility of characterizing Sobolev embeddings in terms of the boundedness of Q_I .

Lemma 4.3.1. Let Y, Z be two q.r.i. spaces on (0,1). Assume that there is a constant $C_0 > 0$ such that

$$\|Q_I f\|_Y \le C_0 \|f\|_Z \,. \tag{4.3.1}$$

Then there exists a constant $C_1 > 0$ such that

 $\|\bar{Q}_I f\|_Y \leq C_1 \|f\|_Z$,

where \bar{Q}_I is the operator defined on Lebesgue measurable functions on (0,1) by

$$\bar{Q}_I f(t) = \int_t^{1/2} f(s) \frac{ds}{I(s)}, \qquad 0 < t < 1$$

Proof. Since

$$\begin{split} \bar{Q}_I f(t) &= \chi_{(0,1/2)}(t) \bar{Q}_I f(t) + \chi_{(1/2,1)}(t) \int_t^{1/2} f(s) \frac{ds}{I(s)} \\ &= \chi_{(0,1/2)}(t) Q_I f(t) + \chi_{(1/2,1)}(t) \int_t^{1/2} f(s) \frac{ds}{I(s)} \end{split}$$

it is enough to prove the boundedness of $\chi_{(1/2,1)}(t) \int_t^{1/2} f(s) \frac{ds}{I(s)}$.

For $t \in (1/2, 1)$, we have that

$$\int_{t}^{1/2} f(s) \frac{ds}{I(s)} = -\int_{1/2}^{t} f(s) \frac{ds}{I(s)} = \int_{1/2}^{1-t} f(1-s) \frac{ds}{I(1-s)}$$
$$= -\int_{1-t}^{1/2} f(1-s) \frac{ds}{I(s)} \quad (\text{since } I(s) = I(1-s)).$$

Thus

$$\begin{aligned} \left\| \chi_{(1/2,1)}(t) \int_{t}^{1/2} f(s) \frac{ds}{I(s)} \right\|_{Y} &= \left\| \chi_{(1/2,1)}(t) \int_{1-t}^{1/2} f(1-s) \frac{ds}{I(s)} \right\|_{Y} \\ &= \left\| \chi_{(1/2,1)}(1-t) \int_{t}^{1/2} f(1-s) \frac{ds}{I(s)} \right\|_{Y} \quad (\text{since } \|\cdot\|_{Y} \text{ is r.i}) \\ &= \left\| \chi_{(0,1/2)}(t) \int_{t}^{1/2} f(1-s) \frac{ds}{I(s)} \right\|_{Y} \\ &\leq C \left\| \chi_{(0,1/2)}(t) f(1-t) \right\|_{Z} \\ &\leq C \left\| f(t) (\chi_{(0,1/2)}(1-t)) \right\|_{Z} \quad (\text{since } \|\cdot\|_{\bar{X}} \text{ is r.i}) \\ &\leq C \left\| \chi_{(1/2,1)}(t) f(t) \right\|_{Z} \\ &\leq C \left\| f\|_{Z} . \end{aligned}$$

Theorem 16. Let Y be a q.r.i. space on (0,1), and let X be an r.i. space on Ω . Assume that there is a constant C > 0 such that

$$\|Q_I f\|_Y \le C \, \|f\|_{\bar{X}} \,. \tag{4.3.2}$$

Then, for all $g \in Lip(\Omega)$, we have that

$$\inf_{c \in \mathbb{R}} \left\| (g - c)^*_{\mu} \right\|_Y \leq \| |\nabla g| \|_X.$$

Proof. Given $g \in Lip(\Omega)$, by part 2 of Theorem 15, g_{μ}^{\bigstar} is locally absolutely continuous on (0,1). Thus, for $t \in (0,1)$, we have that

$$\begin{aligned} \left| g_{\mu}^{\bigstar}(t) - g_{\mu}^{\bigstar}(1/2) \right| &= \left| \int_{t}^{1/2} \left(-g_{\mu}^{\bigstar} \right)'(s) ds \right| = \left| \int_{t}^{1/2} \left(-g_{\mu}^{\bigstar} \right)'(s) I(s) \frac{ds}{I(s)} \right| \\ &= \left| \bar{Q}_{I} \left(\left(-g_{\mu}^{\bigstar} \right)'(\cdot) I(\cdot) \right)(t) \right|. \end{aligned}$$

Then

$$\begin{split} \left\| \left| g_{\mu}^{\bigstar}(t) - g_{\mu}^{\bigstar}(1/2) \right| \right\|_{Y} &= \left\| \bar{Q}_{I} \left(\left(-g_{\mu}^{\bigstar} \right)'(\cdot) I(\cdot) \right)(t) \right\|_{Y} \\ &\leq \left\| \left(-g_{\mu}^{\bigstar} \right)'(\cdot) I(\cdot) \right\|_{\bar{X}} \quad (\text{by } (4.3.2) \text{ and Lemma } 4.3.1) \\ &\leq \left\| |\nabla g| \right\|_{X} \quad (\text{by } (4.2.3)). \end{split}$$

Therefore

$$\begin{split} \inf_{c \in \mathbb{R}} \left\| (g-c)_{\mu}^{*}(t) \right\|_{Y} &= \inf_{c \in \mathbb{R}} \left\| (g-c)_{\mu}^{\star}(t) \right\|_{Y} \\ &\leq \left\| \left| g_{\mu}^{\star}(t) - g_{\mu}^{\star}(1/2) \right| \right\|_{Y} \\ &\leq \left\| \left| \nabla g \right| \right\|_{X}. \end{split}$$

г		п
L		1
L		1
L	 _	_

Theorem 17. Let X be an r.i. space on Ω . Assume that either $\underline{\alpha}_X > 0$ or that there is a c > 0 such that the convex isoperimetric estimator I satisfies

$$\int_{t}^{1/2} \frac{ds}{I(s)} \le c \frac{t}{I(t)}, \quad 0 < t < 1/2.$$
(4.3.3)

Then, for all $g \in Lip(\Omega)$, we have that

$$\inf_{c \in \mathbb{R}} \left\| (g - c)_{\mu}^{*}(t) \frac{I(t)}{t} \right\|_{\bar{X}} \leq \| |\nabla g| \|_{X}.$$
(4.3.4)

Moreover, if Y is a q.r.i. space on (0,1) such that

$$\|Q_I f\|_Y \le \|f\|_{\bar{X}}, \tag{4.3.5}$$

then for any μ -measurable function g on Ω , we have that

$$\left\|g_{\mu}^{*}\right\|_{Y} \leq \left\|g_{\mu}^{*}(t)\frac{I(t)}{t}\right\|_{\bar{X}}.$$

In particular, for all $g \in Lip(\Omega)$, we get

$$\inf_{c\in\mathbb{R}} \left\| (g-c)_{\mu}^{*} \right\|_{Y} \leq \inf_{c\in\mathbb{R}} \left\| (g-c)_{\mu}^{*}(t) \frac{I(t)}{t} \right\|_{\bar{X}} \leq \left\| |\nabla g| \right\|_{X}.$$

Proof. We associate to the r.i. space \overline{X} the weighted q.r.i. space Z on (0,1) whose quasi-norm is defined by

$$||f||_Z := \left\| f^*(t) \frac{I(t)}{t} \right\|_{\bar{X}}.$$

We claim that there is a C > 0 such that

$$||Q_I f||_Z \leq C ||f||_{\bar{X}},$$

and therefore (4.3.4) follows by Theorem 16.

Case 1: $\underline{\alpha}_X > 0$:

$$\begin{split} \|Q_I f\|_Z &= \left\| \frac{I(t)}{t} \left(\int_t^{1/2} f(s) \frac{ds}{I(s)} \right)^* \right\|_{\bar{X}} \le \left\| \frac{I(t)}{t} \left(\int_t^{1/2} |f(s)| \frac{ds}{I(s)} \right)^* \right\|_{\bar{X}} \\ &= \left\| \frac{I(t)}{t} \int_t^{1/2} |f(s)| \frac{ds}{I(s)} \right\|_{\bar{X}} \text{ (since } Q_I |f|(t) \text{ is decreasing)} \\ &= \left\| \frac{I(t)}{t} \int_t^{1/2} |f(s)| \frac{s}{I(s)} \frac{ds}{s} \right\|_{\bar{X}} \\ &\le \left\| \int_t^{1/2} |f(s)| \frac{ds}{s} \right\|_{\bar{X}} \text{ (since } \frac{s}{I(s)} \text{ decreases)} \\ &\le \| f\|_{\bar{X}} \text{ (since } \underline{\alpha}_X > 0 \text{).} \end{split}$$

Case 2 : The convex isoperimetric estimator satisfies (4.3.3). Consider \tilde{Q}_I defined by

$$\tilde{Q}_I f(t) = \frac{I(t)}{t} Q_I f(t).$$

We claim that $\tilde{Q}_I : L^1(0,1) \to L^1(0,1)$ is bounded, and $\tilde{Q}_I : L^{\infty}(0,1) \to L^{\infty}(0,1)$ is bounded, and so by interpolation (see [53]) \tilde{Q}_I will be bounded on \bar{X} . Thus

$$\begin{aligned} \|Q_I f\|_Z &\leq \|Q_I |f|\|_Z = \left\| \frac{I(t)}{t} \left(Q_I |f|\right)^* (t) \right\|_{\bar{X}} \\ &= \left\| \frac{I(t)}{t} Q_I |f| (t) \right\|_{\bar{X}} \quad \text{(since } Q_I |f| (t) \text{ is decreasing)} \\ &= \|\tilde{Q}_I |f| (t)\|_{\bar{X}} \\ &\leq \|f\|_{\bar{X}} \end{aligned}$$

and Theorem 16 applies.

We are now going to prove the claim.

By the convexity of I, $\frac{I(t)}{t}$ is increasing for 0 < t < 1/2, thus

$$\int_0^s \frac{I(t)}{t} dt \le I(s),$$

and therefore

$$\begin{split} \|\tilde{Q}_{I}f\|_{1} &\leq \int_{0}^{1} \tilde{Q}_{I}\left(|f|\right)(t)dt \\ &= \int_{0}^{1/2} \frac{I(t)}{t} \left(\int_{t}^{1/2} |f(s)| \frac{ds}{I(s)}\right) dt \\ &= \int_{0}^{1/2} \frac{|f(s)|}{I(s)} \left(\int_{0}^{s} \frac{I(t)}{t} dt\right) ds \\ &\leq \int_{0}^{1/2} |f(s)| ds \\ &= \|f\|_{1} \,. \end{split}$$

Similarly,

$$\begin{split} \|\tilde{Q}_{I}f\|_{\infty} &\leq \sup_{0 < t < 1} \tilde{Q}_{I}\left(|f|\right)(t) \\ &\leq \sup_{0 < t < 1/2} \frac{I(t)}{t} \int_{t}^{1/2} |f(s)| \frac{ds}{I(s)} \\ &\leq \|f\|_{\infty} \sup_{0 < t < 1/2} \left(\frac{I(t)}{t} \int_{t}^{1/2} \frac{ds}{I(s)}\right) \\ &\leq c \|f\|_{\infty} \quad \text{(by (4.3.3)).} \end{split}$$

To finish the proof of the theorem it remains to show that

$$\|f_{\mu}^{*}\|_{\bar{Y}} \leq \left\|f_{\mu}^{*}(t)\frac{I(t)}{t}\right\|_{\bar{X}}.$$
(4.3.6)

Let $C_{\bar{Y}}$ be the constant quasi-norm of \bar{Y} , then

Since f^*_{μ} is decreasing,

$$f_{\mu}^{*}(t)\chi_{(0,1/4)}(t) \leq \frac{1}{\ln 2} \int_{t/2}^{t} f_{\mu}^{*}(s) \frac{ds}{s} = \frac{1}{\ln 2} \int_{t/2}^{1/2} f_{\mu}^{*}(s)\chi_{(0,1/4)}(s) \frac{I(s)}{s} \frac{ds}{I(s)}.$$

Thus

$$\begin{split} \|f_{\mu}^{*}(t)\chi_{(0,1/4)}(t)\|_{\bar{Y}} &\leq \left\|Q_{I}\left(f_{\mu}^{*}(\cdot)\chi_{(0,1/4)}(\cdot)\frac{I(\cdot)}{\cdot}\right)(t/2)\right\|_{\bar{Y}} & (4.3.8)\\ &\leq \left\|f_{\mu}^{*}(t/2)\chi_{(0,1/4)}(t/2)\frac{I(t/2)}{t/2}\right\|_{\bar{X}} & (by \ (4.3.5))\\ &\leq \left\|f_{\mu}^{*}(t)\chi_{(0,1/2)}(t)\frac{I(t)}{t}\right\|_{\bar{X}}\\ &\leq \left\|f_{\mu}^{*}(t)\frac{I(t)}{t}\right\|_{\bar{X}}. \end{split}$$

Combining (4.3.7) and (4.3.8) we obtain (4.3.6).

Remark 4.3.1. If $g \in Lip(\Omega)$ is positive with m(g) = 0, then it follows from the previous theorem that

$$\left\|g_{\mu}^{*}(t)\frac{I(t)}{t}\right\|_{\bar{X}} \leq \left\||\nabla g|\right\|_{X}.$$

4.3.2 Nash inequalities

In this section we obtain Nash type inequalities. We will focus on the following type of probability measures.

Definition 4.3.1. Let μ be a probability measure on Ω which admits a convex isoperimetric estimator I.

1. Let $\alpha > 0$. We will say that μ is of α -Cauchy type if

$$I(t) = c_{\mu} \min(t, 1-t)^{1+1/\alpha}$$

2. Let $0 . We will say that <math>\mu$ is of extended p-exponential type if

$$I(t) = c_{\mu} \min(t, 1-t) \left(\log \frac{1}{\min(t, 1-t)} \right)^{1-1/p}.$$

In both cases c_{μ} denotes a positive constant.

Theorem 18. The following Nash inequalities hold:

1. Let μ be of α -Cauchy type. Let X be an r.i. space on Ω with $\underline{\alpha}_X > 0$. Let $1 < q \leq \infty$ satisfy $0 \leq 1/q < \underline{\alpha}_X$. Then for all positive $f \in Lip(\Omega)$ with m(f) = 0, we have

$$||f||_X \le \min_{r>1} \left(r \, |||\nabla f||_X + ||f||_{q,\infty} \, \phi_X(r^{-\alpha}) r^{\alpha/q} \right).$$

CHAPTER 4. SYMMETRIZATION INEQUALITIES FOR CONVEX PROFILE

2. Let μ be of extended p-exponential type. Let X be an r.i. space on Ω . Let $\beta > 0$. Then for all positive $f \in Lip(\Omega)$ with m(f) = 0, we have

$$\|f\|_{X} \leq \||\nabla f|\|_{X}^{\frac{\beta}{\beta+1}} \|f\|_{X^{(\ln(\frac{1}{t})^{\beta(\frac{1}{p}-1)})}.$$

Proof. Part 1. Let $f \in Lip(\Omega)$ be positive with m(f) = 0 and let $\omega(t) = t^{-1/\alpha}$ (0 < t < 1/2). Let r > 1 and let $\beta > 0$ be chosen later. Then

$$\begin{split} \|f\|_{X} &= \|f_{\mu}^{*}\|_{\bar{X}} \leq \left\|f_{\mu}^{*}(t)\frac{\omega(t)}{\omega(t)}\chi_{\{\omega< r\}}(t)\right\|_{\bar{X}} + \left\|f_{\mu}^{*}(t)\left(\frac{\omega(t)}{\omega(t)}\right)^{\beta}\chi_{\{\omega> r\}}(t)\right\|_{\bar{X}} (4.3.9) \\ &\leq r \left\|f_{\mu}^{*}(t)t^{1/\alpha}\right\|_{\bar{X}} + r^{-\beta} \left\|f_{\mu}^{*}(t)t^{-\beta/\alpha}\chi_{(0,r^{-\alpha})}(t)\right\|_{\bar{X}} \\ &= r \left\|f_{\mu}^{*}(t)t^{1/\alpha}\right\|_{\bar{X}} + r^{-\beta} \left\|t^{1/q}f_{\mu}^{*}(t)t^{-\beta/\alpha-1/q}\chi_{(0,r^{-\alpha})}(t)\right\|_{\bar{X}} \\ &\leq r \left\|f_{\mu}^{*}(t)t^{1/\alpha}\right\|_{\bar{X}} + r^{-\beta} \sup_{t>0} (t^{1/q}f_{\mu}^{*}(t)) \left\|t^{-\beta/\alpha-1/q}\chi_{(0,r^{-\alpha})}(t)\right\|_{\bar{X}} \\ &\leq r \left\|f_{\mu}^{*}(t)t^{1/\alpha}\right\|_{\bar{X}} + r^{-\beta} \left\|f\right\|_{q,\infty} \left\|t^{-\beta/\alpha-1/q}\chi_{(0,r^{-\alpha})}(t)\right\|_{\Lambda(\bar{X})} (by (2.2.8)) \\ &\leq r \left\|f_{\mu}^{*}(t)t^{1/\alpha}\right\|_{\bar{X}} + r^{-\beta} \left\|f\right\|_{q,\infty} \int_{0}^{r^{-\alpha}} t^{-\beta/\alpha-1/q}\frac{\phi_{X}(t)}{t} \\ &= r \left\|f_{\mu}^{*}(t)t^{1/\alpha}\right\|_{\bar{X}} + r^{-\beta} \left\|f\right\|_{q,\infty} J(r). \end{split}$$

Let $0 \leq 1/q < \gamma < \underline{\alpha}_X$. By Lemma 2.2.1,

$$\int_0^{r^{-\alpha}} t^{-\beta/\alpha - 1/q} \frac{\phi_X(t)}{t^{\gamma} t^{1-\gamma}} \leq \frac{\phi_X(r^{-\alpha})}{r^{-\alpha\gamma}} \int_0^{r^{-\alpha}} t^{-\beta/\alpha - 1/q + \gamma - 1}.$$

At this stage we select $0 < \beta < \alpha (\gamma - 1/q)$. Then

$$\int_0^{r^{-\alpha}} t^{-\beta/\alpha - 1/q + \gamma - 1} \le r^{-\alpha(-\beta/\alpha - 1/q + \gamma)},$$

and thus

$$J(r) \leq \phi_X(r^{-\alpha})r^{\beta+\alpha/q}.$$

Inserting this information into (4.3.9) and using Remark 4.3.1, we get

$$\|f\|_{X} \leq r \left\| f_{\mu}^{*}(t) t^{1/\alpha} \right\|_{\bar{X}} + \|f\|_{q,\infty} \phi_{X}(r^{-\alpha}) r^{\alpha/q}$$
$$\leq r \||\nabla f|\|_{X} + \|f\|_{q,\infty} \phi_{X}(r^{-\alpha}) r^{\alpha/q}.$$

Part 2. Let $f \in Lip(\Omega)$ be positive with m(f) = 0 and let $\omega(t) = \left(\ln \frac{1}{t}\right)^{\frac{1}{p}-1}$ (0 < t < 1/2). Let r > 1 and $\beta > 0$.

$$\begin{split} \|f\|_{X} &= \left\|f_{\mu}^{*}\right\|_{\bar{X}} \leq \left\|f_{\mu}^{*}(t)\frac{\omega(t)}{\omega(t)}\chi_{\{\omega < r\}\}}(t)\right\|_{\bar{X}} + \left\|f_{\mu}^{*}(t)\left(\frac{\omega(t)}{\omega(t)}\right)^{\beta}\chi_{\{\omega > r\}}(t)\right\|_{\bar{X}} \\ &\leq r\left\|f_{\mu}^{*}(t)\left(\ln\frac{1}{t}\right)^{1-\frac{1}{p}}\right\|_{\bar{X}} + r^{-\beta}\left\|f_{\mu}^{*}(t)\left(\ln\frac{1}{t}\right)^{\beta\left(\frac{1}{p}-1\right)}\right\|_{\bar{X}} \\ &\leq r\left\||\nabla f|\|_{X(\log\left(\frac{1}{t}\right)^{\beta\left(\frac{1}{p}-1\right)})} + r^{-\beta}\left\|f\right\|_{X} \quad (by \text{ Remark 4.3.1}). \end{split}$$

We finish by taking the inf for r > 1.

Remark 4.3.2. Let X be an r.i. space on Ω with $\underline{\alpha}_X > 0$. Let $1 < q \le \infty$ be such that $0 \le 1/q < \underline{\alpha}_X$. Then

$$L^{q,\infty}(\Omega) \subset \Lambda(X) \subset X(\Omega).$$

In fact, by Lemma 2.2.1,

$$\|f\|_{\Lambda(X)} = \int_0^1 f^*(t) \frac{\phi_X(t)}{t} dt \le \|f\|_{q,\infty} \int_0^1 \frac{\phi_X(t)}{t^{1+1/q}} dt.$$

The last integral is finite since taking $0 \le 1/q < \gamma < \underline{\alpha}_X$, we get

$$\int_0^1 \frac{\phi_X(t)}{t^{1+1/q}} dt = \int_0^1 t^{1/q+\gamma-1} \frac{\phi_X(t)}{t^{\gamma}} dt \le \int_0^1 t^{1/q+\gamma-1} < \infty.$$

4.4 Examples and applications

In this section we will apply the previous results to the probability measures introduced in Examples 4.1.2, 4.1.3 and 4.1.4.

4.4.1 Cauchy type laws

Consider the probability measure space (\mathbb{R}^n, d, μ) where *d* is the Euclidean distance and μ is the probability measure introduced in Example 4.1.2. Such measures have been introduced by Borell [9] (see also [7]). Prototypes of these probability measures are the generalized Cauchy distributions⁵:

$$d\mu(x) = \frac{1}{Z} \left(\left(1 + |x|^2 \right)^{1/2} \right)^{-(n+\alpha)}, \quad \alpha > 0.$$

⁵These measures are Barenblatt solutions of the porous medium equations and appear naturally in weighted porous medium equations, giving the decay rate of this non-linear semigroup towards the equilibrium measure, see [98] and [19].

CHAPTER 4. SYMMETRIZATION INEQUALITIES FOR CONVEX PROFILE

A convex isoperimetric estimator for these measures is (see [15, Proposition 4.3])

$$I(t) = \min(t, 1-t)^{1+1/\alpha}.$$

Obviously for 0 < t < 1/2, we have

$$\int_{t}^{1/2} \frac{ds}{s^{1+1/\alpha}} \le \frac{t}{t^{1+1/\alpha}}.$$

Thus by Theorem 17, given an r.i. space X on \mathbb{R}^n we get

$$\inf_{c \in \mathbb{R}} \left\| (g-c)_{\mu}^{*} \frac{\min(t,1-t)^{1+1/\alpha}}{t} \right\|_{\bar{X}} \leq \left\| |\nabla g| \right\|_{X}, \quad (g \in Lip(\mathbb{R}^{n})).$$

Proposition 4.4.1. Let $1 \le p < \infty$, $1 \le q \le \infty$. For all $f \in Lip(\mathbb{R}^n)$ positive with m(f) = 0, we get

1.

$$\|f\|_{\frac{p\alpha}{p+\alpha},q} \le \||\nabla f|\|_{p,q}.$$

2. For all s > p

$$\|f\|_{p,q} \leq \||\nabla f|\|_{p,q}^{\frac{\beta}{\beta+1}} \|f\|_{s,\infty}^{\frac{1}{\beta+1}}$$

where $\beta = \alpha(\frac{1}{p} - \frac{1}{s}).$

Proof. 1) By Theorem 17 we get

$$\left\|f_{\mu}^{*}t^{\frac{1}{\alpha}}\right\|_{p,q} \leq \||\nabla f|\|_{p,q}$$

Now by [53, Page 76] we have that

$$\left\|f_{\mu}^{*}t^{\frac{1}{\alpha}}\right\|_{p,q}^{q} = \int_{0}^{1} \left[\left(t^{\frac{1}{\alpha}}f_{\mu}^{*}(t)\right)^{*}t^{\frac{1}{p}}\right]^{q}\frac{dt}{t} \simeq \int_{0}^{1}\left(t^{\frac{1}{\alpha}+\frac{1}{p}}f_{\mu}^{*}(t)\right)^{q}\frac{dt}{t} = \|f\|_{\frac{p\alpha}{p+\alpha},q}^{q}.$$
(b) is a direct application of Theorem 18.

2) is a direct application of Theorem 18.

Remark 4.4.1. If in the previous proposition we take p = q = 1, we obtain

$$\|f\|_{\frac{\alpha}{\alpha+1},1} \le \||\nabla f|\|_1.$$
 (4.4.1)

If $\frac{1}{q} = \frac{1}{p} + \frac{1}{\alpha}$, then we get

$$\left\|f\right\|_{p,\frac{p(1+\alpha)}{\alpha}} \le \left\|\left|\nabla f\right|\right\|_{q,\frac{p(1+\alpha)}{\alpha}}.$$
(4.4.2)

For $p \ge 1$ and $s = \infty$, we have that

$$\|f\|_{p} \leq \||\nabla f|\|_{p}^{\frac{\beta}{\beta+1}} \|f\|_{\infty}^{\frac{1}{\beta+1}}.$$
(4.4.3)

Inequalities 4.4.1 and 4.4.2 were proved in [72, Proposition 5.13]. Inequality 4.4.3 was obtained in [72, Proposition 5.15].

We close this section with the following optimality result:

Theorem 19. Let $\alpha > 0$. Let \overline{X} be an r.i. space on (0,1) and let Z be a q.r.i. space on (0,1). Assume that for any probability measure μ of α -Cauchy type in \mathbb{R}^n , there is a $C_{\mu} > 0$ such that for all positive $f \in Lip(\mathbb{R}^n)$ with m(f) = 0, we get

$$\|f_{\mu}^{*}\|_{Z} \leq C_{\mu} \||\nabla f|_{\mu}^{*}\|_{\bar{X}}.$$

Then for all $g \in Lip(\mathbb{R}^n)$

$$\left\|g_{\mu}^{*}\right\|_{Z} \leq \left\|g_{\mu}^{*}(t)\frac{I(t)}{t}\right\|_{\bar{X}}$$

Proof. Let μ be the Cauchy probability measure on \mathbb{R} defined by

$$d\mu(s) = \frac{\alpha}{2\left(1+|s|^2\right)^{\frac{1+\alpha}{2}}} ds = \varphi(s)dx, \quad s \in \mathbb{R}.$$

It is known (see [15, Proposition 5.27]) that its isoperimetric profile is given by

$$I_{\mu}(t) = \varphi(H^{-1}(t)) = \alpha 2^{1/\alpha} \min(t, 1-t)^{1+1/\alpha}, \quad t \in [0, 1],$$

where H is the distribution function of μ , i.e. $H : \mathbb{R} \to (0, 1)$ is the increasing function given by

$$H(r)=\int_{-\infty}^r \varphi(t)dt.$$

Consider the product measure μ^n on \mathbb{R}^n . By Proposition 5.27 of [15] the function

$$I(t) = \frac{c_{\alpha}}{n^{1/\alpha}} \min(t, 1-t)^{1+1/\alpha}$$

is a convex isoperimetric estimator of μ^n (c_α denotes a positive constant depending only on α).

Given a positive measurable function f with $supp f \in (0, 1/2)$, consider

$$F(t) = \int_{t}^{1} f(s) \frac{ds}{I_{\mu_{\alpha}}(s)}, \quad t \in (0,1),$$

and define

$$u(x) = F(H(x_1)), \qquad x \in \mathbb{R}^n.$$

Then,

$$\left|\nabla u(x)\right| = \left|\frac{\partial}{\partial x_1}u(x)\right| = \left|-f(H(x_1))\frac{H'(x_1)}{I_{\mu}(H(x_1))}\right| = f(H(x_1)).$$

Let A be a Young's function and let $s = H(x_1)$. Then,

$$\int_{\mathbb{R}^n} A(f(H(x_1))) d\mu^n(x) = \int_{\mathbb{R}} A(f(H(x_1))) d\mu(x_1)$$
$$= \int_0^1 A(f(s)) ds.$$

Therefore, by [6, exercise 5 p. 88]

$$|\nabla u|_{\mu^n}^*(t) = f^*(t).$$

Similarly

$$u_{\mu^n}^*(t) = \int_t^1 f(s) \frac{ds}{I_{\mu}(s)}.$$

Since m(u) = 0, by hypothesis we get

$$\left\| \int_{t}^{1} f(s) \frac{ds}{I_{\mu}(s)} \right\|_{Z} = \left\| u_{\mu^{n}}^{*} \right\|_{Z}$$

$$\leq C_{\mu^{n}} \left\| |\nabla f|_{\mu^{n}}^{*} \right\|_{\bar{X}}$$

$$= C_{\mu^{n}} \left\| f^{*}(t) \right\|_{\bar{X}}$$

$$= C_{\mu^{n}} \left\| f \right\|_{\bar{X}}.$$

Finally, from

$$I_{\mu}(t) = \frac{\alpha 2^{1/\alpha} n^{1/\alpha}}{c_{\alpha}} I(t)$$

we have that

$$\|Q_I f\|_Z \le \frac{c_{\alpha} C_{\mu^n}}{\alpha 2^{1/\alpha} n^{1/\alpha}} \|f\|_{\bar{X}}$$

and the results follow from Theorem 17.

4.4.2 Extended sub-exponential law

Consider the probability measure on \mathbb{R}^n defined by

$$d\mu_p(x) = \frac{1}{Z_p} e^{-V(x)^p} dx = \varphi(x) dx$$

for some positive convex function $V : \mathbb{R}^n \to (0, \infty)$ and $p \in (0, 1)$.

A typical example is $V(x) = |x|^p$, and 0 , which yields to subexponential type law.

A convex isoperimetric estimator for this type of measure is (see [15, Proposition 4.5]):

$$I(t) = c_p \min(t, 1-t) \left(\log \frac{1}{\min(t, 1-t)} \right)^{1-1/p}$$

By Theorem 17, given an r.i. space X on \mathbb{R}^n with $\underline{\alpha}_X > 0$, we get

$$\inf_{c \in \mathbb{R}} \left\| (g-c)_{\mu}^{*} \frac{c_{p} \min\left(t, 1-t\right) \left(\log \frac{1}{\min\left(t, 1-t\right)} \right)^{1-1/p}}{t} \right\|_{\bar{X}} \leq \left\| |\nabla g| \right\|_{X}, \quad (g \in Lip(\mathbb{R}^{n})).$$

In the particular case that $X = L^{r,q}$ we obtain the following proposition.

Proposition 4.4.2. Let $1 \le r < \infty$, $1 \le q < \infty$. For all positive $f \in Lip(\mathbb{R}^n)$ with m(f) = 0,

1.

$$||f||_{L^{r,q}(\log L)^{1-1/p}} \leq ||\nabla f||_{r,q}.$$

2. For all $\beta > 0$

$$\|f\|_{r,q} \le \||\nabla f|\|_{r,q}^{\frac{\beta}{\beta+1}} \|f\|_{L^{r,q}(\log L)^{\beta(1-1/p)}}^{\frac{1}{\beta+1}}$$

Theorem 20. Let $p \in (0,1)$. Let \overline{X} be an r.i. space on (0,1) and let Z be a q.r.i. space on (0,1). Assume that for any p-extended sub-exponential law μ in \mathbb{R}^n there is a $C_{\mu} > 0$ such that for all positive $f \in Lip(\mathbb{R}^n)$ with m(f) = 0,

$$\left\|f_{\mu}^{*}\right\|_{Z} \leq C_{\mu} \left\|\left|\nabla f\right|_{\mu}^{*}\right\|_{\bar{X}}.$$

Then, for all $g \in Lip(\mathbb{R}^n)$,

$$\left\|g_{\mu}^{*}\right\|_{Z} \leq \left\|g_{\mu}^{*}(t)\frac{I(t)}{t}\right\|_{\bar{X}}.$$

Proof. Let μ be a probability measure on \mathbb{R} with density

$$d\mu_p(s) = \frac{e^{-|s|^p}}{Z_p} ds = \varphi(s) ds, \quad s \in \mathbb{R}.$$

Its isoperimetric profile is (see [15, Proposition 5.25])

$$I_{\mu_p}(t) = \varphi \left(H^{-1}(t) \right) = c_p \min(t, 1-t) \left(\log \frac{1}{\min(t, 1-t)} \right)^{1-1/p}, \quad t \in [0, 1],$$

where H is the distribution function of μ , i.e. $H : \mathbb{R} \to (0,1)$ is defined by

$$H(r) = \int_{-\infty}^{r} \varphi(t) dt.$$

Consider the product measure μ^n on \mathbb{R}^n . By proposition 5.25 of [15], there exists a positive constant c such that the function

$$I(t) = c \min(t, 1 - t) \left(\log \frac{n}{\min(t, 1 - t)} \right)^{1 - 1/p}$$

is a convex isoperimetric estimator of μ_p^n .

Let f be a positive measurable function f with $\operatorname{supp}(f) \subset (0, 1/2)$. Consider

$$F(t) = \int_{t}^{1} f(s) \frac{ds}{I_{\mu_{p}}(s)}, \quad t \in (0, 1),$$

and define

$$u(x) = F(H(x_1)), \qquad x \in \mathbb{R}^n.$$

Using the same method as used in Theorem 19, we obtain

$$|\nabla u|_{\mu_p}^*(t) = f^*(t) \text{ and } u_{\mu_p}^*(t) = \int_t^1 f(s) \frac{ds}{I_{\mu_p}(s)}.$$

Since m(u) = 0, by hypothesis we get

$$\left\| \int_{t}^{1} f(s) \frac{ds}{I_{\mu_{p}}(s)} \right\|_{Z} = \left\| u_{\mu_{p}^{n}}^{*} \right\|_{Z}$$

$$\leq C_{\mu_{p}^{n}} \left\| |\nabla f|_{\mu_{p}^{n}}^{*} \right\|_{\bar{X}}$$

$$= C_{\mu_{p}^{n}} \left\| f^{*}(t) \right\|_{\bar{X}}$$

$$= C_{\mu_{p}^{n}} \left\| f \right\|_{\bar{X}}.$$

Finally, from

 $I_{\mu_p}(t) \simeq I(t)$

we have that

$$\|Q_I f\|_Z \le \|f\|_{\bar{X}}$$

and Theorem 17 applies.

4.4.3 Weighted Riemannian manifolds with negative dimension

Let (M^n, g, μ) be an *n*-dimensional weighted Riemannian manifold $(n \ge 2)$ that satisfies the CD(0, N) curvature condition with N < 0. (See [72, Secction 5.4].)

A convex isoperimetric estimator is given by

$$I(t) = \min(t, 1-t)^{-1/N}.$$

Obviously for 0 < t < 1/2 we have

$$\int_t^{1/2} \frac{ds}{s^{-1/N}} \le \frac{t}{t^{-1/N}}.$$

Thus by Theorem 17, given an r.i. space X on \mathbb{R}^n we get

$$\inf_{c\in\mathbb{R}}\left\| (g-c)^*_{\mu} \frac{\min(t,1-t)^{-1/N}}{t} \right\|_{\bar{X}} \leq \||\nabla g|\|_X, \quad (g\in Lip(\mathbb{R}^n)).$$

In particular, if $1 \le p < \infty$, $1 \le q \le \infty$) and $X = L^{p,q}$, then for all positive $f \in Lip(\mathbb{R}^n)$ with m(f) = 0, $(1 \le p < \infty, 1 \le q \le \infty)$,

$$\|f\|_{\gamma,q} \le \||\nabla f|\|_{p,q}$$

where $\gamma = \frac{Np}{N-p(N+1)}$ for any p, q satisfying $\frac{N}{N-1} \le p \le -N$ and $\frac{1}{q} = \frac{1}{p} - \frac{1}{N} - 1$. Now by Theorem 18, we have that

$$\left\|f\right\|_{p,q} \leq \left\|\left|\nabla f\right|\right\|_{p,q}^{\frac{\beta}{\beta+1}} \left\|f\right\|_{s,\infty}^{\frac{1}{\beta+1}}$$

where s > p and $\beta = \alpha(\frac{1}{p} - \frac{1}{s})$.

Bibliography

- Aissa, S. Tightness in Besov-Orlicz spaces: characterizations and applications. Random Oper. Stoch. Equ. 24 (2016), 157–164.
- [2] Alexopolus, G. Spectral multipliers on Lie groups of polynomial growth, Proc. Amer. Math. Soc. 120 (1994), 973–979.
- [3] Ambrosio, L. Miranda, M. Jr., Pallara, D. Special functions of bounded variation in doubling metric measure spaces. Calculus of variations: topics from the mathematical heritage of E. De Giorgi, 1–45, Quad. Mat., 14, Dept. Math., Seconda Univ. Napoli, Caserta, (2004).
- [4] Barthe, F. Levels of concentration between exponential and Gaussian, Ann. Fac. Sci. Toulouse Math. 6 (2001), 393-404.
- [5] Bastero, J., Milman M. and Ruiz, F. A note in $L(\infty, q)$ spaces and Sobolev embeddings, Indiana Univ. Math. J. **52** (2003), 1215–1230.
- [6] Bennett, C. and Sharpley, R. Interpolation of Operators, Academic Press, Boston (1988).
- [7] Bobkov, S.G. Large deviations and isoperimetry over convex probability measures with heavy tails, Electron. J. Probab. 12 (2007), 1072–1100.
- [8] Bobkov, S. G. and Houdré C., Some connections between isoperimetric and Sobolev-type inequalities, Mem. Amer. Math. Soc. 129 (1997), no. 616.
- [9] Borell, C. Convex set functions in d-space. Period. Math. Hungar. 6 (1975), 111–136.
- [10] Boyd, D.W. Indices of function spaces and their relationship to interpolation, Canad. J. Math. 21 (1969), 1245–1254.
- [11] Brézis, H. and Wainger, S. A note on limiting cases of Sobolev embeddings and convolution inequalities, Comm. Partial Differential Equations. 5 (1980), 773–789.

- [12] Caetano, A. M. and Moura, S. D. Local growth envelopes of spaces of generalized smoothness: The critical case Math. Inequal. Appl., 7 (2004), 573– -606.
- [13] Caetano, A. M. and Moura, S. D. Local growth envelopes of spaces of generalized smoothness: The subcritical case. Math. Nachr. 273 (2004), 43-57.
- [14] Carro, M J., Gogatishvili, A., Martín, J. and Pick, L. Functional properties of rearrangement invariant spaces defined in terms of oscillations J. Funct. Anal. 229 (2005), no. 2, 375–404.
- [15] Cattiaux, P., Gozlan, N., Guillin, A. and Roberto, C. Functional inequalities for heavy tailed distributions and application to isoperimetry. Electron. J. Probab. 15 (2010), 346–385.
- [16] Cerdà, J., Martín, J. and Silvestre, P. Conductor Sobolev-Type Estimates and Isocapacitary Inequalities, Indiana Univ. Math. J. 61 (2012), 1925– 1947.
- [17] Cianchi, A. A sharp embedding theorem for Orlicz-Sobolev spaces Indiana Univ. Math. J. 45 (1996), 39–65.
- [18] DeVore, R. A. and Sharpley, R. C. Besov spaces on domains in R^d. Trans. Amer. Math. Soc. 335 (1993), 843–864.
- [19] Dolbeault, J., Gentil, I., Guillin, A. and Wang, F.Y. l^q functional inequalities and weighted porous media equations, Pot. Anal. 28 (2008), 35–59.
- [20] Evans, L. C. and Gariepy, R. F. Measure Theory and Fine Properties of functions, Studies in Advanced Mathematics. CRC Press, Boca Raton, FL, 1992, 268 pp.
- [21] Farkas, W. and Leopold, H.G. Characterisations of function spaces of generalised smoothness, Ann. Mat. Pura Appl. 185 (2006), 1–62.
- [22] Federer, H., Fleming, W.H. Normal and integral currents, Ann. Math. 72 (1960), 458–520.
- [23] Feissner, G. F. Hypercontractive semigroups and Sobolev's inequality, Trans. Amer. Math. Soc. 210 (1975), 51–62.
- [24] Furioli, G., Melzi, C. and Veneruso, A. Littlewood-Paley decompositions and Besov spaces on Lie groups of polynomial growth, Mathematische Nachrichten, 279 (2006), 1028–1040.
- [25] Garsia, A. M. and Rodemich, E. Monotonicity of certain functionals under rearrangements, Ann. Inst. Fourier (Grenoble) 24 (1974), 67-116.

- [26] García-Cuerva, J. and Rubio de Francia, J. Weighted Norm Inequalities and Related Topics, North-Holland Publishing Co., Amsterdam (1985).
- [27] Gogatishvili, A. Koskela, P. and Shanmugalingam, N. Interpolation properties of Besov spaces defined on metric spaces, Math. Nachr. 283 (2010), 215–231.
- [28] Gogatishvili, A. Koskela, P. and Zhou, Y. Characterizations of Besov and Triebel-Lizorkin Spaces on Metric Measure Spaces, Forum Math. 25 (2013), 787–819.
- [29] Goldman, M. L. Rearrangement invariant envelopes of generalized Besov, Sobolev, and Calderon spaces. The interaction of analysis and geometry, 53–81, Contemp. Math., 424, Amer. Math. Soc., Providence.
- [30] Górka. P. In metric-measure spaces Sobolev embedding is equivalent to a lower bound for the measure, Potential Anal. 47(2017), 1319–2017.
- [31] Grigorýan A., Hu J., Lau KS. Heat Kernels on Metric Spaces with Doubling Measure. In: Bandt C., Zähle M., Mörters P. (eds) Fractal Geometry and Stochastics IV. Progress in Probability, 61 (2009). Birkhäuser Basel
- [32] Grigor'yan, A., Hu, J. and Lau, KS. Heat kernels on metric measure spaces and an application to semilinear elliptic equations, Transactions of the American Mathematical Society, 355 (2003), 2065–2095.
- [33] Grigor'yan, A. Heat kernels and function theory on metric measure spaces, in Heat Kernels and Analysis on Manifolds, Graphs, and Metric Spaces (Paris, 2002), **338** of Contemporary Mathematics, 143–172, American Mathematical Society, Providence, RI, USA, 2003.
- [34] Gross, L. Logarithmic Sobolev inequalities, Amer. J. Math. 97 (1975), 1061-1083.
- [35] Hajłasz, P. Sobolev spaces on an arbitrary metric space, Potential Analysis, 5 (1996), 403–415.
- [36] Hajłasz, P. and Koskela, P. Sobolev met Poincaré, Memoirs of the American Mathematical Society, 145 (2000), 1–101.
- [37] Koskela, P. and MacManus, P. Quasiconformal mappings and Sobolev spaces, Studia Mathematica, 131 (1998), 1–17.
- [38] Han, Y., Müller, D. and Yang, D. A theory of Besov and Triebel-Lizorkin spaces on metric measure spaces modeled on Carnot-Caratheodory spaces, Abstr. Appl. Anal. 2008, Art. ID 893409, 250 pp.

- [39] Han, Y., Sawyer, E.T. Littlewood-Paley theory on spaces of homogeneuos type and classical function spaces, Mem.Amer.Math.Soc. 110 (1994), 1–126.
- [40] Han, Y. Plancherel-Pôlya type inequality on spaces of homogeneuos type and its applications, Proc. Amer. Math. Soc.126 (1998), 3315–3327.
- [41] Han, Y., Yang, D. New characterizations and applications of inhomogeneous Besov and Triebel-Lizorkin spaces on homogeneous type spaces and fractals, Dissertationes Math.(Rozprawy Math.) 403 (2002), 1–102.
- [42] Han, Y., Yang, D.Some new spaces of Besov and Triebel-Lizorkin type on homogeneous spaces, Studia Math.156 (2002), 67–97.
- [43] Hansson, K. Imbedding theorems of Sobolev type in potential theory, Math. Scand. 45 (1979), 77–102.
- [44] Haroske, D. D. and Schneider, C. Besov spaces with positive smoothness on \mathbb{R}^n , embeddings and growth envelopes. J. Approx. Theory. **61** (2009), 723–747.
- [45] Heikkinen, T., Ihnatsyeva L. and Tuominen, H. Measure Density and Extension of Besov and Triebel-Lizorkin functions, J. Fourier Anal. Appl. 22 (2016),334–382.
- [46] Heinonen, J. Lectures on analysis on metric spaces Universitext. Springer-Verlag, New York, 2001. x+140 pp.
- [47] Heinonen, J., Kilpelainen, T., and Martio, O., Nonlinear potential theory of degenerate elliptic equations, Oxford Univ. Press, Oxford, 1993
- [48] Hunt, R. On L(p,q) spaces. Enseignement Math. 12 (1966), 249–276.
- [49] Hytönen, T., Pérez, C. and Rela, E. Sharp Reverse Holder property for A_{∞} weights on spaces of homogeneous type, J. Funct. Anal. **263** (2012), 3883–3899.
- [50] Jonsson, A. Besov Spaces on Closed Subsets of \mathbb{R}^n , Trans. Amer. Math. Soc. **341** (1994), 355–370.
- [51] Järvi, P. and Vuorinen, M. Uniformly Perfect Sets and Quasiregular Mappings, J. London Math. Soc. 54 (1996), 515-–529.
- [52] Karak. N. Lower bound of measure and embeddings of Sobolev, Besov and Triebel-Lizorkin spaces, preprint.
- [53] Krein, S. G., Petunin, Yu. I. and Semenov, E. M. "Interpolation of linear operators," *Transl. Math. Monogr. Amer. Math. Soc.* 54, Providence, (1982).

- [54] Keith, S. and Zhong, X. The Poincaré inequality is an open ended condition, Annals of Mathematics. Second Series. 167 (2008), 575–599.
- [55] Kolyada, V.I., Estimates of rearrangements and embedding theorems. Mat. Sb. 136 (1988), 3–23 (in Russian); English transl.: Math. USSR-Sb. 55 (1989), 1–21
- [56] Korevaar, N. J. and Schoen, R. M. Sobolev spaces and harmonic maps for metric space targets, Communications in Analysis and Geometry. 1 (1993), 561–659.
- [57] Lerner, A. K. Weigthed rearrangement inequalities for local sharp maximal functions, Trans. AMS. 357 (2004), 2445–2465.
- [58] Lott, J. and Villani, C. Ricci curvature for metric-measure spaces via optimal transport, Annals of Mathematics. Second series. 169 (2009), 903–991.
- [59] Martin, J. and Milman, M. Symmetrization inequalities and Sobolev embeddings, Proc. Amer. Math. Soc. 134 (2006), 2335–2347
- [60] Martin, J. Symmetrization inequalities in the fractional case and Besov embeddings, J. Math. Anal. Appl. 344 (2008), 99–123.
- [61] Martín, J. and Milman, M. Pointwise symmetrization inequalities for Sobolev functions and applications, Adv. Math. 225 (2010), 121–199.
- [62] Martín, J. and Milman, M. Sobolev inequalities, rearrangements, isoperimetry and interpolation spaces, Concentration, Functional Inequalities and Isoperimetry, Contemp. Math., vol. 545, Amer. Math. Soc., Providence, RI, 2011, pp. 16793
- [63] Martín, J. and Milman, M. Fractional Sobolev inequalities: symmetrization, isoperimetry and interpolation, Astérisque 366 (2014), x+127 pp.
- [64] Martín, J., Milman, M. and E. Pustylnik, Sobolev inequalities: Symmetrization and self-improvement via truncation, J. Funct. Anal. 252 (2007), 677– 695.
- [65] Martín, J. and Milman, M. Isoperimetry and Symmetrization for Logarithmic Sobolev inequalities, J. Funct. Anal. 256 (2009), 149–178.
- [66] Martín, J. and Milman, M. Isoperimetric weights and generalized uncertainty inequalities in metric measure spaces, J. Funct. Anal. 270 (2016), 3307–3343.
- [67] Martín, J. and Ortiz, W.A. A Sobolev type embedding theorem for Besov spaces defined on doubling metric spaces, J. Math. Anal. Appl. 479 (2019), 2302–2337.

- [68] Martín, J. and Ortiz, W.A. Symmetrization inequalities for probability metric spaces with convex isoperimetric profile, preprint.
- [69] Mastyło, M. The modulus of smoothness in metric spaces and related problems Potential Anal. 35 (2011), 301–-328.
- [70] Maz'ya, V. G. Sobolev Spaces, Springer-Verlag, New York, 1985.
- [71] Maz'ya, V.G. Classes of regions and imbedding theorems for function spaces, Dokl. Akad. Nauk SSSR 133 (1960), 527–530 (Russian). English translation: Soviet Math. Dokl. 1 (1960), 882– 885.
- [72] Milman, E. Beyond traditional curvature-dimension I: new model spaces for isoperimetric and concentration inequalities in negative dimension, Trans. Amer. Math. Soc. 369 (2017), 3605–3637.
- [73] Moura, S. D. On some characterizations of Besov spaces of generalized smoothness Math. Nachr. 280 (2007), 1190–1199.
- [74] Müller, D. and Yang, D.A difference characterization of Besov and Triebel-Lizorkin spaces on RD-spaces, Forum Math. 21 (2009), 259–298.
- [75] Muramatu, T. On Besov spaces and Sobolev spaces of generalized functions definded on a general region, Publ. Res. Inst. Math. Sci. 9 (1973/74), 325– 396.
- [76] Nagel, A., Ricci, F., Stein E.M. Harmonic analysis and fundamental solutions on nilpotent Lie groups, Analysis and differential equations 249–275. Lecture notes in pure and Appl. Math. 122 (1990).
- [77] Nagel, A., Ricci, F., Stein E.M. Singular integrals with flag kernels and analysis on quadratic CR manifolds, J. Funct. Anal. 181 (2001), 29–118.
- [78] Nagel, A., Stein E.M. Differentiable control metrics and scaled bump functions, J. Differential Geom. 57 (2001), 465–492.
- [79] Nagel, A., Stein E.M. On the product theory of singular integrals, Rev. Mat. Ibero. 20 (2004), 531–561.
- [80] Nagel, A., Stein E.M. The $\overline{\nabla}_b$ -complex on decoupled boundaries in \mathbb{C}^n , Ann. of Math. **164** (2006), 649–713.
- [81] Nagel, A., Stein E.M., Wainger, S. Balls and metrics defined by vector fields I. Basic properties, Acta Math. 155 (1985), 103–147.
- [82] Netrusov, Yu. V. Theorems for the embedding of Besov spaces into ideal spaces. (Russian) Zap. Nauchn. Sem. Leningrad. Otdel. Mat. Inst. Steklov. (LOMI) 159 (1987), Chisl. Metody i Voprosy Organiz. Vychisl. 8, 69–82, 177; translation in J. Soviet Math. 47 (1989), 2871–2881.

- [83] O'Neil, R. Convolution operators and L(p,q) spaces. Duke Math. J. **30** (1963), 129–142
- [84] Peetre, J. Espaces dinterpolation et théoréme de Soboleff. Ann. Inst. Fourier (Grenoble) 16 (1966), 279–317.
- [85] Pohožaev, S. I. On the eigenfunctions of the equation $\Delta u + \lambda f(u) = 0$. (Russian) Dokl. Akad. Nauk SSSR **165** (1965), 36–39.
- [86] Rakotoson, J.M. and B. Simon, B. Relative Rearrangement on a finite measure space spplication to the regularity of weighted monotone rearrangement (Part 1), Rev. R. Acad. Cienc. Exact. Fis. Nat. 91(1997), 17–31.
- [87] Semmes, S. An introduction to analysis on metric spaces, Notices of the American Mathematical Society. 50 (2003), 438–433.
- [88] Shen, Z. Complete manifolds with nonnegative Ricci curvature and large volume growth, Invent. Math. 125 (1996), 393–404.
- [89] Sturm, K.T. On the geometry of metric measure spaces. I, Acta Mathematica, 196 (2006), 65–131.
- [90] Sturm, K.T. On the geometry of metric measure spaces. II, Acta Mathematica. 196 (2006), 133–177.
- [91] Sobolev, S.L. Sur un théoreme d'analyse fontionelle, Mat. Sb., Ser.4 (1938), 471–497.
- [92] Susan G. Staples and Lesley A. Ward. Quasisymmetrically thick sets, Ann. Acad. Sci. Fenn. Math., 23(1998), 151–168.
- [93] Tartar, L. Imbedding theorems of Sobolev spaces into Lorentz spaces, Bollettino dell'Unione Matematica Italiana 1 (1998), 479–500.
- [94] Triebel, H. Theory of function spaces. Birkhäuser Verlag, Basel, 1983.
- [95] Triebel, H. Theory of function spaces II. Birkhäuser Verlag, Basel, 1992.
- [96] Triebel, H. Theory of function spaces III. Birkhäuser Verlag, Basel, 2006.
- [97] Trudinger, Neil S. On imbeddings into Orlicz spaces and some applications, J. Math. Mech. 17 (1967), 473–483.
- [98] Vázquez, J. L. An introduction to the mathematical theory of the porous medium equation, In Shape optimization and free boundaries (Montreal, PQ, 1990), volume 380 of NATO Adv. Sci. Inst. Ser. C Math. Phys. Sci., pages 347–389. Kluwer Acad. Publ., Dordrecht, 1992.
- [99] Varopoulos, N.T. Analysis on Lie groups, J. Funct. Anal. 76 (1988), 346–410.

- [100] Varopoulos, N.T., Saloff-Coste, L., Coulhon, T.Analysis and geometry on groups, Cambridge Tracs in Mathematics. 100 (1992).
- [101] Wadade, H. Remarks on the critical Besov space and its embedding into weighted Besov-Orlicz spaces. Studia Math. 201
- [102] Wu. J.M. Null sets for doubling and dyadic doubling measures, Ann. Acad. Sci. Fenn. 18 (1993), 77–91.
- [103] Yang, D. Real interpolations for Besov and Triebel-Lizorkin spaces of homogeneous spaces, Math. Nachr. 273 (2004), 96–113.
- [104] Yang, D. Yuan, W. and Zhuo, C. Musielak-Orlicz Besov-type and Triebel-Lizorkin-type spaces. Rev. Mat. Complut. 27 (2014), 93–157.
- [105] Yang D. and Zhou, Y. New properties of Besov and Triebel-Lizorkin spaces on RD-spaces, Manuscripta Math. 134 (2011), 59–90.
- [106] Zhan, H. The Manifolds with Nonnegative Ricci Curvature and Collapsing Volume, Proceedings of the American Mathematical Society. 135 (2007), 1923–927.
- [107] Zippin, M. Interpolation of operators of weak type between rearrangement invariant spaces. J. Functional Analysis 7 (1971), 267–284.